Novel Haptic Interfaces and their Impact on Perception and Performance



Jens Maiero

Department of Computer Science Brunel University London

A thesis submitted for the degree of Doctor of Philosophy

Brunel University

April 2021

Abstract

This thesis explores novel haptic user interfaces for touchscreens, virtual and remote environments (VE and RE). All feedback modalities have been designed to study performance and perception while focusing on integrating an additional sensory channel - the sense of touch. Related work has shown that tactile stimuli can increase performance and usability when interacting with a touchscreen. It was also shown that perceptual aspects in virtual environments could be improved by haptic feedback. Motivated by previous findings, this thesis examines the versatility of haptic feedback approaches. For this purpose, five haptic interfaces from two application areas are presented. Research methods from prototyping and experimental design are discussed and applied. These methods are used to create and evaluate the interfaces; therefore, seven experiments have been performed.

All five prototypes use a unique feedback approach. While three haptic user interfaces designed for touchscreen interaction address the fingers, two interfaces developed for VE and RE target the feet. Within touchscreen interaction, an actuated touchscreen is presented, and study shows the limits and perceptibility of geometric shapes. The combination of elastic materials and a touchscreen is examined with the second interface. A psychophysical study has been conducted to highlight the potentials of the interface. The back of a smartphone is used for haptic feedback in the third prototype. Besides a psychophysical study, it is found that the touch accuracy could be increased. Interfaces presented in the second application area also highlight the versatility of haptic feedback. The sides of the feet are stimulated in the first prototype. They are used to provide proximity information of remote environments sensed by a telepresence robot. In a study, it was found that spatial awareness could be increased. Finally, the soles of the feet are stimulated. A designed foot platform that provides several feedback modalities shows that self-motion perception can be increased.

- to my father and my loved little family.

Acknowledgements

There were many companions and many who supported me during the time of my thesis, whether professionally or personally. I would like to thank all of them from the bottom of my heart.

First of all, I would like to thank my first supervisor, Prof. Dr. George Ghinea, for his continuous and precious support and, in some periods, for his patience. He offered me the great opportunity to do my PhD at Brunel University, and he has guided and encouraged me throughout the whole time to achieve my goal. I could not have wished for a better supervisor. I also would like to express my deepest gratitude to my second supervisor Prof. Dr. André Hinkenjann. He provided me with the chance to research with him during my master's studies, and he supported me in all my decisions. Without his persistent help, this thesis would not have been possible.

I wish to show my gratitude to my collaborator Prof. Dr. Ernst Kruijff, with whom I have been discussed a lot about my research, and he always came up with new approaches and ideas that contributed to my research. The numerous research trips that we took together, in which we experienced a lot together, were a significant enrichment both in a professional and personal sense. Many thanks also go to my long-time office colleague Martin Weier, he has experienced me during all the ups and downs. The discussions with him were always professionally excellent and personally refreshing. I would also like to thank David Eibich, who has joined us in recent years, very much for his technical support. All of you have always been a great source of inspiration.

A large part of this work was carried out at the Bonn-Rhein-Sieg University of Applied Sciences, especially at the Institute for Visual Computing. I would like to thank all researchers who supported me. My special thanks go to Oliver Jato, Thorsten Roth, Katharina Stollenwerk, Alexander Marquardt, Christina Trepkowski, Martin Göbel, Saugata Biswas, Jessica Milberg, David Scherfgen and Timur Saitov, the latter two I would especially like to thank for the excellent badminton lessons. A big thanks also goes to all co-authors who have contributed to this thesis with their work. I would also like to thank the Department of Computer Science, which supported me with a scholarship.

My special thanks go to all my friends, who always offered me an exciting alternative program; the philosophical discussions during the numerous hikes were still great!

I would like to pay my special regards to my father, who made my studies possible and made me the person I have become. I would also like to thank my brother and his family for their support.

Finally, I would like to express my sincerest thanks to my little family, thanks for being patient. You were definitely the loved ones who experienced and supported me the most during this time - there are none better, thanks Ramona and Maila!

Contents

List of Figures ix						
Lis	st of	Tables	5	xiii		
Lis	st of	Public	cations	xiv		
1	Intr	oducti	on	1		
	1.1	Contex	xt and Motivation	1		
	1.2	Aims a	and Objectives	5		
	1.3	Thesis	outline	8		
2	Lite	rature	Review - Haptic User Interfaces	10		
	2.1	Funda	mentals	11		
		2.1.1	Sense of Touch	11		
		2.1.2	Actuators	14		
		2.1.3	Touchscreen Technology	17		
	2.2	Resear	ch on Haptics	19		
		2.2.1	Touchscreen Interaction	19		
		2.2.2	Pen Prototypes	26		
		2.2.3	Back-of-Device Interaction	29		
		2.2.4	Tangible and Shape-Shifting User Interfaces	33		
		2.2.5	Flexible and Mid-air User Interfaces	36		
		2.2.6	Virtual Reality	42		
		2.2.7	Remote Environments	46		
		2.2.8	Navigation Cues	49		
	2.3	Summ	ary and Research Questions	51		
		2.3.1	Haptic Feedback for Touchscreens	52		
		2.3.2	Haptic Feedback for Virtual and Remote Environments	56		
3	Res	earch]	Methodology	58		
	3.1	Introd	uction	58		
	3.2	Quant	itative vs Qualitative Research	61		

	3.3	Haptic	es Exploration 63
	3.4	Protot	vning Haptic Interfaces 66
	3.5	Experi	imental Design 71
	3.6	Sampl	ing 74
	3.0	Collog	$\frac{1}{76}$
	J.1 2 0	El	ting data
	0.0 2.0		$\lim_{n \to \infty} \sum_{i=1}^{n} \sum_{j=1}^{n} \sum_{i=1}^{n} \sum_{i=1}^{n} \sum_{i=1}^{n} \sum_{j=1$
	3.9	Validit	
	3.10	Ethics	
4	Нар	otic Int	erfaces for Touchscreens 84
	4.1	Introd	uction
	4.2	Haptic	e Feedback through an Actuated Touchscreen
		4.2.1	Implementation
		4.2.2	Experiment - Low Level Perception
		4.2.3	Application Scenarios
		4.2.4	Discussion
		4.2.5	Conclusion
	4.3	Elastic	e Feedback above a Touchscreen
		4.3.1	Implementation
		4.3.2	Experiment - Perception Study 109
		4.3.3	Application scenarios
		4.3.4	Discussion
		4.3.5	Conclusion
	4.4	Pin-ba	sed Feedback on the Back-of-the-Device
		4.4.1	Implementation
		4.4.2	Experiment 1 - Psychophysical Perception Study 125
		4.4.3	Experiment 2 - Touch Accuracy Study 129
		4.4.4	Application scenarios
		4.4.5	Discussion
		4.4.6	Conclusion
	4.5	Summ	ary 141
5	Foot	t Hant	ic Interfaces for VE and BE 144
-	5.1	Introd	uction
	5.2	Foot-h	pased Haptic Proximity Cues in VE und RE
		5.2.1	Implementation
		522	Experiment 1 - Virtual Environment Navigation 151
		5 2 2	Experiment 2 Real Environment Tacks
		0.4.0	Experiment 2 - meat Environment rasks 199

		5.2.4	Discussion	 168
		5.2.5	Conclusion	 170
	5.3	Combi	ining Haptic Modalities using a Foot-Platform in VR	 171
		5.3.1	Implementation	 172
		5.3.2	Experiments	 175
		5.3.3	Discussion	 188
		5.3.4	Conclusion	 189
	5.4	Summ	ary	 190
6	Con	clusio	n	192
	6.1	Summ	ary	 192
	6.2	Contri	ibutions	 194
	6.3	Limita	ations	 196
	6.4	Future	e Directions	 197
	6.5	Final V	Words	 198
Re	efere	nces		199
A	ppen	dix A	Questionnaires	220
A	ppen	dix B	Further Results	224

List of Figures

1.1	Classification of the individual areas that are important in this thesis	6
2.1	Characteristics of the four mechanoreceptors located in our skin [Gol10].	12
2.2	Inside and outside view of the haptic mouse prototype of Akamatsu [AS94].	20
2.3	Haptic Tabeltop Puck presented by [MNY ⁺ 09].	20
2.4	First tactile feedback introduced by Fukumoto and Sugimura [FS01]. $\ .$.	21
2.5	Poupyrev and Maruyama using piezo electric actuation $\ldots \ldots \ldots$	21
2.6	Frictional Widgets used for tactile feedback in a file manager $[LOM^+11]$.	24
2.7	Electrovibration to provide friction between the finger and the touchscreen	24
2.8	Touch Mover a stereoscopic actuated touch screen	25
2.9	A horizontally mounted tablet for haptic interaction	25
2.10	Second version of the Haptic Edge prototype	26
2.11	Actuation of the tactile pixels at the side of a smartphone $\hfill\hfi$	26
2.12	Hardware prototype showing all built-in components of the haptic pen $% \mathcal{A}$.	27
2.13	RealPen, a pen enhanced with a linear resonant actuator $\ldots \ldots \ldots$	29
2.14	Three main style to hold a smartphone	29
2.15	An example of an one-handed mapping of common gestures \ldots .	31
2.16	A reconfigureable tactile shape-changing interface	34
2.17	A textile tactile display, allows active and passive feedback $\hdots \hdots \$	34

2.18	Pin-based force feedback display in FORM	35
2.19	InFORCE an extentsion of the inFORM display	35
2.20	Jamming interfaces: the technique behind jamming	36
2.21	Flexible interaction using FlexPad	37
2.22	Depthtouch: elastic material to visualize informations	37
2.23	Medusa: a tabletop that can sense three levels of a users \ldots	39
2.24	A highly sensitive capacitive touchscreen	39
2.25	A prototype which renders haptic feedback above a screen $\ . \ . \ . \ .$	40
2.26	Exemplary representation of a magnetic-field based prototype \ldots .	40
2.27	The back of the shoe interface shows two vibrations motors	43
2.28	Foot-based navigation using vibration motors	44
2.29	A conference scenario showing the remote robot beam $\ldots \ldots \ldots$	47
2.30	The difference between control your telerobot, and being telepresent	47
9 21	A summary schematic representation of the dependencies and intersections	51
2.01	A summary schematic representation of the dependencies and intersections	91
3.1	An overarching diagram of all research studies that have been performed	59
3.13.2	An overarching diagram of all research studies that have been performed A standard iterative cycle of scientific methods	59 64
 3.1 3.2 3.3 	An overarching diagram of all research studies that have been performed A standard iterative cycle of scientific methods	51596465
 3.1 3.2 3.3 3.4 	An overarching diagram of all research studies that have been performed A standard iterative cycle of scientific methods	 51 59 64 65 67
 3.1 3.2 3.3 3.4 3.5 	An overarching diagram of all research studies that have been performed A standard iterative cycle of scientific methods	 51 59 64 65 67 68
 3.1 3.2 3.3 3.4 3.5 3.6 	An overarching diagram of all research studies that have been performed A standard iterative cycle of scientific methods	 51 59 64 65 67 68 81
 3.1 3.2 3.3 3.4 3.5 3.6 4.1 	An overarching diagram of all research studies that have been performed A standard iterative cycle of scientific methods	 51 59 64 65 67 68 81 87
 3.1 3.2 3.3 3.4 3.5 3.6 4.1 4.2 	An overarching diagram of all research studies that have been performed A standard iterative cycle of scientific methods	 51 59 64 65 67 68 81 87 87
 3.1 3.2 3.3 3.4 3.5 3.6 4.1 4.2 4.3 	An overarching diagram of all research studies that have been performed A standard iterative cycle of scientific methods	 59 64 65 67 68 81 87 87 88
 3.1 3.2 3.3 3.4 3.5 3.6 4.1 4.2 4.3 4.4 	An overarching diagram of all research studies that have been performed A standard iterative cycle of scientific methods	 51 59 64 65 67 68 81 87 87 88 91
 2.31 3.1 3.2 3.3 3.4 3.5 3.6 4.1 4.2 4.3 4.4 4.5 	An overarching diagram of all research studies that have been performed A standard iterative cycle of scientific methods	 51 59 64 65 67 68 81 87 87 88 91 93

4.7	Box plots showing the detection rate per width	96
4.8	Detection rate at different velocities (left) and learning rate (right)	97
4.9	Distance to the global and local maximum for each shape	98
4.10	Depicts two possible application scenarios	99
4.11	Our multilayer system consists of a stretchable layer	102
4.12	Stress-strain curves at 3 measured positions $(1,2,3)$	104
4.13	Schematic side view of the system, showing the pulling mechanism	105
4.14	Versatility of feedback types afforded by our system	106
4.15	A schematic representation of the performed study $\ldots \ldots \ldots \ldots$	109
4.16	Apparatus, with sound and visual cover	111
4.17	Three representatives of the conducted staircase protocol $\ldots \ldots \ldots$	111
4.18	Psychometric curves (pen press/slide)	113
4.19	Art application: rigid and flexible states simulate different tools \ldots .	114
4.20	HapticPhone: a touchscreen for thumb interaction and force feedback pins	121
4.21	Iteration stages of our back-of-the-device approach	122
4.22	Technical drawing of the top and side view of our interface	123
4.23	Modified scotch-yoke mechanism to translate rotational into linear movements	s123
4.24	System latency measurements for force, tactile and display	124
4.25	A schematic overview of the touch accuracy study	130
4.26	Exemplary representation of the target sizes in the drag-and-drop study .	131
4.27	Exemplary representation of the different target matrix sizes	133
4.28	Box-plot for touch accuracy in the drag-and-drop study	135
4.29	Subjective performance of the performed drag-and-drop task \ldots .	135
4.30	Successfully selected cells per size and mode	137
4.31	Subjective performance of the performed selection study	137
4.32	Exemplary application scenarios for the proposed BoD force feedback approach	138
4.33	Sketch of implemented haptic feedback approaches	142

5.1	The ring consisting of 12 ultrasonic distance sensors attached to the Beam+	148
5.2	The foot platform consisting of six movable actuator blocks	148
5.3	A schematic representation of the whole interface and its components	149
5.4	A diagrammatic representation of the study design	152
5.5	Summary of the individual ratings of the Wittmer and Singer questionnaire	155
5.6	Ratings of the individual questions that showed a significant effect. \hdots .	156
5.7	A diagrammatic representation of the study design	160
5.8	Summary of the individual ratings of the Wittmer and Singer questionnaire	164
5.9	Overview of the individual sub-scores of the NASA-TLX questionnaire $% \mathcal{A}^{(1)}$.	165
5.10	A difference observed between haptic feedback and no feedback	167
5.11	Driving the robot down a ramp (left) and driving in a confined space (right).	167
5.12	Feedback to each foot consists of a loudspeaker mounted \ldots .	173
5.13	Foot pressure distribution zones of a standing user	173
5.14	User in standing leaning (left) and seated (middle) pose	176
5.15	A diagrammatic view on the overall study	178
5.16	In-study questionnaire showed an significant effect	182
5.17	Users could better concentrate on the task and better navigate	182
5.18	Audio and vibration are the most relevant parameters	183
5.19	A significant effect was found for all variables	184
5.20	Gamepad interaction was clearly preferred by users	185
5.21	When enable all haptic cues, perception will increase $\ldots \ldots \ldots \ldots$	186
5.22	Main results of the first three sessions in terms of completion time	187
5.23	Effect of our four feedback components on vection over all sessions	188
5.24	Sketch of both implemented foot haptics approaches	190
B.1	Results of psychophysical study performed to analyse our third prototype	224
B.2	Results of the in-study questionnaire of session 4	225
B.3	Results of the post-study questionnaire of session 4	225

List of Tables

2.1	Structured overview about the main related work	18
3.1	Differences between a quantitative and a qualitative approach \ldots	63
3.2	This table shows common sensors and actuators	70
4.1	Detection rate and error rate of the 4 shapes	95
4.2	List of the results of the psychophysical perception experiment	128
4.3	Completion time of the drag-and-drop task	134
4.4	Completion time of the selection task $\ldots \ldots \ldots \ldots \ldots \ldots \ldots$	136
5.1	Overall ratings of the studied interfaces (likert scale 0-10)	181

List of Publications

The following publications have been accepted for publication as a direct or indirect result of the research during the time as a PhD student. In this thesis, especially [J1] and [C1-C4] are addressed. At the end of each reference is a link to the corresponding section in which parts of the publication can be found. I have designed and developed all of the five prototypes presented in this thesis. In those publications where I am not listed as the first author, I contributed an essential part to the idea, implementation, experimental design and writing part. Results and evaluations which were done collaborative have been evaluated and interpreted anew in the context of this thesis. The second part, further selected publications, aims to show that the research methods used in this thesis can be transferred to other fields.

Journals [J*] and Conferences [C*]:

- [J1] Jens Maiero, David Eibich, Ernst Kruijff, André Hinkenjann, Wolfgang Stürzlinger, Hrvoje Benko, Gheorghita Ghinea, Back-of-Device Force Feedback Improves Touchscreen Interaction for Mobile Devices, *IEEE Transactions on Haptics*, 2019 (refers to Section 4.4)
- [C1] Jens Maiero, Ernst Kruijff, André Hinkenjann, Gheorghita Ghinea, ForceTab: Visuo-haptic Interaction with a Force-senstive Actuated Tablet, IEEE International Conference on Multimedia and Expo (ICME), 2017, [Best Paper Award] (refers to Section 4.2)
- [C2] Ernst Kruijff, Saugata Biswas, Christina Trepkowski, Jens Maiero, Gheorghita Ghinea, Wolfgang Stuerzlinger, Multilayer Haptic Feedback for Pen-Based Tablet Interaction, Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems (CHI) (refers to Section 4.3)

- [C3] Brennan Jones, Jens Maiero, Alireza Mogharrab, Ivan Abdo Aguliar, Ashu Adhikari, Bernhard Riecke, Ernst Kruijff, Carman Neustaedter, Robert W. Lindeman, FeetBack: Augmenting Robotic Telepresence with Haptic Feedback on the Feet, Proceedings of the 2020 International Conference on Multimodal Interaction (ICMI) (refers to Section 5.2)
- [C4] Ernst Kruijff, Alexander Marquardt, Christina Trepkowski, Robert W. Lindeman, André Hinkenjann, Jens Maiero, Bernhard Riecke E., On Your Feet! Enhancing Vection in Leaning-Based Interfaces through Multisensory Stimuli, Symposium on Spatial User Interfaces, 2016 (refers to Section 5.3)

Further selected publications:

- [C5] Jens Maiero, Ernst Kruijff, André Hinkenjann, Gheorghita Ghinea, PICOZOOM: A Context Sensitive Multimodal Zooming Interface, IEEE International Conference on Multimedia and Expo (ICME), 2015
- [J2] Jens Maiero, Ernst Kruijff, André Hinkenjann, Gheorghita Ghinea, Focus plus Context Techniques for Pico-Projection based Interaction, *IEEE Transactions on Multimedia*, 2017
- [C6] Alexander Wilberz, Dominik Leschtschow, Christina Trepkowski, Jens Maiero, Ernst Kruijff, Bernhard Riecke, FaceHaptics: Robot Arm based Versatile Facial Haptics for Immersive Environments, Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems (CHI)
- [C7] Anil Ufuk Batmaz, Jens Maiero, Bernhard Riecke, Ernst Kruijff, Carman Neustaedter, Wolfgang Stuerzlinger, Telepresence Robot Navigation in Conference-Like Environments with Distance-Based Automatic Speed Control, Proceedings of the ACM on Interactive, Mobile, Wearable and Ubiquitous Technologies - in submission
- [C8] Alexander Marquardt, Jens Maiero, Ernst Kruijff, Christina Trepkowski, Andrea Schwandt, André Hinkenjann, Johannes Schoening, Wolfgang Stuerzlinger, Tactile Hand Motion and Pose Guidance for 3D Interaction, In Proceedings of the ACM Symposium on Virtual Reality Software and Technology, 2018
- [C9] Christina Trepkowski, David Eibich, Jens Maiero, Alexander Marquardt, Ernst Kruijff, Steven Feiner, The Effect of Narrow Field of View and Information Density on Visual Search Performance in Augmented Reality, Proceedings of the IEEE conference on Virtual Reality (VR) 2019

Chapter 1

Introduction

New technologies have grown immensely in recent decades. Most people use a smartphone for everyday things. Virtual environments allow users to immerse themselves in computergenerated worlds and meet with others. Telepresence systems allow users to attend remote events in person. Robots and artificial intelligence are other strong trends that reflect the impact of digitalisation. From a philosophical point of view, it can be said that computers and humans have become closer in recent years. Somehow, what used to be science fiction is now reality. However, what does the interaction between humans and computers actually look like? Will perception and performance be improved when multiple senses are used for interaction? This thesis presents five novel haptic user interfaces that show that engaging the sense of touch positively affects touchscreen interaction and perception in virtual and remote environments.

1.1 Context and Motivation

Looking briefly at the history of human-computer interaction (HCI) [Mye98], it can be seen that people have been searching for more effective ways of interacting with computers ever since the beginning of the computer era. The first computing operations were realized using punched card systems, and there was no human-computer interaction in the way that it is understood today. At that time, computers were used to perform specific tasks, sequentially, and only a few experts were able to operate them. With the first personal computers, such as the Apple 2 (1977) and the IBM PC (1981), a command-line was used to operate and interact with the operating system. Although the first research results on graphical user interfaces were already available at this time, they were not included in their systems. In December 1968, at the Stanford Research Institute in California, Doug Engelbart gave a legendary demonstration of how to control content on a computer monitor with a mouse [EE68]. However, it was still time before a graphical user interface (GUI) was available to a mass audience.

In the 1970s, graphical user interfaces were developed at the Californian Research Center Xerox PARC. However, for some reason, GUIs were not accepted by the market. In 1983, the Apple Lisa brought in first ideas of graphical user interfaces, and one year later, the Apple Macintosh was born. Two years later, Microsoft also presented the first version of Windows however, the graphical user interface's concept and implementation were not very user-friendly. The successor version, Windows 3.1; received more acceptance from users. However, the breakthrough came with Windows 95 and the introduction of the start button. Since then, human-computer interaction has continued to evolve, and rapid development has continued.

Mobile phones, later smartphones, have had a significant influence on HCI. The first mobile phones appeared in 1983 (Motorola DynaTAC 8000X), obviously without a touchscreen. The first smartphone (Apple iPhone) with a touchscreen, appeared in 2007, when Apple also introduced gesture-based interaction using touchscreens. Google presented the first Android smartphone about a year later. Multitouch gestures, such as pinch to zoom or rotate, are now indispensable. Early mobile phones had physical buttons to interact with. Display and interaction were physically decoupled. With the introduction of touchscreens, it became possible to combine display and interaction within a single device. However, feeling a button and sensing physical events has an impact on perceptual and performance aspects. That is why many researchers have been working on recreating precisely these lost properties as closely as possible. For example, vibration motors are used to enrich graphical user interface (GUI) elements with tactile feedback; even blind people can benefit. They are much more dependent on the haptic exploration. It seems that the combination of touchscreen interaction with haptic feedback has much potential.

However, haptic feedback can also be useful in other areas. When looking at the history of virtual reality (VR), it becomes clear that the rendering of physical properties, for instance, feeling textures, will enhance the experience. A milestone in VR history was the head-mounted display (HMD) by Ivan Sutherland in 1968. His stereoscopic display system made it possible to perceive three-dimensional environments visually. In the 1980s, it was realised that there was more to VR than just visual stimuli and other senses began to be included; for example, data gloves were explored. Since around 2010, starting with Oculus Rift, there was another VR hype. Alongside stereoscopic displays, HMDs were now available on the consumer market and affordable for everyone. While photorealistic three-dimensional environments can be rendered almost in real-time, the representation of physical properties, for example, not just seeing objects but also feeling them, is still an open issue in this domain.

Interaction, including haptic interaction and exploration, with a computer has been formalized over time; two models will be presented to demonstrate how interaction works. Interaction is a dynamic action that adapts inputs through a dialogue with feedback and provides corresponding results. In particular, haptic feedback aims at either changing a perceptual state or improving interaction performance. Norman's model [Nor13] focuses on the user's view, his interaction model consists of six stages per cycle. Each cycle begins with an execution part intended to achieve an established goal: (i) form the intention, (ii) specify the action sequence and (iii) execute the action. This is followed by an evaluation part: (i) perceive the system state, (ii) interpret the system state and (iii) evaluate the system state with respect to the goals. Norman introduced the terms gulf of execution and gulf of evaluation to provide an understanding of interface failures, while the gulf of execution refers to a misinterpretation of the functionalities provided by the system and how to use them, the *qulf of evaluation* is the difference between the physical state or feedback, and the state assumed by the user. It is important when designing haptic user interfaces to know and consider these possible failures. Abowd and Beale's model [DFAB04] is a universal version of Norman's interaction model which has four components: (i) user, (ii) input, (iii) system and (iv) output. Each component has its own language. An interaction cycle starts with the user's input; the input pursues a certain goal. In this context, a failure in the interface or in the interaction, is understood as a misinterpretation or mistranslation of the corresponding language.

The interaction models presented above are based on the fact that users can perceive and communicate with their world. A person's senses are used to interact with and in the real world; they help individuals communicate with other people and perform everyday tasks. Humans have five senses: sight, hearing, taste, smell and touch [Gol10]. In principle, the senses all function on a similar principle. Environmental stimuli such as light or odours are converted by the appropriate sense, into electrical neural signals that then reach the brain through the nervous system and are processed there [Gol10]. Each sensory organ has evolved its method of transforming an environmental stimulus. The eyes have rods and cones, while the ear converts vibrations through various stages, into nerve signals. The olfactory mucosa and the taste buds transmit smell and taste to the brain. The skin works differently, as it consists of thermoreceptors (hot and cold), nociceptors (pain) and mechanoreceptors (pressure), which convert the stimuli into nerve signals. Although the entire body has these receptors, some areas are more sensitive than others [DFAB04]. Kinaesthesia is a further, little known part of the sense of touch, which is the awareness of the body and limbs' position.

After this brief excursion into human physiology, we return to interaction with computers and look at the sensory channels most commonly used in HCI. The visual and auditory channels dominate in HCI [Sut03] which is in line with the discussion about the hierarchy of the senses, which has existed since Aristotle [Gru08]. However, when considering this order from a technological point of view, it should also be considered that visual and auditory sensations are more straightforward to synthesise than haptics. Of course, tactile feedback can easily be generated using vibration motors, but if looking at the simulation of real properties, haptic feedback becomes difficult. This is due to the fact that visual and auditory sensations are perceived by specialised organs, the eyes and ears. In contrast, a sensation of force can be perceived anywhere on the human body and is therefore directly related to physical contact [Gru08]. Another difference that makes haptic interaction fascinating is that it has a unique property compared to our other senses: interaction is bi-directional, for example, in manipulation tasks or when pressing a button. In both cases, input and output occur at the same location at our body [EOEC11].

Interacting with multiple senses leads to various concepts; to avoid confusion, four often used in this context will be introduced: multimedia, mulsemedia, multimodal and multisensory. The following explanations are inspired by [OSC⁺17]. The term multimedia is probably the first that has been established. Multimedia is a communication of information that uses more than one medium. It is rather a passive way of interaction and focuses on the system components. Whereas multimedia applications are mostly bi-sensorial in nature, mulsemedia applications are those that engage three (or more) of our senses [GTLG14]. In multimodal interaction, the focus is on modalities such as speech, gesture, touch or gaze. Multiple senses are not necessarily involved, in the definition by Oviatt et al. [OSC⁺17], it is assumed that the motor system is involved in the interaction. Multisensory interaction is a more general term and describes an interaction that is based on more than one sensory channel and is in comparison to multimedia more active. If we look at the haptic interfaces that have been developed in the context of this thesis, they can mainly be classified into two of the categories explained above, namely multisensory and multimodal.

When considering the potential of the sense of touch, the haptic sensation has many convincing aspects - haptic feedback can bring a wide range of improvements. This is why researching novel interfaces that address particular challenges is so important. For example, haptic perception can increase presence, which is especially important for virtual environments and for telepresence scenarios in remote environments to represent the whole rather than just a part. Furthermore, the sense of touch allows us to explore the world and objects. This is often referred to as active touch. Not just seeing objects on a touchscreen but also feeling them would be a huge step forward in HCI. Sensing geometry on a touchscreen would not only help blind people but could also create new interaction metaphors. It would be possible to interact with content and information without looking at the display (eyes-off). In this case, the eyes could be used for something different. Another point that makes haptic interaction so interesting is to feel and sense objects or people who are not in the same room. Haptic feedback could make it possible to feel textures of objects without owning them or interacting emotionally with close friends or acquaintances. Moreover, haptic feedback is essential when controlling robots in dangerous areas or during remote maintenance in order to take the right action. The sense of touch is incredibly versatile and offers a broad spectrum for interacting with a computer, so why not explore these possibilities further?

As already mentioned, the field of haptic exploration is extremely diverse, which is why we want to narrow the focus of this thesis to haptic feedback for hands and feet. Besides, we have chosen two application areas as the focus of this thesis. Both application areas (touchscreens and virtual/remote environments) were preferred because we see great potential in these areas. In the next section, the aims and objectives of the thesis are discussed and linked to the five prototypes implemented, which are presented later in the main chapters.

1.2 Aims and Objectives

In line with the above, the aim is to find the right balance, that combines already established interaction approaches and technologies with novel methods and interfaces, focusing on haptic interfaces. This means that we are looking for haptic feedback methods, concepts and technologies that can increase performance, enhance perception and improve user experience. We expect that applying these concepts will shift the focus more to the users, and novel haptic interfaces could be created. Exploring this subject also means that various aspects of several scientific areas are included and required. These scientific areas are outlined in Figure 1.1, which shows the three main research fields (human factors, technology and computer science) in the context of haptic interaction.



Figure 1.1: Classification of the individual areas that are important in this thesis. The three main areas important for researching haptic feedback are human factors, technology and computer science.

How the three research fields (see Figure 1.1) relate to Norman's interaction model [Nor13] to design haptic feedback is explained below. In the haptic feedback principle applied in this thesis (see grey circle in Figure 1.1):

- **Human:** A user is at the beginning. A person wants to achieve a particular objective or experience something specific by interacting with a machine: an action that serves as an input for the system.
- **Computer science:** The user's intention is passed on to the machine through sensors. Then, an algorithm calculates the output state, which is passed to an actuator using a micro-controller.
- **Technology:** An actuator passed on the stimulus to the user. The user adjusts his/her input accordingly through the feedback which in turn serves as new input.

Further disciplines can be found in Figure 1.1. They are just as important and are needed to achieve the objectives of the thesis. Therefore, besides the theoretical interaction model, creating prototypes, their design and evaluation are essential disciplines.

To explore a number of parameters, the two application areas were identified. Both areas will be referred to as *focus areas* in the following and will be presented in the two main chapters Chapter 4 and Chapter 5. The first focus area explores the combination of haptic feedback with touchscreens, as found in smartphones. For this purpose, three novel haptic interfaces are being developed whose feedback spectrum exceeds current research. Each of the three interfaces will follow a unique feedback approach. Hardware concepts will be developed and studied regarding perception, performance and user experience. Haptic feedback in this area focuses on the hands and fingers and attempts to solve common challenges or provide new opportunities.

The second focus area follows the same principle of providing haptic feedback to explore perception, performance and user experience. However, the focus in this case is more on unconventional metaphors - developing concepts and prototypes to show how haptic feedback to the feet can be applied in virtual and remote environments. Foot haptics has great potential; the feet can be used as a reference system for navigation. Moreover, foot-based feedback can enable hands-free interaction so that the users' hands can be used for other purposes.

Both aims and the five objectives are defined by the two focus areas and are reflected in the research questions that have synthesised from the literature review (see Section 2.3). The first three objectives belong to Chapter 4 while the last two are addressed in Chapter 5:

Aim 1: To develop novel haptic interfaces for touchscreens that provide a wide range of feedback stimuli to enhance perception and thus improve performance (see Chapter 4).

Objective 1: Create a motion platform that actuates a rigid touchscreen to explore the limits of haptic perception of geometric shapes (see Section 4.2)

Objective 2: Combine an elastic material with touchscreen interaction to create a novel haptic experience in mid-air (see Section 4.3).

Objective 3: Implement a back-of-device approach for touchscreens, that allows to improve touch accuracy and hence improve occlusion issues (Section 4.4).

Aim 2: Implementing haptic interfaces for the feet to explore self-motion perception and spatial awareness in virtual and remote environments (see Chapter 5).

Objective 4: Build a haptic foot platform that maps spatial data of a remote environment sensed by a telepresence robot to increase spatial awareness (see Section 5.2)

Objective 5: Explore the effect of haptic feedback combinations to the feet on self-motion perception in virtual environments (see Section 5.3)

Based on these five objectives, several results have been achieved, including the design and development of five novel prototypes in each of the focus areas. New aspects of perception will be addressed in the course of evaluating these prototypes. Besides, it is intended to propose various application scenarios that are considered valuable.

1.3 Thesis outline

This thesis consists of six chapters, the bibliography and the appendix. The developed haptic prototype interfaces will be presented in the two main chapters together with the results and evaluations of seven empirical studies. There is also a detailed overview and analysis of the related work, the research methodology, a summary of the results in the conclusion chapter, and finally future work and research directions. Most of the work described in the main chapters has already been published (see List of publications). The structure of the thesis is as follows:

Chapter 1 - Introduction: summarizes the subject area, provides a historical overview and motivates research through the potential for involving further sensory modalities in the interaction, followed by the aims and objectives.

Chapter 2 - Literature Review: provides an summary of the state-of-the-art of all focus areas, resulting in the research questions.

Chapter 3 - Research Methodology: reflects quantitative and qualitative research philosophies and describes research methods that have been applied in this research. This includes e.g. methods originates psychophysic and prototyping as well as experimental design.

Chapter 4 - Haptic Interfaces for Touchscreens: presents three metaphors and prototypes that enable haptic feedback for touchscreens. Studies for evaluation focus on the perception and broadening of system-specific performance parameters. **Chapter 5 - Foot Haptic Interfaces for Virtual and Remote Environments:** studies the potential of feet-based feedback in virtual environments and telepresence systems, mainly targeting the evaluation of the interfaces in terms of self-motion perception and spatial awareness.

Chapter 6 - Conclusion: discusses the findings of the main chapters, summarizes them, and lists all the contributions achieved. Finally, it provides ideas and recommendations for further research.

Chapter 2

Literature Review - Haptic User Interfaces

This chapter provides an overview of haptic user interfaces (HUI) and associated interaction methods and metaphors. It especially illustrates HUI's impact on performance and perception and highlights the need for additional research in our focus areas. Therefore, essential related work is introduced, discussed, presented, and synthesized into research questions.

The literature review starts with the fundamentals of haptic perception and the technical/mechanical principles for generating haptic feedback. This is followed by a literature review that looks into haptic interaction with touchscreens and explores foot-based haptic feedback and interfaces for virtual and remote environments. A final section highlights the thesis's main direction, derived from the literature review, followed by the research questions (see Section 2.3). The questions are the links to the two main chapters of this thesis. The underlying methodology for composing the literature review can be found in Chapter 3.

2.1 Fundamentals

Basic principles of haptic perception and the key technologies for creating HUIs are introduced in this section. Therefore, the first subsection looks at human physiology and explain the sense of touch, which is essential for haptic interaction. After that, the key technologies are presented; this includes the main actuation methods and how capacitive touchscreens work.

2.1.1 Sense of Touch

The sense of touch is probably the most versatile and definitely the largest sense organ of our body. It is distributed over the entire body, unlike the other senses, which are centralized around specific parts of our body [EOEC11]. Haptic perception is often neglected, but it is essential for survival - without the sense of touch, also called the somatosensory system, we would not feel any injuries and would not even be able to walk. Haptics refers to touch and it includes haptic interaction. Our haptic system uses sensory information that comes from mechanoreceptors and thermoreceptors in the skin and mechanoreceptors in muscles, tendons and joints [LK09]. The interpretation and processing of these information allows us, for instance, to interact with object, to use tools or to gesture. The somatosensory system consists of several perceptual channels, including [Gol10]:

- 1. cutaneous senses (skin), which are responsible for perceptions such as touch, pain and temperature that are normally produced by stimulation of the skin
- 2. proprioception, the ability to sense the position of the body and limbs
- 3. kinaesthesia, the ability to sense the movement of the body and limbs

The skin consists of two layers: the visible outer layer is called epidermis and mainly consists of dead skin cells. Below the epidermis is the dermis. These two layers are responsible for our haptic sensations; they contain four receptors that, for example, to perceive pressure, tension, force and vibration. The four receptors are called mechanoreceptors: The Merkel receptor and the Meissner corpuscle, are located close to the skin's surface, while the Ruffini cylinder and Pacinian corpuscle, are located deeper in the skin. Neural connections transmit the signals along the thalamus to the somatosensory cortex where they are processed. According to Goldstein, each of these four mechanoreceptors has a specific task, a particular response behaviour and an assigned frequency (see Figure 2.1) [Gol10]:

- Merkel receptor is a disc-shaped receptor located close to the skin surface, it emits a continuous signal as long as the stimulus is on; the Merkel receptor perceives fine details and frequencies between 0.3-3 Hz.
- Meissner corpuscle is a stacked receptor, which is also located near the skin surface. In comparison to Merkel receptors, it only fires when a stimulus is sensed and when it is removed. They perceive frequencies between 3-40 Hz and allow us to control tools with our hands
- **Ruffini cylinder** consists of multi-branched nerve fibres located within a cylinder. They are located in the dermis, similar to the Merkel receptors they send a continuous signal. They sense the stretching of the skin and respond to stimuli of frequencies between 15-400 Hz.
- **Pacinian corpuscle** is an onion-like receptor that surrounds a nerve fibre. The response behaviour is similar to that of Meissner receptors, they fire to on and off. The Pacinian corpuscle perceives vibrations and fine texture by finger movements; the perceived frequency is between 10-500 Hz.



Figure 2.1: Characteristics of the four mechanoreceptors located in our skin [Gol10].

According to Goldstein et al. the sense of touch can be divided further: there is tactile perception, the perception of vibrations and surface structures and the perception of objects [Gol10]. While tactile feedback is perceived through the cutaneous system, i.e. the perception of surface textures and pressure sensations, force feedback is about the mechanical generation of information perceived by the human kinaesthetic system [EOEC11]. Vibrations such as those generated by vibration motors are referred to as tactile feedback and can perceived to a frequency up to 250 Hz.Not every part of the body reacts equally to haptic stimuli; the different distribution and density of the receptors is responsible for this. Considering our fingers, it can be seen that there are significantly more Merkel receptors in the fingertip than in the palm. This means that the resolution of tactile stimuli is higher at the fingertips than in the palm. The resolution can be measured in a similar way to an eye test, but in this case, two-point discrimination is used. This differs for each region of the body, on the calf or upper arm it is about 45mm, but less than 5mm on the fingers [Wei68]. Mentioned above, our skin can also perceive vibrations, responsible for this are the Pacinian corpuscle, which can react quickly to changes. The perception of textures can be divided into two types: spatial cues and temporal cues. Spatial cues are rather rough structures that can be recognized without active movements just by touching them with the fingertips (for example, Braille). Whereas temporal cues can perceive fine textures, hereby the surface is actively explored. In general, the sensory system perceives sensations such as touch, temperature, and fingers' position. Whereas the motor system enables us to explore objects, e.g. to grasp an object, and the cognitive system combines the information from the two previous systems, resulting in the perception of an object [Gol10].

Alongside the described perception capabilities through the mechanoreceptors, the sense of touch also offers the ability to sense a joint or limb's motion, this called kinaesthesia. Proprioception is often confused with kinaesthesia but refers to something different. While kinaesthesia focuses on movements, proprioception is about the position and orientation of limbs and the body. All of us know where in space our arms, hands or feet are located without actively verifying. Mainly responsible for the two perceptive abilities are so-called muscle spindles. They can perceive an stretching or contraction of the muscles [Gru08]. Another difference between those both is that proprioception is cognitive while kinaesthesia is more behavioural. For example, to walk we need to actively start the behaviour, whereas knowing where which limbs are is more passive.

Another aspect that should be considered is the time it takes for a stimulus to travel to the cortex. For example, in the case of hearing and seeing, the stimulus travels for a certain time outside the body, i.e. physically through space (i.e. 300 000 000 ms for vision vs. 330 ms for hearing), whereas in the case of the sense of touch, the stimulus is transported directly within the body, i.e. nerve system only. In addition, the position of the eyes and ears are at fixed parts of the body, while the skin is distributed over the whole body, which leads to different travel times of a stimulus. Assuming a typical transmission speed of 55 ms, will lead to a difference of about 30 ms between a stimulus perceived by either the toes or the nose [MW11]. Harrar et al. showed that a tactile stimulus have different latencies depending on the distance from the brain [HH05]. A more detailed explanation

of their study is given in the following literature review. Despite the different travelling times of the stimuli, surprisingly the reaction times on the different senses are quite similar. According to Kosinski, the average reaction time on audio is 140 ms to 160 ms, which is faster than the reaction time of sight (180 ms to 200 ms) and touch (155 ms) [Kos10]. The average times can of course vary, this is due to the complexity of the overall system and therefore depends on various other factors such as the spatial separation between the components of the stimuli, the complexity of the stimuli, whether it is speech or not and also the semantic nature of the stimulus [MW11].

The physiology of the sense of touch was introduced in the previous paragraphs. It was shown that the sense of touch is versatile and can, at least through its perceptual prerequisites, support or even enrich the interaction with a computer. In the following, tactile stimuli, vibrations and the perception of movements, such as force feedback or spatial interaction, will be mentioned frequently; all of them are part of the somatosensory perception. This chapter helps to better understand the perception of haptic feedback and to evaluate its quality. In the main chapters, we will present five haptic user interfaces, these ranges from flexible stimuli (see Section 4.3) and force feedback (see Section 4.2). The presented prototypes also focus on different parts of the body, from finger (see Section 4.4) to foot sensations (see Chapter 5).

2.1.2 Actuators

In line with the previous narrative, this subsection will further explore haptic perception and interaction, in detail haptic feedback. Actuators can be considered as the motor system of a machine. The main technologies are introduce in this subsection. We present several systems which are based on different actuation methods. To illustrate the principle of actuating, key concepts are presented: ranging from servo motors to piezo actuators to electromagnetic motion. There are, of course, more approaches, but we would like to concentrate on the most common and applied solutions in our field.

In this thesis **servo motors** were used to create haptic feedback [JKT⁺16], [LGB⁺16], thus this paragraph deals with the basic principle of such motors. In contrast to a regular electric motor, a servo motor can provide a desired position at a desired speed. This high precision and efficiency control is often found in robotics and industrial production. However, it can also be used to provide haptic feedback. A servo motor consists of four main components: a control circuit, a regular motor, a gear assembly and a potentiometer (sensor for positional feedback). The servo is controlled either analogously or digitally [FL14]. For DC motors, the requested speed or position is achieved by the voltage applied.

In the implemented prototypes, all servos were controlled using pulse width modulation (PWM). This requires a pulse width voltage converter, which can convert varying pulse widths to a respective voltage and thus control the speed or position. A potentiometer measures the position, and in case that the current position deviates from the desired position, additional voltage is applied to adjust the motor's position. With this principle, electrical energy can be converted into motion energy, which stimulates the sense of touch.

Piezo actuators have not been used in the implemented prototypes in this thesis, however, they appear from time to time in related work [PM03], [LOM⁺11], so we would like to introduce the technology of piezoelectric actuators. Piezoelectric elementary cells change dimensions when an electric potential is applied. The most common piezo actuators consist of stacks of thin piezoceramic layers that expand when a voltage is applied. Depending on the shape of the piezo element, linear and rotational movement can be achieved. Piezo actuators can be controlled with pulse width modulation similar to servo motors. Looking at an excerpt of the comparison of the two prototypes by Jang et al., it can be seen that piezo actuators can achieve much higher accuracy at similar force while being quieter $[JKT^+16]$.

Another frequently used actuator is a **vibration motor**, which is used to create tactile feedback [GSK⁺20], [CBMB16]. In principle, there are two different types of vibration motors: eccentric rotating mass vibration motors (ERM) and linear resonant actuator (LRA) [PT19]. ERM is a standard DC motor with an off-centered mass attached to a shaft. When the motor is powered, the unbalanced mass creates a centric force. This force is what we perceive as vibrations. While an ERM operates based on a mechanical motor, an LRA moves a mass through an electrical signal. This signal activates an electromagnet that attracts the mass. When the electromagnet is deactivated, it is returned to its original position by a spring. This oscillating movement generates the vibration. Compared to an ERM, a LRA has a direction in which the vibration is emitted, i.e. it is a linear vibration. A key advantage of LRAs over ERMs, especially in terms of haptic feedback, is that they start and stop immediately, whereas ERMs have a certain start and stop phase - this is due to the DC motor. In both cases, vibrations can be described by two parameters: frequency and amplitude. Both parameters are dependent on each other. The frequency indicates how many oscillations (rotary or linear) per second the motor makes. The frequency influences the amplitude, i.e. the offset movement of the motor. Of course, this depends on the mass of the object to which the vibration motor is attached, which is why some manufacturers specify the amplitude of the motor for a 100g object 1 .

¹https://www.precisionmicrodrives.com/vibration-motors/ (last accessed 19th April 2021)

Another technique for generating haptic feedback is using an **electromagnetic actuator**, also known as a solenoid. A solenoid valve can be used to cause a discrete impact or to realize a motor - we focus on a discrete impact [AS94]. A solenoid consists of a copper wire wound into a coil. By applying current to the coil, a magnetic field around the coil is created, forming a south and north pole - also called an electromagnet. To create a discrete impulse, an additional iron core is needed which is placed inside the coil. Whenever the electromagnet is activated, the iron core is moved in one direction, and an impact is created. The higher the voltage, the higher the impact. In this way a stimulus can be created similar to a small knock - we will use this approach later in Section 5.2.

There are other ways to create haptic feedback, for instance, **magnetorheological fluid**. This fluid reacts to magnetism and can change its viscosity. When no magnetism is present, the fluid is liquid; depending on the electromagnet's strength, the fluid changes its firmness from liquid to stiff. Typical reaction times are very fast, less than 2 ms. It can therefore also be used for tactile feedback [JKB10]. The properties of magnetism can also be used individually by using the properties of attraction and repulsion [ZDE16]. However, it should be kept in mind that the force is not linear to the distance. The smaller the distance between the magnets, the stronger the force becomes and is therefore difficult to control at low distances.

Finally, methods for changing friction properties of rigid surfaces are presented. This can be done using two techniques: a) **ultrasonic** and b) **electrovibrations**. With ultrasonic, the surface is brought into vibration, however, the frequencies are in the kHz range and are thus significantly higher as the described vibrations above and can therefore no longer be perceived as vibrations. The main principle of ultrasonic friction screens is that the vibrations alter the contact phase between the finger and the screen [WF95]. This can lead to a slippery or sticky feeling. Electrovibration, in contrast, generates an electrostatic field which influences friction properties [KIP13]. The vibrations and resulting electrostatic fields lead to similar perceptual phenomena as with ultrasonic.

In summary, it can be said that haptic feedback can be generated based on a variety of actuators. Pneumatic and hydraulic actuators have not been considered in this summary, as they are not used frequently in the context of this research. Furthermore, it is difficult to compare the presented actuators in detail, as they often evoke different haptic sensations, which are used in different applications. Parameters such as response time, form factor, weight, power consumption and force should be carefully considered before selecting an actuator. Table 2.1 assigns the different actuators to related work and reflects on further parameters.

2.1.3 Touchscreen Technology

A main theme of this thesis's research is to explore the combination of haptic feedback with touchscreens. It is therefore essential to introduce base technologies of how to sense touch events on a screen. The most well-known techniques are based on a resistive film or on a capacitive mesh to sense a touch event. Further methods, such as touch panels rely on optical sensors or panels based on electromagnetic induction. However, in this subsection, we focus on capacitive touchscreens. These have become established for smartphones and some of the prototypes developed use capacitive screens, for instance, to improve touch accuracy (see Section 4.4).

There are two capacitive methods for determining a finger's position on a touchscreen: surface capacitive and projected capacitive. Both methods are based on a similar principle: Sensors are used to detect small changes in the electrical current generated by touching the screen, which changes electrostatic capacity. Surface capacitive technology is mainly used for large screens and has a limited resolution. A transparent electrode film in the display is necessary to determine the location of an event. Whenever a finger touches this film, the capacitance changes and four electrodes placed at the corners can determine the finger's position. With this technology, it is challenging to calculate more than one touchpoint.

On the other hand, projective capacitive screens can be found in smaller screens such as smartphones [BO10]. This method achieve high-precision multi-touch functionality and high response speed. Similar to surface capacitive technology, electrodes are also used in this case, more precisely transparent electrodes. These electrodes are attached to the four corners, but in a special pattern, a grid, so that far more electrodes can be used. They are additionally divided into x-y layers. An IC chip is used to read the capacitance of the individual electrodes. If a finger is detected, the intensity of the neighbouring electrodes is interpolated to determine the position of the finger. Depending on the configuration and number of electrodes, the resolution can vary, for instance, Kumar et al. presented a deep learning approach that achieves an error offset of 2.35 mm [KMRL19]. In their approach a Synaptics ClearPad 3350 with a 15x27 grid was used.

In this overview, the fundamentals have been presented, on which we will continue to build later. In the following, related work will be presented that specifically deals with haptic interaction in the context of this work, starting with the combination of touchscreens with haptic feedback.

Table	2.1:	Structured	l overview	about	the	main	related	work	in the	contex	t of	this
thesis,	categ	gorised in a	ctuating te	echnolog	gy, fe	eedbac	ek type,	research	h focu	s, body	part	and
applica	ation.											

-					
Reference	Actuator	Feedback type	Research focus	Body part	Application
[JKB10]	Electromagnets	tactile	GUI	finger	Desktop
[WRSS16]	Electrovibration	tactile	GUI/information	finger/hand	Pen-interaction
[CBMB16]	Vibration motor (LRA)	tactile	hand writing	finger/hand	Pen-interaction
[PM03]	Piezoelectric	tactile	GUI/gestures	finger	Touchscreen
[LMBS07]	Piezoelectric	tactile	GUI/gestures	finger/hand	Touchscreen
[YK08]	Piezoelectric	tactile	texture/materials	finger	Pin-display
[PBKW18]	Piezoelectric	tactile	Blind interaction/spatial	finger	Braille display
[PH20]	Piezoelectric	tactile	Multitouch	finger	Touchscreen
[RGOB15]	Pneumatic	tactile	Blind interaction	finger	Braille display
[BCML15]	Servo motor	tactile	Blind interaction	finger	Braille display
[AS94]	Solenoid	tactile	Find targets	finger	Mouse
[TMM ⁺ 12]	Sound	tactile	Walking experience	feet	Desktop
[CSL+13]	Ultrasound	tactile	GUI/information	hand/finger	Mid-air
[BCB07]	Vibration motor (ERM)	tactile	Keyboard typing	finger	Touchscreen
[KLS09]	Vibration motor (ERM)	tactile	GUI/text	finger/hand	Pen-interaction
[NNTS12]	Vibration motor (ERM)	tactile	Presence/immersion	feet	VR
[VBV+12]	Vibration motor (ERM)	tactile	Information	feet	Blind navigation
[TBS13]	Vibration motor (ERM)	tactile	Realism	feet	VR
[CCKB15]	Vibration motor (ERM)	tactile	Eyes-free	finger	Touchscreen
$[GSK^+20]$	Vibration motor (ERM)	tactile	Keyboard typing	finger/wrist	VR
[VTBC20]	Vibration motor (ERM)	tactile	Information	feet	Haptic discrimination
[GMPT13]	Vortex rings	tactile	Entertainment	body	Mid-air
[SGB ⁺ 18]	Magnetic fluid	force feedback	Ground firmness	feet	VR
[IYN01]	Motor	force feedback	Ground properties	feet	VR
[JS19]	Motor (falcon)	force feedback	Manipulation	hand	Desktop
[KIP13]	Servo motor	force feedback	3d features	finger	Touchscreen
[SPB14]	Servo motor	force feedback	3d features/information	finger	VR
[WTC ⁺ 20]	Solenoid valve (air)	force feedback	Ground firmness/Medical	feet	VR
[FLO+13]	Motorized potentiometer	tactile/force	Dynamic Affordances	hand	Pin-display
[NFM ⁺ 19]	Motorized potentiometer	tactile/force	Dynamic Affordances	hand	Pin-display
[JKT ⁺ 16]	Piezoelectric/Servo motor	tactile/force	Dynamic Affordances	finger	Touchscreen
[MNY ⁺ 09]	Servo motor	tactile/force	Textures/heights	finger	Mouse
[BHSO16]	Servo motor	tactile/force	3d shape rendering	finger	VR
[SHS16]	Spandex	elastic	Information	finger	Projection
[BPIH10]	Electrovibration	friction	GUI/information	finger	Touchscreen
[KIP13]	Electrovibration	friction	3d features	finger	Touchscreen
[LOM ⁺ 11]	Piezoelectric	friction	GUI/gestures	finger	Touchscreen
[RVGG17]	Piezoelectric	friction	Texture/object	finger	Touchscreen
[SJM13]	Sheet of Paper	spatial	Information	hand/arm	Projection
[SBC+16]	Bend input	deformation	Proprioceptive feedback	hand/arm	Touchscreen
[PGG ⁺ 12]	Spandex	deformation	Depressing and stretching	hand/arm	Projection

2.2 Research on Haptics

2.2.1 Touchscreen Interaction

Combining haptic interaction with touchscreens is the first focus area. An overview of related work is given in Table 2.1. It shows the variety of different actuator technologies, the research interests and the associated applications. The table highlights the regions of the body that is stimulated by each interface. Furthermore, it introduces various research related to this thesis, which will be introduced and discussed in the following sections. Nevertheless, some approaches do not relate directly to haptic user interfaces but are also important for this thesis.

Haptic feedback and haptic interfaces can be divided into two main categories [Gru08] (see Section 2.1.1). Firstly, there is tactile feedback, a perception that deals with actively touching surfaces and feeling textures. In contrast to tactile feedback, there is force feedback. Force feedback can be perceived by the human kinesthetic system [EOEC11]. In both, the stimuli are mostly produced by movable mechanisms based on actuators, such as motors (see Section 2.1.2). This section starts with a more general part that deals with tactile and haptic prototypes. It is followed by a detailed review of different categories, ranging from pen-based approaches to tangible interfaces and more specific concepts such as flexible materials and back-of-device interaction.

Tactile feedback stimulates the upper skin layers to reproduce surface properties or to differentiate one state from another - it is the system response to touch. Tactile cues can be generated using a variety of different technologies, a representative being vibration motors [BHSO16], linear resonant actuators [GSK⁺20] and electrovibration [WRSS16]. They are often employed to reduce visual or auditory attention, enabling eyes-off interaction, as they operate in a separate perceptual channel. Vibrotactile cues have been shown to improve task performance [BCB07]. However, tactile cues can not only increase interaction performance. The feedback stimuli seem to be excellent for pressing a key during a typing task [GSK⁺20]. Some studies also demonstrated task performance improvements under cognitive load [LMBS07]. Vibrotactile cues have also been used to simulate various surfaces [RVGG17], friction characteristics [LOM⁺11], and rims associated with surface textures [KIP13]. Furthermore, an increase in vibration feedback has been shown to increase perceived softness [VDH14]. Adding audio cues to tactile or haptic feedback has been studied, by looking into which modality works best for widgets [HCBK09] or shapes [CV17]. From a technical point of view, localised feedback on a rigid screen is hard to achieve. However, recent research on localised tactile feedback in the context of multitouch seems to be promising [PH20].





Figure 2.2: Inside and outside view of the haptic mouse prototype of Akamatsu [AS94].

Figure 2.3: Haptic Tabeltop Puck presented by [MNY⁺09].

Rod-like systems have shown promising results for lower finger forces. Akamatsu and Sato developed a mouse-like interface supporting tactile and force feedback [AS94]. For this purpose, a mouse was augmented by solenoids. Tactile stimuli were produced through a pulsing solenoid under the fingertip resting on the mouse button. Force feedback was generated through an embedded solenoid that increases and decreases friction between a mouse and a table (see Figure 2.2). However, to do this, interaction with the prototype has to be performed on a unique magnetic surface. Haptic sensation was combined with visual content. Akamatsu and Sato showed that tactile and force feedback can reduce response time and select targets more efficiently [AS94].

Marquardt et al. enhanced the approach of Akamatsu and Sato [MNY⁺09]. They replaced the solenoids by servo motors (see Figure 2.3). Changing the friction was done by a brake like mechanism using a servo motor pushing a rubber plate against the table. A pressure sensor embedded on top allows Marquardt et al. to position the pin depending on the applied pressure. This prototype enables a set of application scenarios using, on the one hand, different heights as feedback and on the other hand, two pressure modes as input. This state-based interaction shows that both texture and height information can be perceived by a user [MNY⁺09]. A further difference between Akamatsu and Marquardt was the representation of the visual information. In Akamatsu's approach, visual information was perceived through a screen, in which Marquardt's prototype users directly interacted on the visual content through a projection. Both HUIs presented have shown that haptic feedback on the finger either through a servo motor or a solenoid can increase the interaction performance and perception of information.

Many tactile approaches have looked at how to facilitate access to digital information for blind people. Brayda et al. presented another mouse-like device with tactile feedback to assist visually impaired people to improve spatial understanding and mobility skills [BCML15]. Russomanno et al. looked into tactile display technology to improve accessibility for visually impaired people. In their approach, the Braille pins were operated by pneumatic actuators and fluid logics actuation [RGOB15]. Prescher et al. developed a larger tactile display allowing blind people to read Braille and perceive graphical content. The proposed display allows spatial relationships to be perceived [PBKW18].

Combining temperatures with a 6x5 pin tactile matrix integrated into a mouse interface to create more realistic features was proposed by Yang and Know. A Peltier element was used for displaying thermal [YK08]. They investigated whether the perception and reproduction of textures and material properties is possible. Comparing the introduced approaches, diversity of the technology becomes clear: While Akamatsu and Sato realized their actuation with magnets, Marquardt's system relied on servo motors, and Yang and Kwon's prototype is based on the piezoelectric effect.



Figure 2.4: First tactile feedback introduced by Fukumoto and Sugimura [FS01].



Figure 2.5: The improved version of Poupyrev and Maruyama using piezo electric actuation [PM03].

In all systems presented so far, display and interaction were physically decoupled. With the introduction of touchscreens, it became possible to combine display and interaction within a single device. Interacting directly with a display has many advantages, as can be seen in the following sections. Nevertheless, haptic actuators cannot be installed in the
same way as in uncoupled systems; for example, they can be too bulky or occlude content of the display. Fukumoto and Sugimura studied mobile touchscreen interaction with a tactile feedback [FS01]. Their system, Active Click, was the first published tactile interface for mobile touchscreens (see Figure 2.4). The approach is similar to today's smartphones; a vibration motor was mounted directly under the touchscreen. Poupyrev and Maruyama enhanced this approach by replacing the vibration motor with a piezoelectric actuation (see Figure 2.5). They added a soft silicon damper between the touchscreen and the frame to exclude the frame to vibrate [PM03].

Enhancing touchscreen interaction using tactile cues have been explored by Brewster et al. [BCB07] and Hoggan et al. [HKLB08]. Aspects related to latency and perception of tactile buttons were discussed by Kaaresoja et al. [KBL14]. To gain a more detailed understanding of tactile cues in relation to perception and mobile touchscreen technology, the approaches are considered in detail.

While the contributions from Fukumoto and Sugimura as well as from Poupyrev and Maruyama focused on the idea, the system and its hardware, Brewster et al. showed that tactile feedback could significantly improve typing performance for mobile touchscreens [BCB07]. In a study conducted by Brewster et al., subjects were asked to rewrite a text as fast and as accurately as possible. They performed the study two times, once under laboratory conditions and once seated in a train to show that the approach is also valid in real-world environments. In both scenarios, typing with tactile feedback achieved better results. Brewster et al. assumed that the results were achieved by increased awareness through tactile feedback [BCB07]. Additionally, Brewster et al. showed that tactile feedback was significant for mental demand, performance and annoyance using NASA TLX.

Another important aspect associated with the introduction of touchscreen technology on mobile devices and the resulting disappearance of physical buttons was that two essential characteristics are lost - users can no longer feel and hear clicking a real key [HKLB08]. This lack of feedback leads to a different perception of the familiar physical buttons and a more limited usability. While Brewster et al. showed that tactile cues could increase performance in terms of completion time, Hoggan et al. looked into a further issue. They explored how the interaction of tactile, visual and auditory feedback could be used to generate different button styles [HKLB08]. It was found that there is a relation between visual perception and tactile stimuli.

In contrast to Hoggan et al., Kaaresoja et al. were interested in how timing and latency of tactile, audio and visual feedback affect perceived quality [KBL14]. Kaaresoja et al. performed several perception studies, to do so, they varied latency between 0 and 300 ms.

With these timings, feedback (tactile, audio and visual) of a touch event was delayed. To ensure low latency and reliability, a dedicated virtual button simulator was used during the study. As a result, Kaaresoja et al. recommended that tactile feedback latency should be between 5 and 50ms, audio feedback latency between 20 and 70ms, and visual feedback latency between 30 and 85ms [KBL14].

When dealing with touchscreen interaction at present, terms such as pinch-to-zoom or two-finger scrolling have become indispensable - multi-touch gestures are common concept. Therefore, it is not surprising that the combination or enhancement of multi-touch gestures with tactile feedback is also being researched [PH20]. Pantera and Hudin focused on localised tactile feedback in the context of multi-touch interaction. They use the Inverse Filter Method with a grid of 11 piezo actuators to create localised tactile feedback. Their current research concentrates on a calibration and interpolation step and system evaluation. Pantera and Hudin's work again indicates that tactile feedback can improve performance and perception when interacting with a smartphone.

Besides localised feedback [PH20], the interplay between visual and tactile feedback was studied by Harrar and Harris. For this purpose, they explored the simultaneous perception of visual and tactile stimuli [HH05]. Visual information from the immediate environment is perceived independently of location related to the speed of light. Consequently, the reaction time to visual content (light) is constant. In contrast, Harrar and Harris showed that tactile stimuli applied to different body parts result in different reaction times. They found that reaction times to touches varied proportionally with distance from the brain [HH05].

Methods for evaluating the influence of tactile perception using touchscreens are another interesting aspect studied by Alsuradi et al. Whereas currently used methods are based on subjective feedback, [APPE20] has started to evaluate tactile feedback on a touchscreen through EEG. First results seem to be promising. Alsuradi et al. have shown that tactile feedback on a touchscreen can be detected differently in different areas in our brain. Knowing this could help to design haptic feedback more efficient for touchscreens. In a first study, they considered building a haptic model capable of objectively determining and quantitatively assessing the subject's haptic experience.

The usage of friction to provide tactile feedback can also be found in literature [LOM⁺11]. For instance, a friction screen can create the illusion of force, shape and texture on a fingertip when interacting with widgets (see Figure 2.6). While the generation of previous tactile stimuli is based on a physically moving touchscreen, friction forces are based on an electrovibration effect, ultrasonic or high-frequency vibrations generated, for instance, through piezoelectric actuators. TeslaTouch by Bau et al. was the first prototype that





Figure 2.6: Frictional Widgets used for tactile feedback in a file manager [LOM⁺11].

Figure 2.7: Using electrovibration to provide friction between the finger and the touchscreen [KIP13].

uses electrovibration to control the friction between a touch surface and the user's finger [BPIH10]. With the assumption that lateral forces mainly trigger a perception of a physical bump on a sliding finger (see Figure 2.7), Kim et al. explored the rendering of geometrical features [KIP13].

Previous paragraphs have shown how vibration and friction can provide haptic feedback; the next paragraphs focus on force feedback. In comparison to tactile feedback, it is still largely unexplored in the context of touchscreens. Some exceptions: TouchMover shows how a 1D transformation can create a haptic sensation using an actuated stereoscopic display (see Figure 2.8). The system consists of a large screen mounted on a robotic arm [SPB14]. The movements are perpendicular to the screen and dependent on user interaction. As the screen moves, the stereo convergence plane is adjusted according to the user's position. They proposed a 3D physical simulation with force feedback, in which objects displayed on the screen can have, for instance, different force resistance. TouchMover also allows exploring a 3D contour of ridged bodies. It haptically renders object's contour according to the users' 2D finger position [SPB14]. A third application scenario proposed by Sinclair et al., was a volume viewer, in which the user can explore and annotate the different layers by pressing on the screen. In this scenario, the screen moves depending on the location of the layer. Finally, Sinclair et al. showed the integration of GUI elements into the system by implementing buttons with different activation resistance.

Other scientists have studied actuated touchscreens, Kim et al. for example, explored a horizontally mounted touchscreen setup, using a tablet mounted onto a small motion



Figure 2.8: TouchMover a stereoscopic actuated touchscreen with integrate forcing enabling physical simulation [SPB13].



Figure 2.9: A horizontally mounted tablet for haptic interaction with virtual geometries [KHSK14].

platform [KHSK14]. Instead of focusing on user perception, Kim et al. concentrated on the actuation and how to control the screen (see Figure 2.9). Hausberger et al. have also studied actuated touchscreens. With their prototype, they are investigating how haptic feedback can help blind people. They found that additional tactile feedback can help in detecting different sizes of shapes. In their study, they presented the same shape in four different sizes to the user. The task was to find the right size. The detection rate was significantly higher with the support of tactile feedback than without [HTE⁺17].

Actuated touchscreens combined with haptic multi-touch gestures were Nagamatsu et al. investigated [NNTA14]. They used a so-called Stewart platform for actuation (supporting 6DoF). Servo motors were used to operate the platform. Visual content was projected using a projector mounted directly underneath the system. A multi-push approach consisting of two interaction steps was the key contribution of their work. Starting with placing the first finger on the tablet, results in a linear movement downwards. Then, when the second finger hits the screen, the projection screen rotates around the first finger without changing its position, so that both fingers receive independent feedback within on ridged object.

It has been shown how to combine force feedback with a tablet-sized and even larger touchscreen. However, the combination of force feedback with smaller screens, such as those found in smartphones, is mostly unexplored. One reason is that tactile feedback can be generated with comparatively small physical components to stimulate upper layers of the skin. However, when stimulating the human kinesthetic system, more force is required. Consequently, physical components need to be bigger and therefore more challenging to integrate into a mobile system.



Figure 2.10: Second version of the Haptic Edge prototype with 40 pins (2.5mm in pitch), showing a notification example $[JKT^+16]$.



Figure 2.11: Actuation of the tactile pixels at the side of a smartphone is depicted in this figure [JKT⁺16].

A few researchers have looked into actuation at the side of mobile devices. The Haptic Edge Display consists of a linear array of tactile pixels at the side of a smartphone [JKT⁺16]. They introduced two prototype implementations of their Haptic Edge Display, a set of application scenarios, including UI control, tactile display and notifications (see Figure 2.10). Servo motors actuated the first version of the Haptic Edge Display while a piezoelectric actuator was used in the second version (see Figure 2.11). A relatively high number of pins and the limited space for the actuators result in a low output force of the pins. In a psychophysical study, Jang et al. determined the JND of their system. Users were able to perceive a difference of 0.15mm with the Haptic Edge Display.

In this subsection, different approaches, technologies and challenges of haptic feedback combined with touchscreens have been presented. While some research focused on increasing typing performance [BCB07], others looked into the psychophysical evaluation of their interface [JKT⁺16]. In the following subsection, we will continue to focus on interaction with touchscreens, but this time with a pen.

2.2.2 Pen Prototypes

Haptic and tactile cues have also been explored in pen interaction on screens [CBMB16], [LDL⁺04], [WRSS16], including those that provide non-controllable, sideways elastic feedback [HK08]. Pen-based interaction on a screen is directly related to touchscreen interaction. In Section 4.3, a prototype is presented that combines pen interaction with haptic feedback on a screen.

Similar to the previous subsection, where it was shown that haptic could increase performance, it is also possible to use pen-based tactile cues to improve user performance, especially for smaller targets [POM04]. General pen interaction with touchscreens [CST15], or artistic and writing applications [CBMB16], [FPG17] and education [LWLS11] have also been explored. These systems often provide feedback similar to haptic mice [AS94] or pens for spatial interaction [CK17]. In general, haptic feedback is often unfamiliar. It may not be perceived as useful by users during the initial learning phase [LMBS07], which is not only true for pen-based interaction. Finally, there are some hand-held (joystick-scale) devices that provide texture [BHSO16] and variable compliant haptic feedback [TR18], but these are not designed for on-screen interaction.

Tactile pen interaction has been shown to improve task performance on touchscreens. Poupyrev et al., for example, have introduced a tactile pen to improve the user experience when interacting on a touchscreen. In detail, the purposed tactile pen allows feeling displayed content, such as GUI elements or other data [POM04]. Apart from the pen's implementation, the main result was that the performance in a drawing task could be significantly increased. At the same time, Lee et al. have presented Haptic Pen a similar system [LDL+04]. While both explored the interaction with GUI elements, both interaction devices differ in generating the stimuli. Poupyrev et al. used a piezoelectric actuation, and Lee et al. used a solenoid to generate tactile stimuli.



Figure 2.12: Hardware prototype showing all built-in components of the haptic pen introduced by Kyung et al. [KLS09]

The usage of active haptic (pen) interfaces for simulation of different GUI elements is hardly studied. Kyung et al., for instance, proposed a haptic pen for touchscreen interaction. It can provide vibration, an impact and sound. To ensure an interactive wireless system, a built-in battery and a Bluetooth module were integrated. As depicted in Figure 2.12 tactile cues are generated using a typical vibrations motor. However, in comparison to Lee et al. [LDL⁺04] the impact is created by a linear vibration instead of a solenoid. With their designed feedback metaphors, they explored the performance when interacting with GUI elements. Kyung et al. found an interesting correlation between completion time and haptic feedback. In all three GUI tasks (button clicking, icon/file manipulation and text handling), users could solve the task faster with haptic feedback than without. Kyung et al. stated that the impact generator has an advantage of controllability compared to vibrations [KLS09].

Wang et al. introduced the EV-Pen. A pen that uses similar technology as introduced by TeslaTouch [BPIH10]. The key contributions of the EV-Pen are precise interaction and a pen-on-paper feeling [WRSS16]. Findings of the conducted studies, precisely a steering task and a tracing task, showed that users have higher precision with the EV-Pen than the other pen interfaces. Wang et al. argue that the higher accuracy of the EV-Pen is that the resistance occurs directly at the pen's tip and not inside the pen. As the vibration takes place inside the pen, smooth movements are complicated because vibrations hinder a precise control of the pen [WRSS16].

Another pen-based approach that generates haptic feedback is FlexStylus [FPG17]. Fellion et al. have developed a flexible pen; all deformations of the pen are detected by a bending sensor and used as input modality. Various interaction metaphors have been implemented, which are based on rotational and lateral deformations. Compared to the interfaces already presented, Fellion et al.'s approach does not use an active actuator, but rather an individual, flexible material.

As mentioned earlier, there is a strong interest in improving handwriting applications. Compared to writing on a paper, traditional handwriting on touchscreens does not offer the physical characteristics known from real experience. As a result, it leads to less acceptance. Cho et al. introduced RealPen, a with a linear actuator to provide a realistic writing sensation (see Figure 2.13). RealPen can create real tactile sensations. A sensing approach of a real pen on paper writing is used to recreate auditory-tactile feedback for different pen-tips. In a study, Cho et al. were able to show that RealPen could improve paper writing realism compared to other pen interfaces.



Figure 2.13: RealPen, a pen enhanced with a linear resonant actuator to recreate paper-like surface characteristics for more realistic writing sensations [CBMB16].

To summarize, most important pen interfaces providing haptic feedback were covered in this section [CBMB16], [WRSS16]. In most cases, the pens were equipped with additional sensors and actuators to generate tactile sensations [KLS09]. An important aspect of pen interaction seems to replicate real-world pen properties as closely as possible [CBMB16]. Back-of-device interaction is another modality of interacting with touchscreens. For this reason, the next section will focus on such interfaces.

2.2.3 Back-of-Device Interaction



Figure 2.14: Three main style to hold a smartphone. 95% of all users preferred to use the thumb for interaction, only 2% of the observed users preferred using the index finger [GDT⁺12].

Interaction with mobile devices can be performed using different modalities. In recent years, speech has become popular as an input modality for smartphones. However, most users interact with a smartphone using touch. Gold et al. observed that the two main styles of interacting with a smartphone are either typing with one or two thumbs (see Figure 2.14). Only 2% preferred using the leading hand's index finger while holding the

device with the other hand $[GDT^+12]$. Additional studies have shown that 95% of all participants tend to use the thumb as touch input. Generally, the front touchscreen is used for interaction. But what about the back? In Section 4.4, a prototype will be presented that enables haptic feedback on the back of a smartphone. Therefore this subsection discusses Back-of-Device (BoD) interaction.

Integrating the back opens up new opportunities to interact with mobile touchscreens and devices. For example, interaction is no longer limited to the thumbs or fingers on the front screen. It allows the use of more fingers simultaneously [WLF⁺06]. Currently, the fingers on the back are mainly used to stabilize the grip. However, when using the back for interaction screen content will not be occluded by the fingers [CDVB17] and allows, therefore a more precise input and better visibility. BoD interaction can also be used to enter private information such as authentication data like a pattern, a code or a fingerprint [DvZN⁺13] since other people cannot easily observe the back.

Touchscreen interaction suffers from limitations as the fingers can easily occlude screen content. To overcome this limitation, Wigdor et al. presented a mobile system called LucidTouch. Further prototypes have also addressed occlusion aspects, for instance, "under the table" interaction, looked into semantic associations [WLF⁺06], the HybridTouch system [SH06] explored the manipulation on the front, and the back of a personal digital assistant (PDA), or the Behind Touch system [HMT03] studied text input at the back. The performance of dual touch input was explored by Iwabuchi et al., who introduces LimpiDual, a transparent display system with two touchscreens, one at the front and one at the back [IKN08]. Apart from the primary goal of avoiding finger occlusion of screen content, Iwabuchi et al. proposed a set of application scenarios, including the manipulation of layer information and 3D objects.

Besides occlusion issues, researchers have looked into further aspects of BoD interaction, including gestural interaction [WSR14], [XHW13] and the influence of finger agility. Therefore, Le et al. looked into the development and design of BoD prototypes and applications and how to ensure reachability of all interaction elements, a stable grip and ergonomic comfort. Le et al. explored fingers' range and estimated comfortable areas per finger for one-handed smartphone interaction on the back-of-device [LMBH18]. A motion capturing system was utilized to record finger motions on the back of a smartphone. Results showed an effect of hand sizes and grip. They additionally displayed comfortably reachable areas for all finger. Shimon et al. explored back-of-the-device gestures. In their work, they have introduced a mapping of known touchscreen gestures to the back of a device (see Figure 2.15). The conducted user study results showed that users often mimicked gestures that they already knew from the front screen [SMSJ⁺15].



Figure 2.15: An example of an one-handed mapping of common gestures to the back-of-the-device by [SMSJ⁺15].

Wobbrock et al. explored hand postures in front and back interaction scenarios. Therefore three performance studies were conducted [WMA08]. All studies were performed: one and two-handed using either the index finger or the thumbs. Wobbrock's assumption that the thumb is well suited for interaction on the front turned out to be wrong, as Wobbrock et al. found that the index finger works well on the front and back of the device.

Gold et al. came to a different conclusion, they found, that most users prefer to use the thumbs [GDT⁺12]. Nevertheless, this is an excellent example of how interaction metaphors can change in a short term of time. It seems that it is no longer feasible to interact with the index finger on the front screen using one hand as proposed by Wobbrock et al. Interaction metaphors also evolve with the technology build-in modern devices, such as displays and sensors. Comparing the postures introduced by Wobbrock et al. it becomes evident that interacting with a smartphone has progressed.

Besides postures, combining text input with back-of-device interaction was researched by Schoenleben and Oulasvirta [SO13]. They compared typing performance on a conventional physical keyboard with their BoD folded keyboard approach and found that works very well with the Dvorak standard keyboard. The typing speed on a standard keyboard (QWERTY) layout was poor in comparison. According to Schoenleben und Oulasvirta this is due to QWERTY requires longer finger movement paths and flexion than when using a Dvorak standard keyboard (DKS) layout.

BoD interaction also supports creating smaller interactive display devices, which was discussed by Baudisch and Chu [BC09]. They found that BoD touch input is not affected by display size. In contrast, front-screen touch input was affected by display size; a smaller display size significantly reduced touch precision. This highlights that BoD interaction improves performance in several ways, from touch precision to 3D manipulation to error reduction.

Authentication with a fingerprint scanner on the rear is a conventional method and already built in many smartphones available on the market². In this respect, De Luca et al. proposed a system where users draw shapes on the back-of-the-device to unlock the device [DvZN⁺13]. Their main focus was to avoid spying on touchscreen interaction (authentication patters) while maintaining usability and performance, which was proven by a user study.

Up to here, this subsection has shown that BoD interaction mainly refers to an input modality, i.e. the user controls the smartphone so that the communication channel passes from a human to a computer. Feedback, e.g. output and especially haptic feedback, on the backside was rarely found in literature, even though it is suitable.

Nevertheless, a few representatives also use the back as the output channel. Corsten et al.'s approach, HaptiCase, explored eyes-free absolute touch using tactile landmarks on the back [CCKB15]. They studied how exactly proprioceptive pinching of fingers and thumbs can be supported using tactile feedback on the back. In detail, they looked into indirect tapping on a visible mirrored screen. Users had to select specific regions on a distant screen using a touchscreen eyes-off with their thumbs, supported by tactile grids on the back. Different configurations of the tactile grids were analysed. Corsten et al. observed that pinch accuracy also depends on the finger position, e.g. the pinch accuracy in the middle of the smartphone is more accurate than at the border. Complementary non-visual interaction had explored previously, e.g. [WYLS13], showing the potential of using such methods for the in-the-pocket operation of a smartphone for visually impaired people [KJWL09].

Most of the BoD research relates to improving performance parameters, like increasing touch accuracy or visibility. In contrast, Feelsleeve looks at how to increase the reading experience and how to make it more exciting and memorable for children [YILK15]. Feelsleeve uses a kind of gloves attached to the back of a tablet. Vibration motors are integrated to support the displayed story with tactile stimuli. Yannier et al. conducted a user study with 44 participants and showed that haptic sensations while reading have the potential to improve children's experience and make it more memorable.

A detailed description of the state-of-the-art of BoD methods and interfaces were presented in this section. It was shown that BoD methods are mainly used as input modality [SO13], [DvZN⁺13]. In addition, it was also highlighted that BoD interaction has great potential to overcome certain challenges, such as eyes-free navigation [WRSS16] and interacting with small objects [BC09]. Most approaches looked into performance [SO13], [CCKB15]

²Manufactures, like Samsung and Huawei, have built-in fingerprint scanners on the back of the device.

issues while only a few addressed perceptional aspects [YILK15]. The next section is slightly more detached from touchscreen interaction, but it will present work that is also important in the context of this thesis.

2.2.4 Tangible and Shape-Shifting User Interfaces

Tangible and Shape-shifting User Interfaces address haptic sensations by using physical properties, and, according to Rassmussen et al., they have three functional goals [RPPH12]: they can communicate information over their shapes, they can offer possibilities for actions, which is also known as dynamic affordance, and they can provide haptic feedback. Especially the latter makes this kind of interfaces interesting for this thesis. Before discussing relevant research in this area, a brief description of two concepts is provided: 1) Tangible User Interfaces (TUI) use physical objects to represent and manipulate digital data [SH09]. 2) Shape-shifting interface (SSI) use a physical change of shape as input or output [RPPH12]. Both concepts include physical objects, either focusing on manipulating data or changing the shape to interact with digital data.

Ishii and Ulmen published an initial idea of Tangible User Interfaces, and defined the properties of TUIs as follows: it should augment the real world by coupling digital information to everyday objects and environments [IU97]. Fishkin, in contrast, introduced a taxonomy in which TUIs can be classified based on their embodiment: full, nearby, environmental and distant [Fis04]. Two further areas in the field of TUIs that have become increasingly popular in the last years are wearable and mobile textile interfaces [SKBH18] and pin-based display systems [NFM⁺19], [FLO⁺13]. Holmquist et al. looked at TUI's future trends and consider that they will develop in two directions: 1) in the material domain, Human Material Interaction (HMI) and 2) in the Mixed Reality domain, Extended Reality (XR) [HZB⁺19].

Magnetorheological (MR) fluid provides a way to generate tactile and force feedback using a shape-shifting approach. Jansen et al. introduced MudPad. They showed how to use the properties of MR, low reaction time, for example, to create a haptic overlay for a touchscreen [JKB10]. Jansen et al. discuss some limitations of the idea, one of which is that MR fluid can only be used with top projections as MR fluid is opaque. Moreover, the use of a capacitive touchscreen in combination with MR fluid is not applicable as iron particles interfere the capacitive sensing [JKB10]. Finally, the resolution of the system is discussed, it depends primarily on the size and number of the magnets. Jansen et al. used a resolution of 12x7 matrix using 2cm in diameter magnets.





Figure 2.16: A reconfigureable tactile shape-changing interface by Strasnick et al. exploring dynamic affordances for mobile device applications [SYT⁺17].

Figure 2.17: A textile tactile display, allows active and passive adjustable feedback using a 7x7 matrix [BPM09].

Before looking at the work of Strasnick et al., a brief explanation of what dynamic affordances is offered. The term affordance came from the perceptual psychology [Gib80] and was later associated with the usability of objects [Nor13]. Affordance means that a well-designed object/product allows the user to know what to do with the object/product without overthinking, just being intuitive. The object has explained itself to the users by its shape, for example. Dynamic affordance means that an object can change its shape, which affects how a user wants to use it [RPPH12].

Strasnick et al. investigated reconfigurable tactile elements to change the affordance for physical dynamic in and output $[SYT^+17]$. For the actuation of the dynamic elements, Strasnick et al. focus on magnetic actuation. They introduced two different prototypes that can change affordance. Both can dynamically place haptic feedback elements at the side of a smartphone, similar to the Haptic Edge Display $[JKT^+16]$. Strasnick et al. introduce four applications for their interface (see Figure 2.16), ranging from configurable physical buttons, haptic notifications, gaming controller to interactive callipers to measure the external world are explored $[SYT^+17]$.

A further representative is BubbleWrap [BPM09], a haptic display prototype (see Figure 2.17). Similar to Strasnick et al., Bau et al. used a magnetic approach for actuation. They combined a permanent magnet, a coil and a flexible wrapping material allowing a dynamic expanding and contracting of the coil depending on the current flow. BubbleWrap arranges the developed actuators in a 7x7 matrix. Each pin can expand up to a maximum



Figure 2.18: Pin-based force feedback display inFORM, showing interaction with a ball. [FLO⁺13].



Figure 2.19: InFORCE an extentsion of the inFORM display, each pin moves induvidually based on the applied force [NFM⁺19].

of 10 millimetres. BubbleWrap provides active and passive haptic feedback. A conducted user study showed that users could quickly and reliably discriminate among three levels of firmness [BPM09].

Pin-based display systems generally consist of pins arranged in a matrix. The pins can have different profiles, from square to circular. The resolution can be specified similarly to a conventional display, in which the resolution is the number of pixels in width and height direction. However, resolution and element density are lower. Another difference between visual displays and pin-based display systems is that the pins are movable, similar to a 3D nail board. In this way, arbitrary shapes can be dynamically generated. Follmer et al. proposed in FORM, a pin-based display system, which consists of 30×30 motorized white polystyrene pins, squared with a side-length of about 3 mm (see Figure 2.18). A nylon rod inside a plastic housing connects the pins to a motorized slide potentiometer, enabling bi-directional force [FLO⁺13]. Compared to other systems, the pins move relatively slowly, for instance, in Follmer et al.s system, the pins reach a speed of about 1 m/s. To add colour to the white pins, a projector-camera unit was used. The interface opens up a set of novel application scenarios reaching from simple button or handle interfaces up to a painting tool or to actuate other devices $[FLO^{+}13]$. Nakagaki et al. extended this approach to achieve better haptic quality by introducing individual force feedback for each pin (see Figure 2.19) so that users can feel material stiffness through pressing the pins $[NFM^+19]$.

Follmer et al. introduced Jamming Interfaces, that rely on malleable media. According to Follmer et al., jamming techniques can change the shape state. For instance, a user deforms an object, and the object maintains the deformation - this state is called jammed.



Figure 2.20: Jamming interfaces: the technique behind jamming (left); tunable stiffness for deformable interfaces on tabletops (right) [FLO⁺¹²].

Unjammed returns the shape to an initial state which is either influenced by gravity or mechanically caused [FLO⁺12]. Follmer et al. proposed a hydraulic and a pneumatic system, while the hydraulic approach allows an optical sensing of the surface, the pneumatic approach is not able to sense the surface due to the used material. Using this jamming technique, Follmer et al. present a sculpting application, a tabletop lens and a mobile interface approach (see Figure 2.20). The mobile approach differs in the sensing technique, while previously a visual approach has been used, namely a structured light approach, the mobile version of which uses a capacitive membrane to capture the deformed surface. Using capacitive sensing allows in the proposed implementation a resolution of 9x9 and was mounted on the back of a tablet for interaction.

A variety of different prototypes and approaches have been presented in this subsection. Many of the interfaces presented served as a source of ideas [NFM⁺19], [SYT⁺17]. In the following, flexible and elastic interaction techniques as well as techniques for providing haptic feedback in mid-air are presented.

2.2.5 Flexible and Mid-air User Interfaces

In contrast to flexible interfaces, which use material properties, mid-air interaction targets the space around or above a device to provide haptic feedback. In Section 4.3, a prototype is presented that combines both properties. It shows how haptic feedback above a touchscreen, i.e. in mid-air, can be provided using an elastic material. Therefore, related work in this area will be discussed in the following paragraphs.

The usage of elastic surfaces relates to organic user interfaces [HV08]. Researchers studied general interaction metaphors with deformable materials [LKJ⁺10] the use of electromagnets [JKB10] or bendable materials, including paper(-like) [GSV08], [SJM13]



Figure 2.21: Steimle et al. introduced an approach that combines a mobile projection system with a deformation of a plain white sheet of paper to interact with [SJM13].



Figure 2.22: Depthtouch: a Microsoft Kinect is used to detected deformations of an elastic material to visualize informations according to the deformations [PGG⁺12].

or silicon(-like) materials [HGL14], and simulating variable compliance on rigid screens [Kil10].

Various paper-based interfaces like a tracked paper [HVA⁺05] used as display or a flexible input device for foldable interaction [GSV08] are one of the predecessors of Flexpad [SJM13]. Steimle et al. uses spatial metaphors to represent, for instance, layers of a volume (see Figure 2.21). Gallant et al. described their interface as foldable user interfaces (FUI) and foldable input device (FID). Both recognized the manifold possibilities of interaction with just a sheet of paper, reaching from squeezing, and corner bending, and folding, to flipping.

In recent years, the use of new display technologies has also become increasingly widespread. For example, flexible and deformable displays are being developed - with no need for projections [SBC⁺16], [GZBV16]. Most mobile, flexible displays consists of so-called flexible organic light-emitting diodes (FOLED). FOLED displays are a extension of an OLED display. All OLEDs are mounted on a flexible material. An advantage of FOLED displays compared to projection-based systems is that there is no need to track the displays to rectify the visualization. However, tracked data can be used to sense user interaction. Otherwise, additional sensors are required to detect any bending of the device to handle user input.

Elastic variable compliant surfaces have also been explored. Accordingly, Peschke et al. proposed a flexible tabletop using a back-projection (see Figure 2.22). Their DepthTouch system consists of a highly elastic material (spandex, elastane) [PGG⁺12]. While the

interface can be deformed, it also provides feedback based on the natural properties of the surface, where pressure and tension can result in a more natural interaction. An exploratory application scenario in which virtual spheres roll into the created valleys shows validity of the prototype. FlexiWall introduced by Müller et al. looked into the exploration of layered data using flexible displays. Müller et al. concentrated on multi-layer image data that were displayed according to the depth of the deformation [MKG⁺14]. A further aspect of the proposed flexible projection surface is that the material can also be pinched and pulled out.

Sahoo et al. proposed a further elastic screen which can provide self-actuated deformations and tactile feedback. The screen, called TableHop, consists of a 3x3 grid of transparent electrodes used for vibration and deformations (up to \pm 5 mm) and a back-projection. It measured 30x40 cm [SHS16]. In contrast to [PGG⁺12] and [MKG⁺14], Sahoo et al. evaluated their prototype. This includes a tracked polystyrene bead and a mounted highspeed camera to measure the amplitude and the frequency of the deformation/vibration. For all of the three prototypes above, no details were given of the perceived resolution or shapes that can be felt.

In the introduced interfaces, deformations were not always well-controllable [PGG⁺12]. Other kinds of materials, including sand and fluids, have been studied, but are also challenging to control [FLO⁺12]. Jamming [FLO⁺12] and tabletop [SHS16] interfaces could provide flexible and even shape-shifting properties, for usage in mobile devices [FLO⁺12]. Follmer et al. illustrate the potential of haptic feedback through controllable stiffness. Similar feedback has been explored in mechanical actuation for hand-held squeezing [GCHP10] and mouse-like interfaces [MNY⁺09], as well as medical skin simulation [KTC⁺12]. Furthermore, multiple bendable screens that provide "flexible" feedback have been presented [SBC⁺16].

Beside interaction with flexible materials to provide haptic feedback, the interaction space above displays, the "hover space", has also been explored by many researchers. Medusa, for example, is a Microsoft Surface augmented with 138 low-cost proximity sensors to sense the user's hand, arm and body [AGWF11]. This prototype introduces a novel sensing technique which enables novel pre-input metaphors and visual feedback on detected presence. For instance, by detecting the interacting arm (left/right), different functionalities can be assigned per arm (see Figure 2.23). Sahdev et al. explored the same problem. They describe a capacitive sensor approach that can distinguish between the hands of a single or two users [SFJ⁺17].

Similar to Annett et al., recent research on two-handed interaction with touchscreens explored the non-dominate hand's anticipation to assign functionalities to the users' hands



Figure 2.23: Medusa: a tabletop that can sense three levels of a users to support spatial interaction around the device [AGWF11].



Figure 2.24: A highly sensitive capacitive touchscreen enables finger sensing above and around a smartphone [HBH⁺16].

dynamically [WFK⁺19]. Besides the recognition of arm and hands for mid-air interaction, gestures were also investigated. Cheung et al. presented a set of interaction guides on how to adapt and expand common interaction metaphors like mouse interaction into the above the display space. Their research includes considerations about proximity range, feedback location for the hover space and interaction techniques for hover devices [CHSD12].

Mid-air gestures for smartphones have also been developed [HBH⁺16]. The motivation was to extend the constrained vocabulary of 2D touch interaction to a third dimension, the hover space, in order to facilitate continuous 3D interaction with a 2D touchscreen (see Figure 2.24. A highly sensitive self-capacitive touchscreen enables looking into the design space of pre-touch interaction, including anticipation plus system feedback. The system can assign different touch functionalities according to hover trajectory tracking. The hover space, in combination with the 2D touchscreen, can additionally be used for combined gesture detection [HBH⁺16]. Mid-air interaction techniques are not only being explored for smartphones, but can even be found in stereoscopic applications, e.g. for manipulation tasks [LAB⁺15]. According to Lubos et al., recent sensing technologies such as Leap Motion and Microsoft Kinect support the evolution of mid-air and above-the-display interaction.

Previously mentioned mid-air interaction research focused on the design of metaphors in hover space and on sensor technologies. Methods for haptic feedback in non-contact situations, i.e. in the air, have not yet been discussed. Nonetheless, several techniques provide haptic feedback in mid-air, ranging from ultrasound to systems based on magnetic



Figure 2.25: UltraHaptics: a prototype which renders haptic feedback using ultrasound above a screen [CSL⁺13].



Figure 2.26: Exemplary representation of a magnetic-field based prototype, including a glove with magnetic elements and an array of electromagnets [ZDE16].

fields. In the following, selected haptic feedback technologies in mid-air are presented and discussed.

Carter et al. introduced a haptic feedback using ultrasound [CSL⁺13]. Their system, called UltraHaptics, uses a 2D array of ultrasound transducers to create tactile focus points in hover space. A leap motion tracks the user's hand and a screen that consists of a front projection and an acoustically transparent material. The acoustically transparent material allows to focus the ultrasound directly on the users' hands. In the system, multiple tactile focal points could be generated that reach their maximum intensity at a distance of 20cm [CSL⁺13]. For example, GUI elements can be enhanced with tactile stimuli without touching the display. An application scenario of UltraHaptics is depicted in Figure 2.25.

In contrast to ultrasound-based systems, Gupta et al. proposed a prototype that uses socalled air vortex rings[GMPT13]. Their system was not explicitly designed for interaction in the hover space. It focused on non-contact haptic feedback. However, this also enables interaction above a system, i.e. a display. In their research, they introduced a prototype air vortex generator and its physical and perceptional resolution. In an experiment, Gupta et al. showed that the physical resolution ranges from 10 cm to 250 cm with an accuracy from 5 cm to 10 cm. Eight body locations were chosen to estimate the perceptual resolution. After each stimulus (condition), users were asked which region was stimulated - resulting in a resolution from under 10 cm in diameter at each location.

A further method to provide haptic feedback in mid-air includes magnetic fields (see Figure 2.26). The approach is similar for all magnetic field methods. Typically, a magnetic

object, such as a magnetic ring [MS05], is attached to the user's hand. For the display an array of electromagnets is used, the intensity of each single magnet being controlled by varying voltage. Within this mechanism, the magnet attached to the user can either be pushed off and pulled in, to create force event (haptic feedback) above the display. In related work, Zhang et al. introduced an advanced system that enables magnetic force feedback rendering in the air. The system can render three levels of resistance, reaching from very firm and firm to soft in an interaction area from 0cm up to 5.2cm above the display [ZDE16].

When comparing the above non-contact display systems, the feedback ranges varies. While air-based systems can stimulate up to 250 cm, using ultrasound works in an up to 20 cm interaction area. The introduced magnetic field-based approach can provide feedback within a range of 5.2 cm. Besides intensity, different approaches have been used to determine the resolution of the prototypes. Whereas Carter et al. conducted a differentiation study in which users had to distinguish two neighbouring stimuli, Gupta et al. gave the resolution with a mean targeting error.

The aforementioned feedback mechanisms are based on contactless technologies. Instead, Gupta et al. investigated different mid-air feedback metaphors in typing tasks. Unlike their predecessors, Gupta uses vibrating elements that are attached to specific regions of the hand, such as the fingertip or wrist [GSK⁺20].

Besides the feedback ranges, resolution also varies. For instance, the system introduced by Carter et al. achieved a resolution of 2cm and the prototype proposed by Gupta et al. a resolution of 10cm. Zhang did not provide more detailed information about the resolution except for the use of a 15x15 array of magnetic elements. Since resulting resolutions were measured at different distances and different methods, it is challenging to compare them. However, the introduced research shows that hover space, haptic feedback is possible. The combination with visual information is rather rare in this field [ZDE16], some approaches use front or top projection [CSL⁺13], however, they suffer from occlusions.

In summary, while interaction with elastic and flexible materials focused mainly on hands [PGG⁺12], [SJM13], mid-air techniques are no longer specifically tied to the hands, rather can affect the whole body [GMPT13]. This is intended as a transition to the second focus, the next three sections will therefore present haptic interfaces that stimulate the feet, also known as foot haptics.

2.2.6 Virtual Reality

In previous subsections, the focus was on haptic interfaces that mainly focused on providing feedback to the hands. As mentioned in Chapter 1, the main chapters not only deal with the augmentation of touchscreens, but also introduce haptic interfaces for virtual environments and remote environments. We see great potential in stimulating the feet with haptic feedback, firstly because the feet are not used in most applications in these areas and are therefore "free", and secondly because the feet are also stimulated in normal life, for example when walking. With haptic feedback to the feet, a more realistic feeling could be evoked. In Chapter 5, two research prototypes will be presented that looked into performance aspects, as well as on how to increase spatial awareness and self-motion perception. Therefore, in the next three sections, haptic feedback to the feet in the particular fields will be reviewed and related research will be presented.

As already stated in the Fundamentals (see Section 2.1), mechanoreceptors are not equally distributed across the entire body, which of course also applies to the feet. Since there are comparatively many mechanoreceptors in the feet, but still fewer than in the hands, they still offer a good opportunity for haptic feedback. In order to show which locations of the sole are appropriate for tactile stimuli, different frequencies of vibration at different locations were tested [GG11]. In an enhanced study, Jammes et al. showed that the sensitivity of vibrations on the soles decreases with age. In this case, vibration detection threshold was determined at varying frequencies. The main result was that older people (53-67 years) could detect higher frequencies (> 150 Hz) less well compared to younger participants (20-34 years) [JGF⁺16].

When looking specifically at virtual reality research, it can be seen that a research goal of virtual reality is to make experiences as realistic as possible. In order to achieve this and make VR as authentic as possible, it is worth considering the feet, as they are also part of our perceptual system. For this reason, related work that explores foot haptics in VR presented below. Foot haptics can be used to elicit ground texture cues, partly also in combination with audio [NNTS12], [PFC⁺10]. The perception in virtual environments with tactile foot stimulation was subjectively more realistic than without haptics [NNTS12]. For this reason, in Chapter 5 two novel haptic interfaces will be introduced that provide a wider range of feedback.

Furthermore, plantar cutaneous vibration feedback (the stimulation of the foot sole) can be sufficient to elicit a walking experience [TBS13]. In this respect, Turchet et al. proposed a system controlled by a micro-controller, and studied how walking experience in virtual environments can be increased using foot simulation (see Figure 2.27). They performed a



Figure 2.27: The back of the shoe interface shows two vibrations motors and a force resistance sensor [TBS13].

set of studies with the result that haptic feedback can increase the realism of walking in non-interactive (seated) and interactive (walked) virtual environments. In both scenarios, they used the same interface. A pair of sandals extended with four vibration motors were used as interface (two in front (toes) and two behind (heel)). Two additional force resistance sensors were embedded per shoe to detect the steps (again at the front and the back). Three different soil conditions were tested: snow, sand and forest ground [TBS13]. Subjects participating in the experiment felt that the haptic stimulation on the sole worked best for the snow scenario.

A further approach to enhance walking experience was proposed by Terziman et al., who used visual and vibration patterns. A low-frequency loudspeaker was used instead of vibration motors [TBS13] to create vibrations patterns. The patterns simulate either one or two ground contacts. One ground contact corresponds to a heel strike, and two contacts reflects a heel strike and then a toe strike, similar to natural walking [TMM⁺12]. It was found that users preferred one contact point, and that more complex stimuli are perceived as less natural. Users also prefer simulating the contact point with the ground more than the texture of the soil. In general, Terziman et al. showed that multisensory feedback does increase walking sensation. A further, more recent approach is RealWalk, the proposed interface consists of a pair of actuated shoes. The interface is designed to designed to create realistic sensations of ground surface deformation through MR fluid [SGB⁺18]. It can create a variety of ground material deformation in VR, such as snow, mud, and dry sand by modulating the viscosity of the MR fluid in the shoe sole. Compared to the prototypes based on vibrations, RealWalk is able to render not the texture but the surface firmness haptically.



Figure 2.28: Foot-based navigation interface with a regular grid with 16 vibration motors and the distribution of mechanoreceptors at the sole of the foot $[VBV^+12]$.

Similar to Terziman et al. Feng et al. look into the influence of a multisensory design space to enhance the walking experience in virtual environments. They introduced a set of multisensory feedback, including wind, vibration and audio. They showed that foot stimulation could help maintain spatial orientation, and it can increase user experience. Furthermore, non-directional tactile cues (e.g., floor vibrations) have been shown to provide some self-motion perception [FDL16]. Auditory feedback, footstep sounds have also been used to elicit self-motion sensations [FDL16], as the frequency of steps provides some information about how fast the user is moving.

Recent research follows the same line, Wang et al's interface reproduces both roughly sloping terrain and subtle sensations when stepping on small objects or uneven surfaces [WTC⁺20]. With their interface, Wang et al. pursue two goals. Firstly, they want to make the walking experience as realistic as possible, and secondly, they want their system to provide a training environment for people suffering from Parinkson's disease. Similar to the Gaitmaster [IYN01], which will be presented later, the SmartShoe can map inclinations up to 10°. This is done through a set of chambers, each chamber is controlled by a solenoid valve. For example, a left inclination can be achieved by closing all valves on the left side and opening those on the right side. In this way, the air in the left chambers is maintained while it flows out of the right chambers. In their application they use a treadmill interface to walk realistically in place. In a study Wang et al. looked at spatiotemporal gait properties and kinematics which are important for training in this area.

Some studies specifically looked into navigational cues, for instance, "turn right", by deploying a regular dense grid of 16 vibrations motors under the mid-foot [VBV⁺12]. All vibration motors support a frequency between 10 and 55Hz. The metatarsus (inner foot sole) was chosen to provide feedback because of the high density of mechanoreceptors Figure 2.28. Accordingly, Velazquez et al. conducted three user studies to demonstrate that foot-based stimulation is suitable for navigation. Users have to detect directional

cues, shapes and patterns. Both blind and sighted people took part in the study. Results showed that there are signals that distinguishable and therefore, suitable for navigation. In particular, results additionally suggest that blind people can perceive tactile cues on the sole of their feet better than sighted people [VBV⁺12].

Vibrations have also been used for providing collision feedback [BB10a]. In contrast to previously mentioned feet-based interfaces which were included into a shoe or a sandal, Blom and Beckhaus integrated haptic feedback directly into the floor. The "sound floor" consists of bass-reflex speakers that are directly mounted to the floor surface of an L-shape display system. Audio and haptic feedback were studied - each in a rumble and a thump mode. The conditions: no feedback, automatic stopping and visual feedback were taken into account. In a questionnaire, users specified the fastest, most realistic, and the condition that best representing solid walls. According to them, floor thump is considered to be the most practical method, and also the method for representing the most solid walls. No feedback and automatic stopping were subjectively perceived as the fastest methods. However, objective data showed that this was only the case for one user [BB10a].

It was shown that foot-based tactile feedback can be improve navigation and collision in VR, Vyas et al. showed that tactile feedback on the toes enables more complex information to be transmitted. Instead of taking a single actuator and not encoding the information temporally, [VTBC20] used multiple actuators to create a spatial display. Hands would be a good target of their spatial display, however, they are often occupied by other tasks, therefore Vyas et al. decided to concentrate on the toes. But the differentiation of stimuli on the fingers is not comparable to that on the toes. Tactile stimuli can be distinguished better on the fingers than on the toes. Using a special encoding that incorporates neighbouring toes in a temporal sequence, they showed in a study that encoded tactile rendering can significantly increase performance [VAB+20]. The presented toe interface has shown how sensitive and what abilities the feet, especially the toes, have to perceive information.

Closely related to foot stimulation is gait. Therefore the following paragraphs briefly discusses the gait. The physical aspect of locomotion can be defined by gait, the bipedal (forward) propulsion caused by the human limbs, which is affected by, for example, velocity and ground surface [FN10].

Gait is comprised of the different stride phases, in which the legs are moved, and the feet hit the ground. Stride phases differ in both frequency and length, depending on how fast the person moves. They include the stance phase (where a foot touches the ground) and swing phase (where the leg is moved and the foot is airborne); combined, they form

one gait cycle. Thereby, ground contact of the foot is defined by a roll-off process of the human foot, affected by different force (pressure) phases underneath the foot sole. Furthermore, the amount of ground contact per roll-off (step) differs with velocity. Foot strike can differ between different people, as for example runners commonly have either heel or mid-foot strike [SF11].

Feedback for both physically moving and non-moving people has been approached from various directions. Not only the feet themselves but also the legs have been stimulated, for example through two motion platforms [IYN01]. The GaitMaster is specially designed for uneven terrain; two platforms on which the user stands can simulate the ascent and descent of stairs. In a more advanced version, the GaitMaster also allows omnidirectional movements. This was achieved by using a turntable as base.

In summary, it was shown that foot haptics were mainly explored to simulate ground textures [NNTS12] and walking-like cues [TMM⁺12]. In most cases, these cues were provided by vibrations attached to a shoe [TBS13] or surface [FDL16]. Specifically for the sole of the foot, it was shown that it is sometimes better to give less stimuli than too much [TMM⁺12]. Foot sensations have great potential because, on the one hand, the distance between two stimuli to be distinguished is small [Wei68] and thus more sensitive than many other parts of the body, and on the other hand, despite these capabilities, they are simply not included in interaction in most settings. In addition to tactile interfaces, interfaces related to gait were also presented [IYN01]. Gait is controlled by our kinaesthesia and proprioception, which are also part of our sense of touch. The following section synthesises necessary research on remote environments and how to br presence and interact in a remote location with the help of a robot. Finally, the context to haptics is established.

2.2.7 Remote Environments

Overall, robotic telepresence helps people to communicate over distances, it allows users to control and navigate a system to interact in a remote environment, and it usually contains a video conferencing system. As this thesis explores a combination of telepresence robots and haptic feedback to the feet, the state of the art of in this area will be discussed in the following. In Section 5.2, a haptic user interface is introduced, this interface supports users steering a remote robot by providing proximity and collision feedback to the feet.

Telepresence robots have been studied in a variety of settings, including academic conferences [NVPH16], [RN17], offices [RTM12], schools [NO17], and health care settings [KKL14]. Across these settings, researchers have found strong benefits, mostly stemming



Figure 2.29: A conference scenario showing the remote robot beam (left) attending a talk [NVPH16].



Figure 2.30: The difference between control your telerobot, and being telepresent through a tablet controlled by a third person [YJNS18].

from the ability to be mobile in the remote environment with a form of "physical body" [NVPH16] (see Figure 2.29).

Digital mobility has supported people attending meetings and maintaining informal awareness in the workplace [RTM12], attending social gatherings and mingling with others [NVPH16], [RN17], engaging in small group activities such as workshops [RN17], supporting long-distance relationships [YJNS18] (see Figure 2.30), establishing friendships and participating in learning activities [NO17].

In all described situations, the user has a view into the remote space and can easily change that view by moving the telepresence robot. Control over what is seen is in the hands of the remote user. This contrasts with other video conferencing setups (e.g. Skype) where the view is typically in the hands of the local user [JWB⁺15].

There are still some challenges. For example, it can be hard to understand the body language of the operator of the telepresence robot [NVPH16]. Many telepresence robots do not have hands or arms so they cannot easily touch objects or interact in the physical world [NO17]. Privacy challenges stem from being in two environments at the same time [NVPH16], [NO17]. Telepresence robots also create challenges in understanding how one is presented in the remote environment, including how one looks and sounds [PST⁺11], [TH13], as well as where one is spatially located [NVPH16]. This makes it challenging to manoeuvre telepresence robots while performing other tasks like talking [NVPH16], [RMT⁺14], [TH13].

It can also be hard to know where one is in a building, especially if it is a new space for the user [NVPH16]. In crowded spaces, it is difficult to avoid obstacles or people with a telepresence robot [RN17]. Researchers have investigated a variety of means to increase the amount of feedback that users have when interacting through a telepresence robot. This includes wider fields of view [JRMT15] [JP02] and audio feedback to know how one sounds in the remote space [JP02]. Yet there remains a design gap in providing telepresence robot users with means to help them receive feedback and improve their spatial awareness of obstacles while moving through a space.

The combination of haptic feedback with a robot is quite advanced in the fields of humanoids [AFHW⁺18] and also in human-robot interaction [Oka18]. However, when looking in particular at haptic feedback to support the control of a robot in a remote environment, it becomes apparent that it is mainly in the area of assembly tasks and less often in the area of awareness. Hannaford et al., for example, showed in a very comprehensive study that in different manipulation tasks the performance could be improved with additional feedback [HWMZ91]. The tasks included peg-in-hole insertion, electrical connectors, Velcro attach-detach, and a twist-lock multipin connector, all involving high-precision control of the robot.

Haptic feedback to enhance navigation is much less widespread. Hacinecipoglu et al. investigated how haptic feedback affects performance in obstacle avoidance scenarios [HKK13]. In a study in which an unmanned vehicle had to be steered through a course with a set of obstacles, they were able to show that the participants completed the course faster and with fewer collisions with haptic feedback. Nevertheless, it should be mentioned that the course was simple and a conventional gaming steering wheel was used for the haptic feedback. Using the NASA TLX, Hacinecipoglu et al. also found that the mental demand is lower with haptic feedback. Level of frustration and effort also perform better with haptic feedback. A similar result was shown by Ju and Son. [JS19] also looked at the impact of haptic feedback on obstacle avoidance. Instead of a steering wheel, a commercially available haptic device was used. However, even the somewhat unusual or unknown navigation technique does not seem to have a negative impact on performance.

Another approach using a commercial interface to enhance navigation performance was presented by Lee et al. [LKSP04]. Although the authors' placed their research in the telepresence domain, it consisted of an unmanned vehicle and no social or other interaction other than navigation was described. However, an interesting aspect compared to the predecessors was implemented and explored. Whereas the others focused mainly

49

on collisions, Le et al. have also included the environment in the feedback spectrum. However, an interesting aspect compared to the predecessors was implemented and explored. Whereas the others focused mainly on collisions, Le et al. have also included the environment in the feedback spectrum. The results reflect the trend described above and Le et al. showed that the number of collisions can be reduced by haptic feedback.

2.2.8 Navigation Cues

Navigation is one of the key tasks performed in real, virtual and remote environments, it encompasses both physical and psychological aspects. It is necessary to interact in both: virtual realities and telepresence scenarios. Since navigation is often enriched with haptic feedback in order to optimise it, different approaches are presented below. These range from light vibrations [VBV⁺12] to interfaces based on kinesthesia, for example walking in place [BKLP05] or leaning [GPI⁺15]. Navigation means how to move from one place to another in an environment. In principle, there are three different navigation tasks: exploration, search and manoeuvring. Each of the tasks pursues a different goal, while exploration focuses on exploring the environment to gain knowledge. In the search task, a specific location is targeted, and the shorter the path and the time, the better. Manoeuvring also involves heading for a specific location, but the focus here is on avoiding possible obstacles. Short, precise movements are important to complete the task well.

Physical navigation interfaces have been studied widely and can increase the overall usability and user experience of the system [BKLP05], [RBM⁺10], enhance spatial perception and orientation important for a wide range of tasks [BKLP05], and reduce motion sickness [BBG08]. Most of the factors mentioned above also affect virtual environments as well as controlling a remote robot. In both cases, orientation in space and spatial perception are essential. In the following, we look more closely at related studies on leaning-based navigation interfaces, as well as feedback to support navigation.

Leaning-based interfaces are often used for travelling and navigating in virtual environments like games or virtual environments. Some prototypes utilize a Wii balance board, and its pressure sensors, to perform navigation tasks in virtual environments [VSBH10]. Through the shifting of weight users are able to navigate in a 3D world. Other types of leaning interfaces use the body position determined by an optical sensor [GPI⁺15] or a spring mechanism to navigate through a environment [MPL11]. Even unusual locomotion metaphors, such as surf-like metaphors, were examined using a Wii Balance Board [JL12].

Leaning interfaces to some degree resemble other interfaces that keep the user physically at one location while walking, these include metaphors such as walking in place (WIP) interfaces [TDS99], natural motion interfaces such as those supported by treadmills [DCC97], or navigation systems for seated users. An overview of many techniques can be found in [BKLP05], while a focused overview of how, in particular feet can be used for interaction purposes is described in [VSA⁺15].

Another aspect that is important for navigation is the so-called self-motion perception. Embodied self-motion illusions (e.g. vection) have been studied and can be induced in stationary observers by moving visual flow fields, moving spatialized sounds, and biomechanical cues from walking on circular (but not linear) treadmills [RFR09]. Visuallyinduced vection can be enhanced by adding simulated camera motions that mimic jitter [BKAP11] or head bobbing, the vertical and horizontal oscillatory motion of the head during natural walking [BB10b], which can be communicated as a purely visual cue [TMM⁺13], as well as through physical movement of the user [ISK⁺14].

Some studies showed that minimal provision of vestibular cues can enhance self-motion [SF11], as do footstep sounds [TMM⁺12], wind [FDL16], and tactile patterns associated with walking [RSP13] or leaning in sideways directions [KMT⁺15], all of which showed positive effects. Likewise, body pitch could also affect self-motion [BVB10]. Studies on actively tilting the body have shown that while horizontal (sideways left-right) vection was not affected by body tilt, vertical (a.k.a. elevator) vection was reduced for upright posture and increased to the level of horizontal vection as body tilt increased [NS98]. In contrast, static leaning has also been shown to affect self-motion for seated users positively [KRTK15].

The usage of vibration to provide cues related to navigation other than self-motion has also been studied. Directional information through vibrations has been explored through foot-based [VBV⁺12] and body-worn devices. For the latter, it has been shown that directional cues can be successfully provided by stimulating a specific side of the body [LSMM⁺05]. It has also been used to convey collisions [BB10a], or even to avoid collisions during navigation [AB11]. Collision avoidance also relates to the breed of interfaces that have been developed to indicate proximity, which has been studied for general 3D selection tasks to indicate how near the user is to a target [ABKS18], [MKT⁺18], but also has found specific application in navigation systems. For example, SpiderSense [MHL⁺13] uses vibration motors distributed over the body to support navigation for the visually impaired. This kind of feedback is similar to a distance-to-obstacle feedback approach to communicate distances to surrounding objects [HOAH15] and wheelchair operation using a glove-based interface [UCP08].

Many different navigation methods have been presented in this section. They range from slight navigation cues based on vibrations $[MHL^+13]$ to leaning-based interfaces

[GPI⁺15]. Some navigation techniques were presented to increase self-motion perception [KRTK15], while others focused on more efficient navigation [MKT⁺18]. In the next section, a summary of related work is presented that leads to the two research questions of this thesis.



Figure 2.31: A summary schematic representation of the dependencies and intersections between the two research areas. The diagram also shows the dependencies of related work and subject areas on the research questions. The references represent a representative of each subject area or refer to a specific section or chapter.

2.3 Summary and Research Questions

In the previous sections, related work have been introduced, discussed, organized in sections in order to present corresponding research. An overview of haptic interfaces from both focus areas is given at the beginning of this chapter (see Table 2.1). The table shows related work and categorises them into five groups:

- 1. Actuator (the key technology responsible for providing the haptic feedback)
- 2. Feedback type (sensation triggered by the feedback)
- 3. The **focus** of each research
- 4. The part of the **body** that is stimulated
- 5. Classification into application areas

These five groups appear directly or indirectly in many of the related work presented so far. For example, looking at the sectioning into body regions, it can be seen that the first sections of this chapter are more focused on haptic feedback for the hands and fingers, while the last three sections address the feet and present haptic interfaces from this area. When looking at the application areas, it becomes apparent that the hand or finger's stimulation focuses on touchscreen interaction, while haptic foot interfaces mainly refer to virtual reality and remote environments.

A schematic overview of the main dependencies and intersections is given in Figure 2.31. It highlights three main aspects. Firstly, the graph shows the connections, intersections and dependencies of the related work presented. Secondly, it shows which related work or areas have an influence on finding the research questions and thus a need for further research. Finally, the link from the research questions to the corresponding chapters is provided. Looking at the intersection (see Table 2.1, highlighted in blue), it becomes clear that the humans, their perception, various aspects of performance and tactile and force feedback are are part of this thesis, which can also be summarised under the overarching concept of haptic interfaces and interaction [EOEC11].

Both summaries, Table 2.1 and Figure 2.31, are a starting point and the base for the following two subsections. These will lead to the research questions and highlight the need for further research that has been dealt in this thesis.

2.3.1 Haptic Feedback for Touchscreens

The first focus area addresses the challenge of pairing haptic feedback with touchscreens. Three open challenges have been identified and highlighted, which are then brought together and linked in a single research question. The three challenges all explore ways in which haptic feedback can enrich touchscreens. More specifically: **RQ1.1:** What are the limitations of force feedback provided through a rigid screen?

RQ1.2: How can elastic materials be combined with touchscreen interaction?

RQ1.3: How could a small mobile touchscreen be equipped with haptic feedback?

All three sub-questions of the described challenges are taken up and dealt with again in Chapter 4.

In Section 2.2.1, it was shown that two modalities of haptic feedback can be found in combination with touchscreens, namely tactile [FS01] and force feedback [HTE⁺17]. Furthermore, it was described that both types of feedback can be generated using different technologies (see Section 2.1.2), which in turn has an influence on the type and manner of feedback. Finally, we have shown that tactile cues, combined with mobile touchscreens, are well researched [GSK⁺20]. Various systems and prototypes have been presented that can provide tactile feedback, ranging from simple vibrations [BCB07] to friction screens [KIP13]. Tactile feedback was used, for example, to improve the perception of GUI elements [KBL14] or to improve performance [BCB07], e.g. to minimise the error rate when writing on a virtual keyboard. Force feedback, in contrast, combined with mobile devices, such as smartphones, has been rather rare researched and still has further potential.

Actuated display systems such as the vertically mounted display prototype of [SPB13] and the horizontally mounted tablet proposed by [KHSK14] have shown that their systems can make contours and shapes tangible. However, not all limitations have been explored, in particular, Sinclair et al. used a stereoscopic system and their screen was mounted vertically. In addition, TouchMover only supported 1DoF. In the study conducted, the perception of different geometric shapes was investigated, but the users were able to see the screen, which provides additional visual cues and can potentially support haptic perception. Furthermore, the users were directly informed of their result after an exploration, which in turn can lead to a faster learning effect [GS05]. In contrast, Kim et al. focused strongly on haptic rendering techniques, such as the control of the screen, without considering perceptual limitations [KHSK14]. Although Kim et al's system is similar to the one in presented Section 4.2, they differed in the evaluation and implementation of the system. Any complex shape can be represented using such a system, apart from a few physical limitations. The question that should be in focus, however, is what information can be processed with an actuated screen through our haptic perception. To some extent, Hausberger et al. have addressed this question and found that the size of a geometric shape can be better detected when tactile feedback is added to the screen actuation [HTE+17]. Also the work of Nagamatsu et al. which deals with multi-touch in the context of actuated

touchscreens could benefit from more detailed information [NNTA14]. In general, the existing literature does not explore the resolution which can be provided by an actuated screen. We want to investigate this in Section 4.2 by showing the implementation of our interface, determining the horizontal and vertical resolution, and also looking at the limitations of such a systems.

We also discussed flexible and mid-air interaction (see Section 2.2.5), and it turned out that there is still much potential for further research in this area [HBH⁺16]. We consider the combination of pen-based interaction [WRSS16] with elastic and mid-air interaction [PGG⁺12] to be promising, we will introduce at multi-layer interaction with a touchscreen in Section 4.3. Therefore, we propose a flexible surface that combines tension feedback with vibration as well as rigid feedback to improve pen-based interaction with touchscreens. In summary, the interaction with pens on touchscreens is manifold [CBMB16], [LDL+04], [WRSS16]. Both active and passive pens were presented, and different technologies were used for actuation. The field of applications ranges from the simulation of realistic pens [WRSS16], e.g. friction, to other interaction metaphors [BPIH10], e.g. supporting the selection of objects. It is undisputed that interaction with elastic surfaces opens up new design spaces [PGG⁺12]. Similar to pen interaction there are passive [MKG⁺14] and active systems [SHS16]. Neither the interaction with a pen combined with an elastic material nor the combination between elastic materials and a touchscreen has been sufficiently researched, as was shown in Section 2.2.5. Our approach introduced in Section 4.3 opens a new kind of multilayer interaction space, extending previous above-the-table multilayer approaches [SAL06] by adding the exploration of haptic properties in the spatial flexible haptic layer. In contrast to both non-contact-based and mechanically linked systems, our system extends hover space approaches through contact-based variable compliant haptic feedback. We principally divide above-the-display space into virtual (without contact) and physical (with contact) space. Hover space generally explores virtual objects in a virtual space [ZDE16], whereas our work looks at virtual objects in a physical space.

So far, the focus was on touchscreen interaction, without considering the portability of the screen. Since the components needed for tactile feedback are very small and require low power, it is not challenging to provide mobile or handheld touchscreens with tactile feedback [PH20]. As already mentioned, there are some force feedback and pin-based approaches [NFM⁺19], [SPB13], but all of them are not portable. The literature review has shown that haptic feedback is usually used in prototypes to represent the real world's physical properties [RVGG17], like buttons [HKLB08], or make these properties tangible to the user. Less publicised, however, there are approaches that use haptic feedback to improve performance [BCB07], e.g. to enable blind people access to digital content [PBKW18].

The design problem that it is difficult to place larger components, like actuators, on the front of the screen, as they would obscure the view and interfere with the interaction with a touchscreen. In this thesis, the back of a smartphone became a feasible solution for placing force feedback. There is another advantage when the back is included: interaction on the back of a device can improve visibility on a touchscreen [CDVB17]. The fingers that are usually used to hold a smartphone can now be used for interaction [BC09]. BoD interfaces can create entirely new or transfer already familiar metaphors for interaction with a smartphone $[SMSJ^{+}15]$. At the moment nearly all prototypes are based on dedicated hardware and highly specialized on particular applications [JKT⁺16]. A fingerprint scanner on the back for authentication is a technology that has prevailed with several manufacturers and accepted by the users. BoD interaction is mostly used as an input method, i.e. to control the behaviour of an application [HMT03]. Feedback is primarily visual through an integrated LCD or a visual projection. Rarely do researchers deal with haptic feedback on the back-of-the-device. However, Yannier et al. have shown that the reading experience can be increased with haptic feedback, which suggests that there is still potential [YILK15]. Another issue that has had a considerable impact on the acceptance of BoD interaction are ergonomic aspects, for example, the accessibility of interaction elements without taking an uncomfortable posture by maintaining a stable grip can considerably affect the usability [CDVB17].

A number of TUIs and SSIs have been introduced (see Section 2.2.4). While traditional TUIs use the natural physical properties of objects to manipulate data, newer interfaces usually include additional technologies. Although the interfaces presented are not directly related to touchscreens, they fit very well into the context of this work. For example, pin-based interfaces [NFM⁺19] were presented, which can also be found in a modified form in Section 4.4. In addition, deformable interfaces [FLO⁺12] and interfaces that change their state [SYT⁺17] were dealt with, which partly served as sources of ideas for the research in this thesis.

In general, haptic feedback for mobile devices has not been researched in depth in the existing literature. Summarising the previous sections, it is clear that the combination of haptic feedback and touchscreen interfaces has the potential for further research, which led to the first research question addressed in Chapter 4:

RQ1: How can touchscreens be augmented to provide haptic feedback that goes beyond tactile sensations?

2.3.2 Haptic Feedback for Virtual and Remote Environments

While the first section explored the combination of touchscreen interaction with haptic feedback and thus concentrated on the hands and fingers, the second section will switch focus on the feet. In detail, haptic feedback to the feet in virtual and remote environments is explored. Figure 2.31 gives an overview and puts this area in the context of the thesis. Furthermore, it shows the dependencies to the first area as well as direct aspects and domains that are important. Haptic interface design and the haptic feedback principle are further aspects that connects both areas.

Similar to the previous summary, open challenges will first be identified and highlighted, and then brought together and linked into a single research question. The main challenges explore ways in which haptic feedback can enrich virtual and remote environments. In particular:

- **RQ2.1:** How can foot haptics support users awareness when interacting in remote environments?
- **RQ2.2:** What influence do foot haptic cues combinations have on self-motion perception in VE?

In Section 2.2.6 it was shown, that haptic foot stimulation pursues different goals. On the one hand, foot-based feedback is used in virtual environments to increase immersion [TBS13] - to make virtual environments more realistic [BB10a] [WTC⁺20]. On the other hand, tactile stimuli on the sole of the foot are also used for navigational cues [BB10a], shape recognition [VBV⁺12], and rendering surface texture and firmness [NNTS12] [SGB⁺18]. Interfaces based on weight shifting or leaning can also improve self-motion perception [KRTK15]. Different prototypes were also presented that simulated the feet with haptic stimuli [TBS13], [VTBC20]. In general, however, the literature did not show how the combination of leaning-based interfaces and foot haptics behaves. In addition, it is therefore interesting to study, how the combination of different feedback types perform, as they have achieved positive results individually. In Section 5.3, we introduce a new foot-stimulating approach that increases self-motion perception in virtual environments.

However, we do not only want to deal with foot-based feedback/stimulation in the context of VE, we further want to explore foot haptics in combination with telepresence robots [NVPH16]. With regard to the implementation of haptic feedback, both areas (VR and RE) are very similar. In both cases, the user is immersed in a remote world. With the only difference that once it is synthetically generated and the other is real. The telepresence robots can be seen as virtual avatars and are used today in many different ways, as discussed in Section 2.2.7. Nonetheless, there are still open challenges, such as how to control the remote robot in dense environments and how to improve the spatial perception of the remote environment. These challenges are discussed in Section 5.2.

The usage of haptic feedback to enhance robotic telepresence systems has mostly been to support manipulation tasks in teleoperation to improve accuracy and awareness [HWMZ91]. Using haptic feedback to support navigation is far less common. Some examples provide haptic cues for collision avoidance to improve situational awareness [HKK13], [JS19]. However, all studies used hand-operated devices instead of providing feedback to the feet, which can provide an additional sensory channel that is mainly unused in current telepresence robots despite being the body part that might naturally bump into things. Proxemics, the notion of proximity, has also found some interest in robotics, albeit foremost in relation to social aspects [VJ17]. Only a few studies have focused on the effects of haptics on navigation performance, including [LKSP04] that explored driving a mobile robot (non-telepresence). Results showed improved performance and presence. However, systems were studied in environments that were neither complex nor dynamic. Using haptic feedback to improving self-motion has been probed before, especially in Virtual Reality systems. Embodied self-motion illusions (e.g., vection) have been studied for quite some time [RSP13] and can be realized with a variety of methods, including visual cues [BB10b], sound cues like moving sound sources [RVSP09].

Nonetheless, all these systems do not have the granularity of feedback - that is, providing directional proximity and collision feedback - afforded through the system introduced later (see Chapter 5). Overall, we do not see explorations in the related work of how haptic feedback systems can be designed to aid telepresence robot driving, especially when systems are designed for the feet. Furthermore, we do not see related work on how people would experience such systems, which is the focus of our work. The above sections have shown that there are still open questions, in this thesis the following research question will be addressed:

RQ2: How can foot haptics for navigation tasks be designed to improve selfmotion perception and spatial awareness in virtual and remote environments?
Chapter 3

Research Methodology

3.1 Introduction

There are several ways in which humans can interact with computers, similar to the interaction between humans, whose interaction involves both our sensory perception and the motor control of our effectors. Humans have five senses: sight, hearing, touch, taste and smell, and according to Dix et al., sight, hearing, and touch are currently crucial for human-computer interaction [DFAB04]. Taste and smell are not so widespread and mostly used in research at the moment. Just like humans, computers also need an input and an output channel in order to communicate.

Sensors and actuators are essential in this context, as they can, to a certain extent, substitute our senses and effectors. Later, a mapping between human (functionality) and machine (sensor or actuator) is presented (see Table 3.2). Both are necessary to provide haptic feedback: a sensor is responsible for recording the haptic stimuli that the user applies to an object and sends these force readings to a computer (input). It converts this information into a form that can be perceived by the user through an actuator (output) [EOEC11].



Figure 3.1: An overarching diagram of all research studies conducted. It highlights the common research methods and analyses as well as the outcomes.

In line with previous work in this area, the methodology is adapted to do multimodal and multisensory research, focusing on haptic interaction. Consequently, both the human being and the interface are at the centre of this research. An overview of the methods used and the interplay of the studies conducted can be found in Figure 3.1. The diagram includes the motivation and the identified research questions (see Section 2.3 and Figure 2.31). It depicts the two main research areas, and their associated sub-areas are highlighted. Furthermore, the methods used for prototyping, evaluation and analysis are presented. Finally, a brief outline of the outcomes is given. The diagram shows the connections in this chapter but also beyond this chapter and links the different parts of this thesis. For a more detailed view of the methods used, a procedure and design diagram is presented below before each study.

The following provides a summary of the literature review, a definition of HCI and two philosophies on what research is and how to perform research, especially in our field. Subsequently, research methods are discussed to illustrate the diversity of research in the area of haptic interaction. The chapter deals with creating haptic interfaces and methods for performing user studies, including collecting data, sampling, evaluation, and ethical aspects.

The Literature Review and the resulting research questions (see Section 2.3) have led to a more precise classification of the subject area. Thus, the focus is on the design, the implementation and the evaluation of hardware interfaces to provide haptic feedback. The thesis looks into both human perception issues and interface design. That implies that following models, techniques and procedures will be applied to find answers to the research questions.

To improve the understanding of the methodology, first of all, we would like to refer to a definition of HCI and show that HCI research draws on many different disciplines, including computer science, sociology, communication, psychology, human factors, industrial engineering and design [LFH17]. This makes research in this field undoubtedly exciting, but also more complicated since in most cases no "standard protocol" can be applied. When considering the number of disciplines, it is also common that research methods are mixed and modified for the use in HCI [LFH17].

One of the first definitions of human-computer interaction highlights this diversity. Accordingly, Hewett et al. define human-computer interaction, in the ACM SIGCHI curricula as follows [HBC⁺92]: Human-computer interaction is a discipline concerned with the design, evaluation and implementation of interactive computing systems for human use and with the study of major phenomena surrounding them.

On the one hand, the definition lists different technical disciplines, such as implementation (e.g. Computer Science/Engineering) and evaluation (e.g. Statistical Data Analysis), and on the other hand it includes the human being (e.g. Physiology) and further subject areas. In considering haptic perception, for example, many disciplines such as neurophysiology, psychology and psychophysics come together [EOEC11].

In the following, two fundamental research approaches are briefly introduced to get an understanding of the selected methods. Then the several research methods that used in the thesis are presented and linked to the respective sections.

3.2 Quantitative vs Qualitative Research

The quantitative approach is inspired by the philosophy of positivism, which is based on the assumption that everything around us is real and that this reality can be described logically. Within this approach, an order exists, which means that a specific description can define everything. It is a rational and objective approach compared to a qualitative approach which is instead a more subjective approach. New knowledge develops from existing knowledge [LFH17], meaning that the research conducted in a quantitative approach is based on a deductive principle, where one concludes from the general to the specific.

Quantitative study designs are specific, well structured, have been tested for validity and reliability [Kum10]. They are explicitly defined and, therefore, systematically designed. This, in turn, means that gathered data typically also have a specific structure. Results are obtained by evaluating the data. In a quantitative approach, questions such as how something is or how it relates to each other are mainly asked.

According to Rogers et al., a quantitative approach can be used to answer specific research questions [RSP11]. For this purposes, mainly questionnaires are used, but other methods can, of course, be used for gathering data, e.g. log files recorded during an experiment, or user activity can be recorded. A quantitative approach uses statistical methods for validation. Validation is often done employing a significance test, in which collected data are examined for individual properties and/or correlations between properties, such as error rate, speed and distance. Findings can be presented with charts or figures. In contrast to a quantitative approach, which uses objective methods deductively to answer specific questions, qualitative methods are based on subjective data and conclusions are drawn inductively. Similar to the quantitative approach, a philosophy can also be assigned to the qualitative approach, namely interpretivism. Similar to positivism, there is a real world, but it cannot be explained in the same way for everyone. Everyone experiences the world individually through his/her perceptions which are influenced by their preconceptions, beliefs and values [Wal11]. In this philosophical approach, humans are part of the world, so they are not neutral.

A qualitative study design usually is not well structured or follows a logical process such as quantitative study design do. It is, therefore, difficult to plan an experiment. A qualitative study design is less specific and precise, and does not has the same structural depth as quantitative study design [Kum10]. Since research results originate from the subjects' own opinions, questionnaires may be a wrong way to conduct research within a qualitative approach. Since there is no special protocol and subjects participate interactively in the study, a pre-designed questionnaire cannot dynamically deal with potential new aspects. Therefore, this approach mainly works with interviews, which, of course, must also be well prepared. This allows the subjects to be addressed individually in order to find out their opinions and preferences on a specific or broad topic.

Unlike quantitative approaches that use statistical tests for validation, a qualitative approach does not have a clear, straightforward validation, and the process of analysis is often more complicated. The focus here is not on numerical measurements, but on the evaluation of observations, videos, interviews or studies texts, and therefore the understanding of complex situations [LFH17]. What makes it even more complicated is that the statements of for example two test persons contradict each other, and since the truth is not known either, the comments have to be interpreted and discussed to find a feasible solution to the problem.

In summary, it was shown that doing human-computer interaction research is very diverse since many different factors influence this domain. Therefore, it is challenging to decide using either a quantitative or qualitative approach. Consequently, it makes no sense to strictly separate the two methods but rather connect them through a continuum. Partly mixing these two philosophies becomes visible in this work (see Figure 3.1). However, our research philosophy is based more on a quantitative approach. Nonetheless, especially at the start of creating new prototypes, qualitative methods, like discussions and interviews, are often crucial as pointers. An overview of the main characteristics of quantitative and qualitative approaches is illustrated in Table 3.1. In the following, we will further discuss on the methods used in this thesis.

	Quantitative	Qualitative
Philosophy	positivism	interpretivism
Approach	objective	subjective
Reasoning	deductive	inductive
Sampling	random	purposive
Data	structured	individual
Records	numerical	verbal
Evaluation	statistic	interpretive

Table 3.1: The table shows the main characteristics and differences between a quantitative and a qualitative approach.

3.3 Haptics Exploration

There are many methods and procedures for researching haptic user interfaces. To find the right research direction, it is essential first to understand what contributions can be expected and then work towards those contributions. Wobbrock and Kientz present seven potential contributions, ranging from empirical, artefact, methodological, theoretical, dataset, survey to opinion contributions [WI16]. When considering the number of contributions published in HCI, it can be seen that empirical contributions dominate, followed by artefact contributions. Since the contributions that have been made belong to these two categories, both will be briefly explained. An empirical contribution is the analysis of qualitative and quantitative data collected, for example, through an experiment or a survey. An artefact contribution is a contribution that introduces, for example, a new interface or a new toolkit. The combination of an artefact followed by an empirical contribution reflects the contributions achieved within this thesis. According to Lazar et al. this combination is also known as HCI system research or HCI interaction techniques [LFH17].

However, how can one achieve these contributions? Shneiderman et al. describe ways detailing how to gain findings or contributions $[SPC^+16]$. The most frequently used methods to collect HCI data are observations, field studies, surveys, usability studies, interviews, focus groups, and controlled experiments $[SPC^+16]$. Each of these options has its strengths and weaknesses. For instance, observations can be extremely time-consuming and complex to analyse [LFH17] (for more detail see Section 3.7). A general scientific research approach (see Figure 3.2) can also assist in gaining contributions. This cycle reflects standard scientific methods, but they can, of course, be transferred to researching

haptic interfaces. Observation and the resulting idea and design of a new hardware prototype is the starting point of each cycle in this thesis. An evaluation follows them: an empirical study, to collect and analyse data.



Figure 3.2: A standard iterative cycle of scientific methods, inspired by [Nor13], [Wal11]

As introduced in the fundamentals, the sense of touch is different from the other senses. When, for example, researching visual feedback, the focus is usually on algorithms for generating image content rather than on prototypes [Gol10]. However, it is currently common in haptics research to develop prototypes [EOEC11], due to the versatility of the sense of touch. An essential method for developing haptic interfaces is **pilot testing**. This method supports more efficient planning and helps determine the parameters for the experiment to be conducted [Gru08]. In many cases, it is sufficient to conduct the pilot study with a very small sample in order to collect initial results and feedback [LFH17]. This will help to iteratively improve the feedback and its design. The method of conducting pilot studies can often be found in related work [WRSS16], [HNW+14], [MKT+18]. In this thesis, pilot studies were performed on all prototypes in order to make haptic feedback as appropriate as possible. A good example of pilot studies can be seen in the design of the haptic back-of-device feedback for smartphones. The Figure 4.21 from Section 4.4 depicts early prototypes optimized through pilot test.

Adopting a quantitative approach and aiming to quantify haptic perception, there are methods that can measure haptic perception [JT13]. Methods that are often used to quantify the properties of haptic feedback come from psychophysiology. Psychophysiology is a field of research that deals with measuring physical changes in response to mental states [CW11]. A question that often arises with haptic interfaces is what can be represented with a particular interface or how accurately changes can be perceived through our sense of touch. The main method to determine the perception threshold is the **just-noticeable difference** (JND). It measures how much we need to change the stimulus along a certain dimension before people even notice the change. A table of sensory resolutions of tactile and haptic stimuli can be found in [JT13]. There are classical methods, such as the method of the constant stimuli [Ges97], and adaptive methods, such as the staircase method [JT13]. A crucial difference between the two methods is that when using the method of constant stimuli, all values have to be set in advance. In contrast, using an adaptive method, the stimuli presented depend on the user's responses. Both methods were used in our experiments to quantify haptic perception (see Section 4.3.2 and Section 4.4.2).

In this context, two concepts will be discussed more closely. Firstly, there is the Weber's Law. It states that the ratio of the perception threshold to the stimulus intensity is constant. The constant of $\Delta I/I$ is called the Weber fraction. Secondly, there is the psychometric curve. It describes the relationship between a person's responses or perception and the physical stimulus. In other words, it represents the probability of detecting a stimulus correctly along the vertical axis and the intensity of the stimuli on the horizontal axis. It is the basis to determine the JND [JT13]. As an example, the psychometric curves obtained from the experiment in Section 4.4 can be shown (see Figure 3.3). Obtained data of all individual participants were fitted using Weibull function. Each plot shows the stimulus level on the x-axis and the percentage of correct stimuli detected on the y-axis. The blue dots show the probability that a particular stimulus was detected.



Figure 3.3: Psychometric curves obtained from the experiment in Section 4.4

Other methods that can be used to quantify haptic feedback are the active palpation and **detection of shapes** or geometries [SPB13] and surface textures [SGB⁺18]. When recognising shapes, different geometric shapes are rendered haptically and the users have to identify the shapes, usually without visual feedback. The detection likelihood can be represented by a matrix [SPB13], [HTE⁺17], as can also be seen in Section 4.2. Explicitly controlling variables to limit the parameter space of a experimental design is a common used method. A example of limiting the parameter space is to follow a path or marker [GPI⁺15], [RF12]. With this method, the user is asked to follow a certain object or marker. It can be used, for example, to control the velocity. This method, for instance, was used in Section 4.2 and Section 5.3, among others. In both cases, this was done to better compare the haptic sensation.

In the following, further research methods are discussed, including developing the prototypes that will be introduced in the main chapters, and methods for performing user studies/experiments are discussed, including designing, performing, and evaluating those.

3.4 Prototyping Haptic Interfaces

It can be asked whether a prototype is a right approach to explore haptic interaction. Would it not be enough to do a questionnaire or a survey? The answer is yes and no - to get specific trends or opinions (qualitative) - a survey is sufficient. However, if someone wants to explore an aspect more closely, it is essential to develop a prototype. Especially for haptic or spatial interfaces, as introduced in the main chapters, it is essential to try things out [OSC⁺17]. Practical experiences are often necessary to draw conclusions, especially when it comes to the sense of touch. As five prototypes were created, prototyping methods as used in this work are presented below.

Initially, it must be clarified how the haptic interface should look like and how it should behave. Therefore a physical and conceptual model must be created, while the conceptual model includes functionality and behaviour [RSP11], the physical model, the prototype, looks into the appearance, the materials and the physical constraints. In this thesis, methods ranging from virtual and real prototyping, rapid prototyping, vertical and horizontal prototyping, low fidelity prototyping and high fidelity prototyping were used to create haptic user interfaces. Usually, however, it starts with a pen and a sheet of paper on which the first sketches are made. For a deeper understanding of these techniques, they are discussed in the following subsections.

According to Buxton, a sketch can help: to think more creatively about ideas, to instantly generate a lot of ideas (quality does not matter), to record ideas, to share them with others and to have fun while designing and sketching [Bux07]. In addition, sketching does not require many tools; for instance, a paper and pen are portable, cheap and always on. It is difficult to draw a clear line between sketching and prototyping. In some cases, it merges into each other or in other cases sketches can be the first stage of a prototype [RSP11], [Nor13]. Of course, digital sketches can also be created, the principle remains the same, but with the digital counterpart templates can be created, components can be



Figure 3.4: A sample sketch using the developed template. The sketch was continued and contributed to this thesis.

reused and deleting a component is easier. A disadvantage of digital sketches is that users take longer to sketch an idea, as users try to create a high-fidelity screenshot instead of merely sketching an idea.

To further simplify sketching, a sketching template was developed as part of this thesis. This template contains five categories:

- a heading (the *idea*)
- a brief summary (*abstract*)
- a drawing area (*sketch*)
- the problem you're trying to fix (*target*)
- and *comments*.

The template aims to record ideas quickly and with the template it is easier to collect ideas. It is not necessary to fill in all points; it is just a framing. Figure 3.4 shows four of the many sketches that were made during the time of the thesis.

Similar to sketches, prototypes can be created either digitally or physically. Digital prototyping is known as virtual prototyping. An excellent example of virtual prototyping is design evaluation. Designers evaluate products that have not yet been built [Sut03]. Meanwhile, there is an increasing number of computer-aided design (CAD) tools that

allow a simple engineering and additionally support photorealistic rendering¹. The latter enables visual design evaluation based on a virtual prototype.

A further advantage of virtual prototypes is that a digital model is available. In combination with a 3D printer, a physical twin can be created quickly. 3D printing is increasingly being used in research to create new user interfaces [HZB⁺19], [OOH⁺19], [GZBV16]. Some user interfaces developed in this research are based on virtual prototypes and then being printed in 3D. Figure 3.5 shows, for example (from left to right) a virtual prototype of a haptic module for smartphones, the corresponding photorealistic representation and the physical 3D model. More details about this research prototype can be found in Section 4.4.



Figure 3.5: An example of a virtual prototyping process from left to right: a virtual prototype, a photorealistic rendering and a 3D printed physical representation.

Rapid prototyping is a further technology that allows to create haptic interfaces quickly and is often used in combination with manufacturing. It includes all technologies that can be associated with nowadays common 3D printing techniques. The main representatives are, among others: stereolithography, selective laser sintering, solid ground curing and, as mentioned, 3D printing [MRS15]. Similar to virtual prototyping, the geometry data must be available digitally. For this purpose, computer-aided X programs are used. Rapid prototyping is also used in the design and development of novel user interfaces; for example, the interface of [BHSO16] presented in Chapter 4. All prototypes were developed using rapid prototyping technologies. However, 3D printing can not only produce housings, it can also produce mechanisms up to a certain complexity, which will be seen later (see Figure 4.23).

Finally, there are low-fidelity and high-fidelity prototypes [LFH17] - both have their advantages. At the beginning of a development where the direction is not yet 100 per cent given, it is necessary to react quickly to new requirements. Therefore the prototypes must be built quickly and easily (low-fidelity). In contrast, in a final stage of the development,

 $^{^1 \}rm Autodesk$ Fusion 360 https://www.autodesk.com/products/fusion-360/ (last accessed 19th April 2021)

meaning before the prototype is complete and ready to be used for studies, it must have a higher quality and functionality (high-fidelity). This type of low-fidelity prototypes were often created to explore certain aspects, whether ergonomically or to answer the question: "how does it feel?"

It can be emphasized that prototyping is suitable for exploratory and agile development of haptic user interfaces. Most of the mentioned methods can also be found in many publications referenced in Chapter 2, directly or indirectly. Of course, prototypes can also be created with other methods, but the ones presented here are probably the most prominent. Engineering and prototyping is a part of the outlined research process. Researchers have to try and iterate to create a physical user interface, especially when developing haptic interfaces $[OSC^+17]$, and this is exactly what prototyping is appropriate for. The prototypes developed can be found in the main chapters. The combination of haptic feedback with touchscreen interaction is presented in Section 4.2, Section 4.3 and Section 4.4. Foot haptics prototypes that could increase presence and spatial awareness can be found in Section 5.2 and Section 5.3.

The following will discuss tools, materials and equipment used to build and design haptic interfaces. Since most developed user interfaces consist of several components, an overview of individual components and how to prototype them is given. The five main components are sensors, actuators, housings, one or more mechanisms and a control unit.

Sensors and actuators are as important to a physical user interface as our senses and motor system are to humans. With these components, interfaces can interact with the environment and with humans, Table 3.2 gives an overview of sensors and actuators. The table shows a wide range of sensors and actuators, the parameters to be influenced and their focus. For example, a touch sensor (type) can track the position (parameter) a finger of a user (focus). Alternatively, a projector (type) can change the appearance (parameter) of the environment (focus). Especially for haptic interfaces, actuators are necessary [EOEC11].

A control unit is required to control a sensor or an actuator. Therefore, several microcontroller platforms are suitable for prototyping user interfaces, for instance Phidgets², Adafruit³ and Arduino⁴. According to $[OSC^+17]$ the open-source platform Arduino is the most impactful platform in the context of user interfaces. However, the main difference between them is that Adafruit and Phidgets are proprietary products and offer additional hardware such as sensors and actuators. All three prototyping environments mentioned

²https://www.phidgets.com/ (last accessed 19th April 2021)

³https://www.adafruit.com/ (last accessed 19th April 2021)

⁴https://www.arduino.cc/ (last accessed 19th April 2021)

Туре	Parameter	Focus
Sensors		
Distance	Distance, proximity	user, object
RFID	Existence, proximity	user, object
Touch / force	pressure, force, position	user
Vibration	Surface vibration, displacement	user, object, environment
IR reflection	Light, interruption	objects
Spatial (IMU)	Movement, orientation	user, object
Bend	Flex angle / flex radius	user, object
Magnetic	Magnetic strength	object
Load	Weight	object
Loudness	Decibel	user, object, environment
Light	Lux	object, environment
Temperature	Celsius/Fahrenheit	user, object, environment
Humidity	Humidity	object
Soil moisture	Humidity, conductivity	object
Steam	Humidity, rain	object, environment
Alcohol	Alcohol (fumes)	object, environment
Gas	Gas, vacuum, pressure	object, environment
Camera image	pixels, objects, contours, light	
Actuators		
Servos	Displacement, rotation	user, object
Motors	Displacement, rotation	user, object
Linear servo	Displacement	user, object
Muscle wire / contraction	Contraction	user, object
Vibration	Vibration	user, object
Heat	Temperature	user, object, environment
Wind	Air flow	user, object, environment
Smoke, smell	Air particles	user, object, environment
Light	Illumination	user, object, environment
Sound	Sound	user, object, environment
Fluidity/taste	Taste	user
Visual/projection	Appearance	user, object, environment

Table 3.2: This table shows common sensors and actuators. In the first column the type of sensor/actuator is shown. The second column describes the parameters that can be mapped to the sensor/actuator. And the last column shows potential focus of the sensor/actuator. Cells highlighted in yellow are important for this thesis.

above also appear in the research discussed in Chapter 2. For example, the Haptic Edge Display by Jang et al. uses components of Adafruit [JKT⁺16]. In contrast, the research of Deber et al. Hammer Time, uses an Arduino as the control unit [DAJ⁺16]. Phidgets components are, for instance, include TouchMover by [SPB13]. As shown in the two main chapters, the interfaces developed were also implemented using at least one of these components.

In this thesis, housings and moulds have been manufactured in different ways. For example, for a low-fidelity user interface, modelling clay was used to see how it is felt in hand or to explore further form factors. Methods such as rapid prototyping, including 3D printing, was used to create high-quality prototypes. However, it is also worth going to dollar stores or hardware stores. One can find beautiful materials for prototyping, or it may just be useful for finding an idea [Bux07].

Finally, it could be necessary to include a mechanism into a user interface. According to [Tsa00], a mechanism is a device that transforms motion and torque from one or more links to the others. In other words, whenever a motion has to be redirected, a mechanism exists. The methodology of designing a mechanism can be compared to that of an interface design [Tsa00]. Besides, there are already components that contain a mechanism, for example, a linear actuator. A modified scotch-yoke mechanism providing haptic feedback was developed in this work, which turns a rotational movement into a linear movement (see Section 4.4).

In summary, this section showed how and with which resources a new interface could be built. At the beginning, methods for prototyping were described. These methods range from simple sketching with paper and pencil to the creation of high-quality prototypes. Sensors and actuators are required to create interfaces that can interact with humans. Sensors and actuators differ and must be adapted to the particular sense channel. Table 3.2 gives an overview of the variety of sensors and actuators.

3.5 Experimental Design

Whenever a haptic interface has been built, it must be evaluated in order to show quality, limitations and potentials. This is usually done by conducting a user study. Therefore, in the following, an overview of the design, implementation and evaluation of user studies is given. All studies that have been conducted in this thesis have applied theses methods. In general, a user study aims to collect data for later analysis to answer specific questions, be designed. The following paragraphs discuss the methods for conducting user studies.

In order to create an experimental design, certain framework conditions must already have been identified. For example, literature research and the resulting research questions have to be formulated. The research question(s) is/are then used to determine the parameters to be evaluated, to which the user study must be adapted. In most cases, the research question clearly defines which parameters have to be examined [FH03]. In theory, there is always a cause and an effect. This is known as a dependent (effect) and independent (cause) variable. How does the effect change when the cause is changed?

How these parameters are related can be illustrated using the example of the following question (see Section 4.4): How does the haptic feedback on the back of the device affect touch accuracy on the front screen? In this case, the dependent variable is the touch accuracy (cause). We would like to know how touch accuracy performs with varying haptic feedback. Consequently, the independent variable is haptic feedback (effect). Thus, the dependent variable determines the outcome of an experiment. In general, a way of manipulating the independent variable is complicated. There is no standardized approach as to how and to what extent the variable must be changed to achieve a proper result [FH03], [LFH17]. This usually depends on each experiment individually. In this case it helps to perform a pilot study to determine potential values in advance.

A factorial design is an experimental description that contains more than one independent variable. The number of conditions in a factorial design is determined by the total number and the levels of independent variables [LFH17]. Levels indicate how and how often a variable is changed.

For example, studying the typing performance on different keyboard layouts - in this case, the keyboard layout would be the independent variable, and the layouts used in the study (e.g. QWERTY, DVORAK) represent the number of levels. The influence of one-handed and two-handed typing should additionally be studied. The independent variable number of hands also has two levels. Therefore the factorial design is: 2x2. In other words, the factorial design provides the total number of conditions [LFH17].

Another aspect when planning experiments is how to divide the subjects into groups. In human-computer interaction, there are two standard group designs: a within-group design and a between-group design. A within-group design means that all subjects have to participate in all conditions. There is only one group. In contrast, a between-group design means that there are as many groups as conditions. Each group is allowed to complete only one condition [LFH17]. Consequently, a between-group design has more participants than a within-group design. Consequently, in a within-group design participants have to study more conditions than in a between-group design.

Finally, there is a design that is a combination of the two mentioned above. It is called a split-plot design. With a split-plot design, conditions can be combined. This allows creating fewer groups than conditions. Looking at previous typing example, a within-group design would require one group, a between-group design would require 4 groups and a split-group design could require 2 groups. One group is studying both hand conditions on a QWERTY layout, and another group is studying both hand conditions on the DVORAK layout.

Comparing the two approaches, it becomes apparent that from a statistical perspective, between-group is a clearer concept, since each user contributes to only one condition, so there is no learning effect. Besides, parameters such as fatigue and frustration can be better controlled [LFH17]. However, there are limitations of between-group design, the two main disadvantages are of which being: firstly, the performance of several groups with disjoint participants are more difficult to compare; secondly, a comparatively large number of participants is required, which means that the study will take much more time. In a within-group design, individual differences are effectively isolated, as each participant is tested for all conditions. However, there are also disadvantages in using this method. Users might become fatigued and a learning effect have a negative impact on the results [LFH17]. As the split-plot design is a combination of the two mentioned designs, the advantages and disadvantages are to be transferred proportionately to the design.

In exploratory studies with a focus on discovering new trends and initial results, HCI prefers a within-group design [LFH17]. For a more detailed study, all three presented groups are appropriate designs. According to Caine, who has analysed all CHI⁵ papers regarding their group design, Caine reported that 17 within-group, 26 between-group and 25 split-plot design were published in 2014 [Cai16]. Of course, Caine's analysis cannot be generalised, however, it shows some interesting findings.

A further factor we have considered when designing our experiments was the order in which studies are conducted. If all participants complete the study in the same order, results can become biased. The influence of the conditions on each other can bias the results. For example, if condition A always follows condition B, events A and B cannot be considered independent. A and B are dependent. Therefore, it is important to provide the conditions in either random order or a counterbalanced order [FH03]. Latin Square,

⁵Top tier conference in the field of HCI: ACM CHI Conference on Human Factors in Computing Systems https://sigchi.org/conferences/conference-history/chi/ (last accessed 19th April 2021)

for example, is a counterbalanced approach. An extension of Latin Square considers the number of successive conditions, the so-called balanced Latin Square.

To conclude, this subsection has shown which factors have to be taken into account when planning an experiment. These factors were also considered and applied in the experiments performed in this thesis. Of course, everyday factors have also to be considered, e.g. should one stand or sit during the study? Does it make a difference? Or are there other factors that can influence the results, e.g. is the user dazzled and does this influence the study? Additionally, it should be already defined at this stage if and which questionnaires have to be used. Furthermore, it should be clear whether interviews will be conducted after/before/during the study. The studies conducted in the main chapters applied a within-group design. With the knowledge that for some studies, a between-group design might have produced more generalisable results. However, since the experiments conducted were more exploratory in nature, it is sufficient for evaluating the implemented haptic prototypes. This guarantees a proper experimental design to aim the three goals of the research: reliability, validity and importance [FH03].

3.6 Sampling

Before proceeding with more detailed methods for collecting data and statistical data analysis, the population and sample size should be discussed. When conducting experiments, the term population refers to a group of humans for whom the analysis results are valid and can be generalized. The choice of the population can influence the results [CW11]. However, in most cases, the whole population is not able to participate in an experiment. Therefore certain people are selected from the group, either randomly or determined. The number of selected persons determines the sample size.

Skills fluctuate throughout life, the attention span is shorter in childhood, and other skills such as reading and writing still need to be learned. Moreover, abilities also decrease again, for example, the sensory capabilities of older people decrease. According to Lazar et al., most computer applications are designed for people between 20 and 50. This age group also seems to be the target group for our studies [LFH17]. This means that in our case, the population and the resulting sample group should also be in this age range, in which case the very young or the elderly would bias the results.

There are different methods for determining a population, including random sampling, voluntary sampling, all of which have their advantages and disadvantages. Of course, the approach (quantitative or qualitative) also influences the sampling procedure. Whereas

quantitative methods tend to be based on random sampling, qualitative methods tend to use non-random sampling techniques [Kum10].

A specific method of determining samples is called convenience sampling. Convenience sampling is a common practice in human-computer interaction [BACM11]. The advantages of convenience sampling are that the population consists, for example, of all university members and the samples/persons can be collected fast and without much effort using, i.e. a mailing list. The disadvantages are obvious, this sampling technique can lead to bias; for example, if only students, locals or university employees are selected [BACM11]. Caine examined the convenience sampling technique and concluded that it is a method used at ACM CHI. However, the student population cannot be considered as representative for the whole western society [Cai16].

In most of the studies performed in this thesis, convenience sampling was used. Participants were invited using a university mailing list, word to mouth or could register online to participate at a particular experiment. In addition to the studies by Caine and Bernstein et al., it should be mentioned that our studies were voluntary, and the results could have been slightly different if we had paid the participants [SC18]. Nonetheless, there are studies, such as the research of perception thresholds, where it is challenging to bias them, as these depend entirely on a person's sensations. In contrast to perception studies, there are performance studies, and these depend more on motivation and further parameters that can be influenced unconsciously or consciously by the user [LFH17].

In addition to the group of people, the size of the sample should be considered. Various techniques can be used to determine the sample size, these depend, similar to the sampling methods, on the individual approach. In quantitative studies, for example, power analysis is often used, it is an a priori approach to determine the sample size [FH03]. Caine examined not only the group design but also the sampling size of CHI 2014, she has shown that the sampling size ranged from 1-916,000, and the most common number of participants was 12 [Cai16].

Studies presented in this thesis have mostly been performed at our university (Bonn-Rhine-Sieg University of Applied Sciences), and two studies (see Section 5.2) have been conducted at our partner university (Simon Fraser University, Canada) where the population was also determined using a convenience sampling technique.

3.7 Collecting data

Data collection is one purpose of an experiment. An experiment typically follows the design and is part of the evaluation. Data can be collected in several ways: through questionnaires, interviews, focus groups, observation or by collecting performance data. In order to collect data, a study must fulfil several requirements. This section discusses both the data collection methods and the requirements that are important when conducting a study.

Questionnaires The responses to the questionnaires can be given in several ways: either as free text or using a specific scale. The former makes an analysis more complicated because the answers must be interpreted; the latter allows a statistical evaluation. In the field of human-machine interaction, the so-called Likert scale is often used [LFH17]. When using a Likert scale, quantitative data are collected, for example, users give a rating similar to a 5-star rating, which is a common practice on the Internet. In this way, ordinal data are gathered. There are further scales like the Thurstone scale (interval data) or the Guttman scale (ratio data), which were not used in this thesis, because both scales are more difficult for the user to understand and also more challenging to construct.

Gathering data through questionnaires is, besides other methods, considered as a standard in the evaluation of haptic investigations [Gru08]. In HCI questionnaires are often used to query subjective data, preferences or task load [BLC⁺12], [WRS17], [HS88]. In the case of Boring et al. and Wang et al., subjects had to complete the questionnaire after the actual experiment. Studies presented in this thesis also include questionnaires, which were usually asked after an experiment to obtain subjective data about user experience and self-perceived performance.

Especially when using a Likert scale, there are some details to be considered. First of all, the number of options available to choose from; if an odd number is chosen, there is the possibility of being neutral. In contrast, if an even number is chosen, the user is forced to choose, for example, positive or negative. The presentation of the scale can additionally influence the results, there are categorical scales (strongly agree, agree, neutral, disagree, strongly disagree) or numerical scales (from 1, 2, 3...7) [Kum10]. Attention should also be given to cultural aspects and effects of order [CW11].

When designing questionnaires, it is essential how questions are formulated because unclear formulation can lead to confusion [EOEC11] or can influence the subjects [LFH17]. Moreover, according to El-Saddik, the problem using a questionnaire is that subjects may misunderstand the implication of the question or may not wish to report feeling or emotions.

Nowadays, there are simple tools that allow to create a questionnaire and put it online. This has the advantage that users can fill in the questionnaire themselves. This can be achieved with a reliable and specialized tool, e.g. SoSci Survey⁶. These tools further allow us to reach many people with few resources [RSP11].

Questionnaires can either be designed individually or standardized questionnaires can be used. As already mentioned, when creating a questionnaire, it is essential to ensure that the phrasing is clear to avoid misunderstandings. However, there are standardized questionnaires, for example, the System Usability Scale (SUS) is a reliable tool for measuring usability; it consists of ten questions [Bro96]. Another standard questionnaire is the Nasa TLX [HS88], which can be used to determine a task load index. It indicates how challenging the present task is for the user. The use of standardized questionnaires has advantages: results can be compared, and thus a benchmark about the researched system can be given, and they have been used several times before, so the questions have been reviewed quite often. In this thesis, we used both - we created our questionnaires that asked for specific properties, but we also used the two standard questionnaires above to evaluate our interfaces.

Interviews A further method of collecting data is to conduct interviews. Interviews are normally used to collect qualitative data. Advantages and disadvantages described in Section 3.2 are also valid in this case. Interviews are an excellent technique to obtain in-depth knowledge about a specific topic. It is possible to obtain information that might have been lost with a questionnaire, and it is a method to get individual feedback. However, interviews are hard work and can take a lot of time [RSP11]. Apart from that, it is challenging to manage potentially unbounded discussions. Compared to the evaluation of quantitative data, e.g. from questionnaires, the evaluation is more demanding and complex. Identifying important aspects is challenging.

According to Lazar, direct feedback from subjects is a key element in human-computer interaction research [LFH17]. Therefore, interviews were used in initial explorations and after performed user studies. Interviews were mainly done unstructured to get individual feedback. Gathered information and comments were used to enhance the design of new prototypes or to improve the usability of the system and device, respectively. However, interviews were used noticeably less often than questionnaires.

⁶https://www.soscisurvey.de/ (last accessed 19th April 2021)

Focus groups In contrast to interviews, which are usually conducted one-to-one, focus groups are a method of sharing and discussing a particular topic with a group of stake-holders. Again, it is complex to evaluate individual responses. A further disadvantage of focus groups is that there is always the possibility of having one or more dominant characters in a group and that other participants may not be as engaged as they would like or can be. An advantage of this method is that it brings together different groups or opinions. Focus groups promote contact between users and developers or researchers [RSP11]. Within this thesis, focus groups have been rarely used. It may have been used unconsciously in group sessions or discussions, but no data was consciously collected in focus groups.

Observation Similar to focus groups, this technique was used rarely. The subjects were, of course, observed during the experiments and pilot studies, and of course, results of the observations contributed to the respective systems. However, observation was slightly unstructured. In general, monitoring users collects qualitative data and has its origin in ethnographic studies. Caine examined the distribution of data collection methods at ACM CHI 2014 and showed that the observation method was used comparatively as often as interviews [Cai16].

Experiments In Section 3.5, three group designs of experiments were presented. This paragraph shows how experiments can be used, for example, to collect performance data. By performance data, we mean data to which a metric can be applied in order to ensure comparability. In our research area, these are, e.g. accuracy, error and time to completion [RSP11]. This means that during the experiment, we record data that we can later statistically analyse, i.e. the data are quantitative.

It should be mentioned in this context that experiments can also be combined with other methods, described above. For example, in almost all experiments presented in Chapter 4 and Chapter 5 both performance data and questionnaires have been used. With this methodology, both objective and subjective data about the experiment can be collected. Also, numerous related research presented in Chapter 2 make use of exactly this methodology. For instance, a combination of one or more experiments with a questionnaire is used in the evaluation of haptic user interfaces [KLS09], [LOM⁺11] as well as in the evaluation of other advanced user interfaces [NKW11].

There are several methods for conducting user studies. However, especially when evaluating physical devices, a usability test becomes important [LFH17]. According to Lazar:

Often, in usability testing, we're not researching the user, we're researching the interface. We're trying to figure out how to make a specific interface better.

This quote from Lazar describes the situation quite well. Of course, there are pure human-centred usability studies, for example, when evaluating graphical user interfaces. Nonetheless, when researching physical prototypes, other questions arise, e.g. how can the prototype be improved to perform a specific task? Or how can the prototype be optimized for the used sensory channel? This can be applied to the understanding of interface failures introduced by Norman in Chapter 1 [Nor13], both the gulf of execution and of evaluation must be considered.

Caine shows that the assignment of an experiment to a particular category can often be ambiguous [Cai16]. According to the categories of Caine, an experiment includes within-, between-group and split-plot designs. Therefore, in this thesis, usability studies and experiments have been conducted.

3.8 Evaluating

After designing a new interface, creating a prototype and collecting the data through an experiment or a usability study, the evaluation procedure can begin. Evaluation is an essential issue for haptic devices, as it determines contribution, quality and novelty to a certain degree. As mainly quantitative data have been collected in all experiments conducted, the focus of this subsection is on the analysis of quantitative data. However, it also describes the methods used to evaluate qualitative data as well as psychophysical methods.

In order to identify a particular trend in quantitative data, basic statistical methods can be used. These include, for example, mean, median, mode, standard deviation, standard error of the mean and percentiles. This analysis is known as **descriptive statistics** [CW11]. As soon as conclusions are to be drawn, further methods have to be applied. Such methods are summarized under the term **inferential statistics** and allow to generalize results and test hypotheses. The main idea of these methods is to test research hypotheses [FH03]. Usually, hypotheses are specified through the research question(s) [LFH17]

Which method or statistical test can be used depends on some parameters. Consideration must be given to whether the samples are dependent or independent, the scale of the data, whether there is an underlying distribution or not and finally, whether correlations or tendencies are in focus. Since a within-group design was applied in all the studies conducted, there is a dependency within the individual studies. If a between-group design had been used, the groups would be independent, and a different test would have to be chosen [FH03]. There are three main scales of data: nominal, ordinal and interval scales. Nominal data have no order, ordinal data have an order, but the inter-distances between subsequent items are not equal. Interval data have an order and an equal inter-distances. The data collected in the studies are, in most cases, interval scaled [CW11]. For example, in Section 4.2 and Section 4.4, distances to a certain point were collected to estimate the error deviations.

A standard approach used in statistical testing is the introduction of a null hypothesis. The null hypothesis assumes that there has been no effect. In the next step, the probability of the null hypothesis is determined. If the probability is high, no effect has occurred, and the hypothesis is true. However, if the probability is low, the hypothesis that no effect was present is false, and it has to be rejected. This probability is referred to as p-value. There are several statistical methods to test hypotheses. Popular parametric tests are the analysis of variance (ANOVA) or the t-test [CW11]. ANOVA with repeated measurements has to be used when comparing more than two groups or items. In contrast, the t-test is used to compare pairs. Nonetheless, before testing for the null hypothesis, all data must be checked for normal distribution, as this influences the method to be used and is, therefore, a precondition. A popular test to check for the normal distribution is the Shapiro–Wilk test. Most standard statistical tools, such as R or Python, provide the necessary functions for this purpose. These two tools were used for the analysis in this thesis. If the data are not normally distributed, it is still possible to find a central tendency, i.e. a significant difference. The counterpart to the t-test for interval-scaled dependent data that are not normally distributed is the non-parametric Wilcoxon Signed Rank Test [FH03]. In both cases, parametric and non-parametric tests, it is common practice to speak of a statistically significant effect when the probability is less than 5%, in other words, p < 0.05 [FH03].

The previous paragraphs have dealt with the analysis and evaluation of quantitative data. The statistical methods presented were primarily used in all conducted studies. Nevertheless, qualitative data was also used, although relatively rarely. The qualitative analysis aims to transform the unstructured data found recorded from interviews and other sources into a detailed description [LFH17]. It is essential to consider the problem and context. In this thesis, the following methods were often used for data collected through pilot testing (see Section 4.2) or for the analysis of multimedia content (see Section 5.2). Qualitative analysis is an iterative process. The most important aspect is first to identify the key direction or aspects in the data. To get an overall picture, the identified items

need to be compared with each other, which can help to find potential effects of different groups. Several methods can be found in [LFH17]. As mentioned above, it naturally takes less effort to analyses quantitative data, as there is usually a strict procedure and special tools for this. In contrast to the qualitative data analysis, it is obvious that the analysis requires more time and effort.

3.9 Validity and Reliability

All aspects discussed previously in this chapter influence the validity and reliability of research. The basic idea behind validity and reliability can best be explained using a figure (see Figure 3.6), for visualization a target is used [Bab15]. In the first case, the target is always hit at a similar position, which means that all answers are consistent, but have an error. If we now refer to the quality of research, it can be said that the research is reliable, but not valid. Looking at the second case, here, the target is hit everywhere with a similar distribution. The centre of the data is located in the centre of the target, i.e. a small amount gives the correct answer. Moreover, if you look at all the data points, the overall quantity of data is valid, but individual samples are not. This example shows that we have valid data, but it is not reliable. In the third case, both validity and reliability do not apply. The target is not hit with sufficient accuracy. There is a high degree of variability in the data, and the answers consistently miss the target. The fourth case now seems clear: the shooter has high accuracy, she/he hits the target with great accuracy.



Figure 3.6: A metaphor showing the four cases to illustrate validity and reliability, adapted from [Bab15].

For the conducted studies that explore how the implemented interfaces affect the addressed issue, many factors have an influence, as already mentioned. In the first place, the designed and implemented prototype has to fulfil its purpose, Sutcliffe refers to this as requirements validation [Sut03]. In other words, if the feedback is ambiguous or not tangible or distinguishable, it will not be easy to fulfil the purpose, and of course, validity and reliability will suffer. Potential ambiguities can be improved through iterative design and pilot testing, which was also a standard method for the creating the haptic interfaces introduced in this thesis. Back to the metaphor from the beginning, when a shooter gets a bow with which it is impossible to hit the target consistently, his/her result will not be valid or reliable.

As described in Section 3.5, the experimental design has an impact on validity and reliability [FH03]. Perhaps the most prominent factor in a design that affects validity and reliability is the order of the conditions. If the chosen approach of the order of conditions is not described, different results may be obtained when the experiment is repeated. In the experiments presented in the three main chapters, either a counterbalanced or randomized approach was used. Another essential aspect to consider when designing an experiment, especially in our field of research, is the length of the experiment. Since most of the studies, we conducted involved cognitive as well as physical tasks; therefore, care must be taken that the participants are not fatigued [FH03]. This can lead to biased results. Further aspects of experimental design that can improve the quality of research are listed in Section 3.5.

Sampling and the associated population has a fundamental influence on the validity of the research. The research is only valid in the selected population. To evaluate the proposed haptic user interfaces, we have chosen a population from the university environment, as shown later. Further parameters that can influence the validity of research are according to Kumar [Kum10]: the method of data collection can be inaccurate (see Section 3.7), or the statistical data analysis (see Section 3.8) and subsequent conclusions can be incorrect.

In summary, the results obtained with our methodology are internally valid once all the methods have been performed correctly, and the requirements have been met. Examples that have to be considered or possible sources of error are described above. The key to achieving a high internal validity depends on the experimental design [FH03]. However, we cannot guarantee external validity; nonetheless, in the studies, that have bee conducted, we have used some methods that can improve external validity. Pilot studies have always been performed to find out initial trends that have been confirmed in the main experiments. Moreover, the pretest were used to adjust the levels of the conditions to ensure feasible experiments. Likewise, our questionnaires have also been pretest, to ensure that they are clear, unambiguous, and unbiased [LFH17].

Furthermore, we tried to improve the external validity with our sampling method. Therefore, we chose an age span of the participants between 20-50 years, as we intend to generalize our results for this group of persons. Moreover, in many of the studies that have been conducted, we have focused on haptic interfaces. Still, in doing so, we have always tried to create a level of independence from the interface so that it could be easily replaced. This, in turn, improves external validity in respect of different devices.

3.10 Ethics

Our studies are mainly concerned with haptic interaction. However, since we have conducted experiments with users, we have to address ethical aspects as well. All studies were conducted in the field of HCI, there was no risk of injury or harm to the participants. In addition, ethical approval for the study was obtained from the Ethics Department of the Brunel University. There are different ethical categories, and the type of studies we conducted were considered by the Ethics Committee to be classified in the low-risk category.

We respected the dignity of the participants at all times - before, during and after the studies. This implied that users were informed about the study in detail. We also informed the participants prior to the study that they could quit the study at any time without consequences. Participation in each study was voluntary, and no money was paid for participation.

Before the study, users were informed about the above points in a consent. After the study, a debriefing took place. It was specifically reiterated that if concerns arise after the study, participants could contact a representative to discuss their concerns.

In summary, this chapter has given an overview of the research methods used in this research. A methodology was presented, which has been applied to research and develop our haptic interfaces: Chapter 4 and Chapter 5. It was also shown which methods were used to develop new prototypes and how to perform user studies. Finally, the variety of data collection methods and their analysis was presented.

Chapter 4

Haptic Interfaces for Touchscreens

4.1 Introduction

The augmentation of mobile touchscreens to explore the versatile of haptic feedback is examined in this chapter. Therefore, we present three prototypes, all of which provide haptic feedback using an individual approach. While most of the smaller touchscreens already support tactile feedback generated by small vibration motors [FS01], we want to present further potentials for haptic feedback. These include perceptual issues, but also performance and, to a certain extent, user experience in using these interfaces.

Specifically, this chapter is dedicated to the first research question. All three prototypes will highlight the versatility of haptic feedback and introduce novel interfaces. Each interface will address a sub-questions that will contribute in answering RQ1 defined in Section 2.3:

How can touchscreens be augmented to provide haptic feedback that goes beyond tactile sensations?

In Section 2.3.1, it was shown that there are still many possibilities to develop haptic feedback for touchscreens. However, all smartphones are almost all equipped with tactile feedback, so haptic feedback is actually available. However, RQ1 aims to explore further perspectives and explore novel questions. Besides adding haptic feedback the focus in this chapter is on touchscreen interaction. In the last decade, touchscreen interfaces have become popular, enabling a wide variety of applications through established interaction styles. These interaction styles afford finger or pen-based interaction with a multitude of graphical user interface (GUI) elements. Often, interaction is aided by audio and simple vibrotactile cues, which has been shown to improve interaction [KSL+13]. While apt for many applications, interest is also growing to explore other directions, including haptic feedback. However, in particular, the combination of haptics and touchscreen interaction has not been widely studied and is still challenging. For example, sensing the shape of an underlying geometry like a button is still hard to implement.

The structure of this chapter is as follows. Firstly, the three prototypes are introduced and connected to a sub-question. Then, each prototype is considered individually, which means that in each section we present: the implementation, the procedure of the particular experiments, the results and potential application scenarios. Finally, the results are then discussed, and in a summary section, all prototypes are considered from an overarching perspective.

This chapter aims to design and evaluate novel haptic interfaces for touchscreens (Aim 1). Therefore, our objectives are coupled to the sub-questions in order to pursue three clearly different approaches. Using the first prototype seeks to establish the limitations of force feedback provided through an actuated rigid screen (RQ1.1). On the one hand, enhancing touchscreen interfaces with non-visual cues has been shown to improve performance [BCB07]. On the other hand, similar actuated haptic interfaces focus on realising and developing an actuated screen interface [KHSK14], but only barely show the limitations or possibilities of perception [HTE⁺17]. TouchMover, a large actuated stereoscopic display, explores the perception of shapes but it misses the connection to a small horizontal touchscreens [SPB14]. ForceTab, our prototype based on a force-sensitive motion-platform enhanced interface to improve multi-modal interaction based on haptic (force feedback) instead of tactile feedback is introduced in Section 4.2. Extending mobile touchscreens with force-sensitive haptic feedback can enhance performance interacting with GUIs and improve the perception of visual information. In this respect, a user study was performed to determine the haptic perception of different 3D shapes and the perception of different heights. Furthermore, two application scenarios are proposed to explore our proposed haptic user interface.

While ForceTab actuated the entire touchscreen, our second prototype focuses on the interaction above a display and how an elastic material can be used for this purpose. By doing this, a novel haptic feedback sensation is designed that explores the combination of elastic material and haptic feedback (RQ1.2). Motivated by the interaction with elastic materials [PGG⁺12] and the interaction above a display [HBH⁺16], FleXurface was implemented. Furthermore, it includes a novel approach to provide haptic interaction with a pen. The prototype provides a multilayer interaction approach that enables state transitions between spatially above-screen and 2D on-screen feedback layers (see Section 4.3). Our approach supports the exploration of haptic features that are hard to simulate using rigid 2D screens. We accomplish this by adding a haptic layer above the screen that can be actuated and interacted with (pressed on) while the user interacts with on-screen content using pen input. The haptic layer provides variable firmness and contour feedback, while its membrane functionality affords additional tactile cues like texture feedback. Through a user studies, we look at how users can use the layer in haptic exploration tasks, showing that users can discriminate well between different firmness levels.

Both introduced prototypes explore haptic feedback of tab-sized screens, however, they can hardly be considered as mobile systems. Within our third prototype we aim to extend the variety of approaches and show how force or pin-based feedback can be combined with a small mobile touchscreen (RQ1.3). Therefore, we want to introduce a further prototype. HapticPhone is a back-of-the-device haptic feedback approach for smartphones (see Section 4.4). Since touchscreen interaction suffers from occlusion problems as fingers can cover small targets [CDVB17], this makes interacting with small targets challenging. To improve touchscreen interaction accuracy and consequently the selection of small or hidden objects we developed HapticPhone. The proposed approach combines force feedback from the back to enhance touch input on the front screen. The interface consists of three actuated pins at the back of a smartphone. A psychophysical study shows the differentiability of haptic events, which were used in a second study to improve touch accuracy.



Figure 4.1: Potential scenarios feasible with ForceTab, reaching from simple touch input and force input to up a combination of both (left). A photo of one of the first versions of the presented tablet (right).



Figure 4.2: Shows the hardware prototype: the force actuated touchscreen lowered into a table (right). The platform is driven using 3 high speed servos and 4 force resistance sensors mounted underneath the tablet (left)

4.2 Haptic Feedback through an Actuated Touchscreen

ForceTab is the first prototype that explores the versatility of haptic cues in touchscreen interaction (see Figure 4.1). The interface deploys a motion-platform enhanced tablet interface that has been extended with pressure sensors to sense finger pressure (see Figure 4.2). While related systems without pressure sensing exist, an in-depth analysis of actuated tablet interaction techniques, with and without pressure support, is lacking. We present and validate a refined set of feedback mechanisms tailored to such systems.

In contrast to previously mentioned 1DoF movements [SPB14], our approach also supports inclination, according to the position and the underlying geometry. Since most available touchscreens are capacitive, capturing finger pressure without additional sensors is challenging or even impossible [HL13]. ForceTab presents the potential of visuo-haptic devices by exploring the detection rate of different, static and dynamic, 3D geometries. To do so, we introduce a velocity guided detection approach, to determine a relationship between velocity, shapes and detection. Through a set of user studies, we report on both the low-level potential and limitations of the feedback methods to elicit shapes. The studies were mainly conducted to answer the Research Question 1 raised in Section 2.3.1. We are interested in the resolution of perceived shapes and how exactly an individual peak can be identified. In Section 4.2.3, it is additionally shown how visuo-haptic feedback can be deployed through two application scenarios.

4.2.1 Implementation

This subsection provides an overview of the hard- and software components that have been used setting up ForceTab. However, beforehand, Figure 4.3 illustrates how the interaction concept is designed: there is a sensor unit, which is responsible for user inputs, and an actuation unit, which controls specific feedback. Additionally, all associated components and relationships are depicted in Figure 4.3.



Figure 4.3: The interactions principle: from user intention through system feedback to a state perceived by the user, inspired by the seven stages of action from Norman [Nor13].

Hardware

ForceTab comprises of a motion-platform driven by three high-speed digital servos (Savox SC-1257) and four force resistance sensors (Interlink 402). This provides a 3DoF and a low latency feedback system combined with a touch-sensitive display for user interaction. It enables constrained haptic feedback with screen content and supports 3D touch events. The servo motors are arranged in an equilateral triangle with a side length of 150 mm and connected to the platform using ball heads. They are operated at 6 V using an external

power supply, affording movement of 60° in 0.07 s, enabling the platform at high speed (average system response is 75 ms).

The platform is made of medium-density fibreboard and measures 200 mm in length and 170mm. Four Force-Resistance Sensors (FSR) are used to sense force and touch sensitivity at each corner on the top of the fibreboard. Each FSR can detect applied forces in the range of 100g-10kg. ForceTab uses an Android tablet (Asus Nexus 7), with a 7 inch touchscreen, as primary display device. All low level introduced hardware components (servo motors and FSR) are connected to a micro-controller (Arduino Uno), that controls the servo motors and receives force events. The system is driven by a conventional desktop computer running a Linux system (Debian). Our prototype is designed in a way that with minor modifications, any Android smartphone or tablet can be used with the platform. Therefore we have developed a dynamic holder for the platform, which can easily be printed in 3D.

Software

A desktop computer (manager) handles time-consuming tasks and the communication between our touchscreen and the micro-controller. It receives touch events over Wi-Fi (from the tablet) and forces events over USB (from the micro-controller). To adjust the servo motors, signals (pulse-width modulation) are sent back to the micro-controller. A ray-based approach estimates the position of the platform. More precisely, a ray is cast into a 3D scene at the location of the finger using a virtual orthographic camera model. This 3D scene can be seen as representative for the 2D scene display on the touchscreen. If there is an intersection between ray and geometry, the position and the normal at the intersection can be calculated.

In a final step, we map the estimated normal to our platform. The position of each servo is determined using forward kinematics. Inclination mapping of the platform is a culling process, meaning that angles which are larger or smaller will be mapped either to the maximum or minimum inclination. ForceTab can map heights up to 32 mm and an inclination up to 14°, according to the fingers position and the underlying geometry. To simplify height computation, all scenes are pre-processed by scaling them to a specific height. This height corresponds to the height constraints of the platform.

Force input is computed using a distance-weighted function to linearly interpolate over all four force sensors. To calibrate the force sensors, all force values are requested whenever there is no touch event and thus no motion event, so that these values are considered as a new initial point. Normalizing the values within a logarithmic scale enables us to precisely capture pressure events.

Our implementation supports a velocity guided mechanism to enforce users to interact within a pre-defined velocity range. To do this, users were asked to follow a animated visual disc with their fingers. This mechanism increases the controllability of the experiment and, as introduced in Section 3.3, the levels of the condition can be specified more precisely. Thus, disruptive factors can be excluded and the data is less noisy. The system can adjust the velocity of an object on both touchscreen and desktop screen. This is realized using a one to one mapping from the touchscreen to the main screen. Through the described configuration of hardware and implementation, ForceTab is able to support 3DoF haptic interaction, namely movement in upwards and downwards (height) as well as roll and pitch (gradient). In the following user study, we focus on how this feedback can be employed.

4.2.2 Experiment - Low Level Perception

This study was designed to show the possibilities as well as the limitations of the implemented haptic interface (see overview in Figure 4.4). It falls into the category of a lower-level exploratory perception study. For this purpose, three different characteristics are investigated: combination of resistance levels (pressure sensitivity) and force feedback, the perception of geometric shapes and the differentiation of heights. The aim is to determine a perceptual resolution of geometric shapes. What can users be perceived with this haptic feedback approach? In the following it will be referred to as force, horizontal and vertical resolution.

The force study has been conducted to test five hypotheses:

H1: Haptic feedback will help users to differentiate three resistance levels.

While for the horizontal resolution study it was assumed that the geometric shapes are perceivable and it was hypothesized that:

H2: Geometric shapes can be best perceived with free exploration.

H3: Larger shapes (a) and slower movements (b) help to perceive the geometric shapes better.

Finally, the following hypotheses will help to explore the vertical resolution of the haptic interface:

H4 Smaller differences in height will be more difficult to perceive.

H5 The tip of a pyramid can be better localized than that of a Gaussian shape.



Figure 4.4: A schematic overview of the study conducted. The diagram shows the three sessions for determining force, horizontal and vertical resolution.

Participants

This user study was performed as a within-subjects study in which 12 participants took part (age 23-38, 3 female, 9 male). All participants took part in the study voluntarily. No funds were paid for participation. Furthermore, all participants were informed prior to the experiment that they can stop the study at any time without consequences.

Apparatus

Participants were seated and wore noise cancellation headphones, to block sound patterns from the servos that could provide unwanted cues. In order to exclude visual cues a wooden panel was placed between the tablet and the user, so users could not see the interaction device actuation (eyes-off). ForceTab was integrated into a table so that the display was at the same height as the table surface (see Figure 4.2). The users were asked to rest their palm onto an ergonomic wrist pad. Using this pose, users would receive feedback primarily over the finger. This avoided potential bias when the hand would need to be held in mid-air, where feedback would be partly perceived over the wrist-arm lever system too.

So that users knew where their finger was on the tablet (which was not visible to the users), the position of the finger - the touch event - was mapped on a separate (desktop) screen visible to the user. A circle was used to indicate the finger position. In order to validate the detection rate, users were prompted to select the perceived shape or force value displayed on the desktop monitor after each condition.

We conducted a pilot study to reduce the number of conditions and thus exclude any influence on fatigue on the results. In the pilot study we investigated whether it makes a difference if perceived shapes are convex or concave. Six users validated the potential of rectangular concave and convex shapes. As a result, both shapes had a similar error and detection rate: Users detected concave shapes with 85% accuracy versus convex shapes with 82%. So convexity or concavity does not seem to be a difference in the detection rate. Subsequently in the main study we concentrated on the only one convex shapes. In the following, the term resolution refers to the detail (size and shape) of a single convex shape that can be perceived by the user.

Design and Procedure

Force In the first part of the study, users were asked to differentiate between two squares, each square representing a resistance level. As already mentioned, there were three levels: sensitive, middle, hard. The three levels were found to be differentiable through a pilot study. When a user exceeded the resistance criterion, a mechanism was triggered; the touchscreen moved 10 mm downwards and back to the starting position within 750 ms. The idea behind this study is to find out how well the participants can distinguish between the levels. Each participant completed 12 trials in a randomized order. To estimate the three pressure levels, we asked participants to press two squares (randomly paired levels) sequentially. To avoid possible cues, the study was conducted eyes-off. Therefore, finger position and the two squares were shown on a separate screen. A 2AFC response paradigm was used, which means there was always a unique solution, and users were asked to select the square with higher resistance.



Figure 4.5: Depicts the 4 patterns used for estimation horizontal resolution (a) rectangular (rect), (b) saw, (c) triangular (pyra) and (d) semi-ellipse (bump) shape).

Horizontal resolution The second part of the study focused on the perceived level of detail in the horizontal direction, and employed a $3 \ge 4 \le 3 \le 2$ factorial design. It deployed the factorial combination of 3 exploration types (guided finger movement, simulated patterns and free exploration), 4 shapes (see Figure 4.5), 3 widths (35px = 2.8mm, 57.5 = 4.5mm, 80px = 6.5mm) and 2 velocities (3 sec and 4 sec per trial). To avoid a constant change between the configurations we have decided to group all trials according to the exploration types. Thus, each participant completed $3 \ge 24$ trials in randomized order. We specifically chose these geometric shapes because they represent a wide spectrum. In this way, symmetrical, asymmetrical, rectangular and round shapes could be tested, while keeping the number of shapes and the length of the study manageable to avoid fatigue.

Within **guided** finger movement, users were asked to follow a straight line from left to right with a specified velocity. This velocity was visualized on the separate screen using a circle. Users were asked to follow the circle, and their finger was displayed as a second
circle to provide a reference frame. In the middle of the guided path, the shapes with the described parameters appeared.

Instead of moving the finger over the touchscreen, in the **simulated** mode the feedback passes underneath the fingertip (like a wave), while the finger is held at the same position. To activate the system the user has to move her/his finger onto a predefined rectangle at the centre of the screen. The rectangle changed colour to indicate the user that a shape would appear.

Finally, since the previous exploration types were based on a specific velocity a third type was added. Within this exploration type the users were asked to **freely** explore the shapes for 5 seconds. After each condition users had to select the perceived pattern. For this purpose all patterns were displayed on the separate screen and users could select the perceived one.

Vertical resolution The last session of the low level perception study looked at the estimation of the perceived vertical resolution, and employed a 4 x 3 factorial design, using the factors: 4 elevations (varying between 3mm, 5mm, 7mm, 9mm) and 3 shapes (diameter/side length 35mm, a box shape, a pyramid shape and a bump (Gaussian) shape). No equal heights were allowed, ensuring that there was always a unique solution. Each user completed 18 trials in randomized order. In contrast to our prior study we enabled inclination, as we wanted to investigate whether this had an influence on finding the tip of a pyramid or a Gaussian shape.

Users were asked to find the highest point on the tablet. Again, a circular shape on the second screen indicated the finger position on the tablet while no additional information was displayed. In each trial, two elevations with different heights but same shape were mapped on the tablet with a random position but no overlaps - equal heights were not allowed. When the user has found a maximum, he/she can submit the position of the finger by pressing the space bar the distance to each maximum local and global was recorded. After that, similar to the previous study, users are invited to select the shape on the screen.

Results

Force All users could differentiate between all three levels correctly, which means that the detection rate of the three force levels was 100%. Hypothesis (H1) can be considered true. The 12 users always chose the correct square. This also reflects the consistency of our system. This further enables the system to response discrete to each level, for instance,

	rect	saw	bump	pyra
rect	81%	11%	0%	8%
saw	7%	80%	4%	11%
bump	9%	9%	35%	44%
pyra	3%	0%	61%	35%

a button with three states. Looking further, ForceTab is also able to map continuous force input to height, for example, navigating through a hierarchy using three velocities.

Table 4.1: Detection rate and error rate of the 4 shapes. Orange cells depict the detection rate of the considered shape, additionally each column depicts the falsely detected shapes. The table indicates the confound of pyra and bump shapes.

Horizonal resolution Table 4.1 shows the overall detection rate over all trials and indicates that rectangular (rect) and saw shapes were detected very well (81% and 80%). However, semi-ellipse (bump) and triangular (pyra) shapes were detected poorly (35% and 35%). This is due to users tending to confuse semi-ellipse and triangular shapes. It was therefore not correct to assume that all shapes are perceptible.



Detection rate per exploration type and shape

Figure 4.6: Overall detection rate of the three exploration types (free, guided and simulated). The graph shows the difference in detection rate including both velocities. The shapes that were confused have been marked in grey.

Figure 4.6 illustrates the confusion of pyra and bump-like shapes in a further way. The figure contrasts the relationship between exploration types and the four shapes in terms of detection rate. To clarify the difference between the shapes, an ANOVA was performed, which highlighted that a significant difference indeed exists (F(3, 140) = 56.66, p < 0.001).

Post hoc analyses were done with Tukey's HSD test and showed that there is a difference between all pairs except rect-saw and bump-pyra. Since the confusion of bump and pyra shapes affects the influence of the exploration types, they were left out of our analysis. Hypothesis H2, the users can perceive geometric shapes better when they have the opportunity to explore them freely must be rejected. A significance analysis showed that the way of exploring does not influence the detection rate.



Detection rate per width and shape

Figure 4.7: Box plots showing the detection rate per width (35px, 57.5px, 80px) for each shape and overall interaction metaphors.

Furthermore, the width of the shapes was examined. Figure 4.7 compares the detection rate of the different shapes at the three widths. Since bump and pyra shapes would again bias the results here, they have been omitted in the analysis of the width. For the analysis, all the same widths of rect and saw shapes were combined. So we only investigate the impact of the width, independent of the shape, on the detection rate. ANOVA showed a significant difference between the three widths (F(2, 69) = 3.3, p < 0.05). A post-hoc analysis using the Tukey HSD indicates that there is a significant difference between 35px and 85px (p<0.05). Part a of the third hypothesis (H3) can therefore be accepted, the size of the geometric shape has an influence on the detection rate; the larger the shape, the better it can be detected.



Figure 4.8: Detection rate at different velocities (left) and the learning rate comparing first and second half of the free exploration method.

The results show that the faster, the less accurate the users were - as expected. Although the detection rate at the faster velocities was on average 12% worse, a significance analysis could not detect any statistical differences (see Figure 4.8, left). Thus, it can be concluded that a trend has been identified, but to fully accept part b of H3 further research is needed.

In addition, there is a learning effect, as users detected rect and saw shapes in the second free exploration with a 12 % higher accuracy compared to the first free exploration (see Figure 4.8, right). Due to the confused recognition of bump and pyra shapes the performance and learning effect analysis is not indicative. It only shows that within the given interaction metaphors and velocity users can differentiate between 3 different shapes.

The study of perceived horizontal resolution showed several important outcomes. Firstly, the impact of the shapes' properties, like form, are an essential issue when estimating the perceived detail and especially designing applications for ForceTab and similar devices. Distinct properties, such as strong discontinuity, can be accurately recognized by almost all users. Participants reported they are able to observe particular properties in the shapes, whereas similar shapes with no unique features are hard to distinguish.

Secondly, there is a correlation between velocity and width, showing that shapes with a width smaller than 57.5px ($\doteq 4.5$ mm) are not feasible for finger interaction. To increase the detection rate we would recommend an even larger width (80px $\doteq 6.5$ mm).

Lastly, as users habitually interact with touchscreens in a visual guided way - meaning that visual elements are the main factor controlling finger position and velocity - users have to adapt to the novel interaction properties. This means that when adding haptic cues to touchscreens users should be motivated to adapt finger velocity to the underlying geometry (fine shapes require slow movements and coarse shapes allow faster movements). A finger velocity of around 0.025 $\frac{m}{s}$ evidenced good results for the proposed scenarios.

Vertical resolution Users were able to find the global maximum with an accuracy of 87%, which indicates that even small differences could be perceived by the users. Looking closer to the relative differences in heights (2mm, 4mm and 6mm), on average users were able to find the global maximum with an accuracy of 78%, 90% and 90% for each height, respectively. Smaller differences in height are therefore more difficult to perceive (H4).



Figure 4.9: Distance to the global and local maximum for each shape (left). Detection rate of the global maximum shape is depicted in the bar plot (right).

Boxes were either found or not, but for pyramids and bumps we can clearly determine the distance to the maximum. Figure 4.9 depicts this distance to the global and local maximum grouped by the three shapes. Figure 4.9 on the left shows that the global maximum was found with a lower error. A significance analysis shows that the global maximum was found significantly more accurate (F(1, 214) = 11.4, p < 0.001). The median distance to the global maximum for pyramid shapes was 2mm and for bump shapes 4.8mm. The maximum of the pyramidal shape was felt significantly more accurately than bump-like shapes (F(1, 214), p < 0.01). H5 is true, the tip of a pyramid can be localized more precisely than that of a Gaussian shape. We assume that this has to do with the sharp tip of the pyramid compared to the blurred top of the Gaussian shape. Furthermore, Figure 4.9 shows the detection rate of the three shapes, which was high for most but not all shapes (box 94%, bump 93% and pyramid 61%).



4.2.3 Application Scenarios

Figure 4.10: Depicts two possible application scenarios. A layered image stack of a anatomy data set (left) and a physical plane with gravity (right).

Lastly, we combined pressure, position and visual feedback within two different application scenarios (see Figure 4.10). In the level of detail viewer scenario, we visualized a layered image stack of an anatomy data set. This dataset was chosen because we believe that our interface is well suited for educational use because visual content is combined with physical movements. Users would be able to better memorise learning content, which of course needs to be proved. Furthermore, we chose this dataset because the different levels are very clear in this dataset. The position/height of the tablet indicated the different layers. From the muscles, over veins to the skeleton, all levels were accessible through different finger pressure. In addition to the vertical layers, each layer could also be accessed individually for additional layer information. As such, the system employed 2 dimensions of information, one in vertical direction (layered data) and the other around the pitch axis (additional information). In the **physical interaction plane** scenario the tablet acted like a physical plane. Based on the angle of inclination, the elements displayed on the tablet moved to the deepest point of the plane, based on the friction and gravity parameter of each object. Small obstacles were integrated to show collision detection. We designed two interaction metaphors to access either different levels of detail (first scenario) or to manipulate the physical plane (second scenario). A force-based metaphor (finger pressure) and a position based metaphor (finger position) was implemented to control the touchscreens position.

Users explored each scenario for at least 90 seconds. Thereafter, the users were asked to answer a set of questions (7 point Likert scale, ranging strongly disagree (-3) to strongly

agree (3), see Questionnaire 1). Users rated state based feedback well in both the layered image viewer M = 2.6/SD = 1.2) and physical plane (M = 1.8/SD = 0.7). In addition, the novel force based metaphors (M = 2.8/SD = 1.4) gained almost the same result as position based metaphors (M = 2.7/SD = 1.6). The overall satisfaction with the system was rated with (M = 2.2/SD = 1.1).

4.2.4 Discussion

The results show that the proposed interface is able to generate different heights/elevations and resistance levels that can be perceived well by the users. To answer RQ1 (stated in Section 2.3.1) we conducted a user study with three different sessions to determine force, horizontal and vertical resolution. As expected, users have more difficulties differentiating between smaller differences than with larger ones. We have additionally shown that confusion can occur with smaller shapes, especially if they are similar. Moreover, we have examined four shapes with different widths, for unique shapes we found that if the shape has a width larger than 4.5mm users were able to perceive them robust within our device.

As the top of pyramid shapes was detected more precisely than the top of bump shapes, we expected that the detection rate of pyramid shapes should be equally or even higher compared than the other two shapes: we thought a user might draw conclusions from the shapes top, which was actually not the case. However, finding the top of pyramids was underlined by feeling a peak at the maximum. Pyramids were often perceived as bumps (34%) but not the other way round (4%). However, in general users commented that inclination of the actuated platform helped to detect the shapes.

With regard to the sub-question (What are the limitations of force feedback provided through a rigid screen?), it can be concluded that complex geometric shapes cannot be represented, as the discrimination of simple geometric shapes has partially failed. Which is in accordance with previous system [SPB13]. Looking at the detection of shapes, haptic feedback could conceivably support visual feedback. In contrast, to locate different locations of the touchscreen, the proposed approach is well suited. Users could find the maximum well regardless of the geometric shape. In this case, pure haptic feedback is sufficient. In combination with the presented resistance levels, the proposed approach would be ideal for the use of realistic buttons.

4.2.5 Conclusion

Within this section we looked into the pressure levels, as well as vertical and horizontal resolution users can perceive while using our actuated platform. This understanding is crucial to design and develop real applications scenarios. Interacting with mobile actuated displays is new to almost all users. In our results, we identified system constraints as depicted by the boundaries of the system feedback, while also highlighting user's perceptual capabilities in frame of the provided feedback range. While not necessary for many GUI element interactions, as a next step it would still be beneficial to perform a just-noticeable difference experiment to further study finer-grained feedback levels.

We presented results of feasible mappings that need to be adjusted to particular applications. Our results create an understanding of the possibility and limitations of comparable platforms such as presented by Kim et al. [KHSK14]. The system by Sinclair et al. [SPB14] showed a better detection rate in 1-DOF, but in comparison to our study, users were able to see the visual display device which strongly supports the identification of shapes.

Another novel aspect is the introduced force-sensing method. An actuated force-sensitive visuo-haptic platform can enable a set of novel application scenarios and interaction metaphors, which we partly explored in the application section. For instance, in education ForceTab can enhance knowledge transfer: data can become "tangible" to support users to understand relations and conditions better. Based on users' feedback, we will extend ForceTab to a broader range of multimodal feedback, in particular vibrotactile and audio feedback. Both can help to further enhance perceptual issues of mapped geometry. For example, tactile and audio cues might enhance the perception of edges and borders, which is an issue of the current system. Since we analysed perceived resolution and proposed two scenarios of state based interaction, in the future we would like to focus on the third feasible interaction component: continuous tracking of force and position to explore additional object properties, like stiffness and surface texture.



Figure 4.11: Our multilayer system consists of a stretchable flexible surface above a tablet display. (A) The bars (light grey) stretch the flexible surface to change its haptic properties, which can be further augmented through vibration (dark grey: speaker, vibration motor). (B) The user can press down on the flexible surface to explore the haptic properties. (C) This enables the simulation of, e.g., different brushes and paper surfaces.

4.3 Elastic Feedback above a Touchscreen

In our daily lives we frequently explore different kinds of haptic properties. In the previous subsection (see Section 4.2) we investigated how such properties could be reproduced and perceived by a user through a rigid screen. The limitations and benefits in using a rigid screen have been clearly shown.

Research in this area includes considering several properties that can be perceived through our sense of touch, including kinaesthetic and tactile cues such as texture, firmness, temperature, weight, enclosure and contours [LK09]. While interacting in the real world, we experience haptic features that are constrained by a surface. However, we also explore features more spatially. For example, we pass our fingers over the contour of an object or experience the spring force of a flexible tool (e.g. a brush) (see Figure 4.11).

In general, haptic rendering can be very useful for certain applications (see Section 2.2.1), e.g. for medical training [KPP13]. Tactile cues can improve performance, e.g. typing on a touchscreen [BCB07], improving expressiveness [BSLM01], accuracy of perception and ambiguity [LGH+03]. It is obvious that not all haptic properties can be easily rendered on a rigid tablet screen. Therefore, in the following subsection, a novel user interface will be presented, which shows the versatile of haptic feedback on touchscreens by introducing an elastic multilayer approach.

4.3.1 Implementation

To investigate further aspects of versatility of haptic feedback and thus explore RQ1 we have developed our second prototype: a novel multilayer interaction approach for pen-based tablet interaction have been designed, developed and explored. It offers a haptic layer to support the exploration of haptic features and direct manipulation of visual screen content. The interface consists of a transparent stretched elastic material attached above a tablet screen. Due to the gap between the flexible surface and the tablet, the system augments pen interaction with the rigid tablet screen into the interaction space above the display itself. We change the stiffness of the layer by stretching it, which can be sensed through pen interaction. We use membrane properties of the haptic layer to relay vibrations, also in cohesion with audio cues. The motivation for adding audio-tactile cues is twofold. Firstly, previous work has shown high relevance of vibro- and audio-tactile cues for improving performance [BCB07], [LMBS07], [VDH14]. Secondly, some haptic exploration tasks can depend on audio-tactile information (e.g., texture).

Through the gap between the flexible layer and the touchscreen, the pen can be pushed down to the screen. The interfaces allows spatial exploration of the trackable space above the 2D display and at the same time enable haptic feedback to the pen. Users can easily switch between the two layers as the haptic layer, under low tension, can touch the rigid screen with sufficient pen pressure. The transition between flexible and rigid states is an important feature of our approach: e.g. such transitions are often used by engineers, designers and architects while using devices, tools and systems, see a discussion of state changes (and shape-changing UIs) [FLO⁺12].

The proposed multi-layer interface supports the research of material and spatial haptic properties in flexible as well as in rigid space. It is able to simulate firmness and shape contours by using spatial and flexible properties. It is also able to render texture properties (roughness) using the mounted membrane. In this way, FleXurface can generate tactile feedback, as the surface moves slightly when vibrated. Furthermore, the elastic layer can create smooth feedback that can be associated with soft tools. Finally, the rigid tablet screen can be used for direct manipulation, while it can limit pen movement to a 2D layer. As a result, it is possible to recreate a real event, e.g. the pressing of a soft or spring button before the actual activation, i.e., state transitions between flexible and rigid states (see Figure 4.12). To obtain this functionality, we seamlessly blend between the dynamic force haptics above the screen (which simulates pushing down the spring-loaded button) and the rigid layer that constrains the downward movement to select (click) the button. Such dynamic force buttons can create different types of buttons with different functionalities. For example, a delete button may require more force to operate, so that the user is warned and it becomes more difficult to activate it accidentally.

System

The main idea of the interface is to provide a further approach on how to design haptic feedback. The prototype is based on a transparent, flexible material with variable tension over a touchscreen. By stretching the elastic material, it is possible to achieve different levels of tension, which is useful, for example, to simulate firmness. The touchscreen is used for visual feedback, i.e. displaying information. In contrast to the elastic feedback layer above the screen, the rigid display can be considered as a second feedback layer, which provides rigid feedback. The management of dynamic tension of the elastic material and the mounting of the different components, such as the servos was realized with 3D printed parts. The housing is built from medium density fibreboard. The software is based on Unity3D, and a client-server system controls the interface. The server (tablet) sends its current touch event via WLAN to the client (laptop). The laptop handles all computations for controlling the motors, it is therefore responsible for the haptic rendering. Thus all actuators (servo and vibration motors) are controlled by the client using a micro-controller (Arduino Mega).



Figure 4.12: Stress-strain curves at 3 measured positions (1,2,3). Due to symmetry, positions 1 and 4, and 2 and 5 produce about the same results.

Display

A Samsung Galaxy Tab 8 tablet, with an 8" touchscreen was used as the main display. It natively supports interaction through a pen, as well as through direct finger input. The pen's 2D location on the screen can be tracked even if the pen-tip stays within a range

of about 12 mm above the display. As the S-pen is pressure sensitive, the interface is able measure the relative pressure placed on the tip of the pen for interaction purposes. Pressure sensing is reliable at all tension levels: Calibration measurements were collected at 6 mm penetration depth under all tension levels. (see Figure 4.12). For example, the interface is able to trigger a button feedback reliable at the same penetration depth under all tension conditions, which is important for repeatability.

Elastic haptic layer

A fully-transparent 0.8 mm silicone sheet was used to provide haptic feedback (Silex Superclear 40). It is glass-clear so that the content displayed on the tablet screen is fully visible. The flexible material is mounted between two 3D printed bars, which are able to stretch and release the material above the screen. Directly onto the flexible material, a bendable ultra-thin plastic layer was placed. This second ultra thin layer was necessary because the interaction between pen and elastic material caused too much friction, which led to undesired effects during the interaction, for example sliding with the pen over the surface was difficult. The second transparent layer is mounted on the sides using a thin highly flexible material (neoprene), which allows the plastic layer to deform along with the main elastic layer (see Figure 4.13).



Figure 4.13: (Left) Schematic side view of the system, showing the pulling mechanism to change the variable compliant tension, (Right) exploded diagram of all the system components. Light grey objects with reduced contrast (to improve image clarity) are further instances of the main "highlighted" functional components.

The material protrudes about 2 cm beyond the edge of the display, thus avoiding the appearance of wrinkles. During our experiments no such disturbing wrinkles were observed. The distance between the flexible surface and the touchscreen is 10 mm. In all studies, users looked at the screen almost perpendicularly, limiting any parallax effects that might be caused by the space between elastic layer and screen. There were no disturbing visual

effects observed. While parallax can be a problem when interacting with oblique viewing angles, we can also emphasize that parallax effects are implicitly attenuated when the user pushes the pen down towards the rigid display surface to manipulate the screen content directly.

Each of the two 3D-printed bars that stretch and release the elastic haptic layer are connected to two servos using 0.5 mm nylon cables. These servos were mounted below the surface on which the touchscreen is located. The cables are tightened between the bars and the digital servos with adjustable spanners. This allows the system to be dynamically readjusted. The bars were operated with digital servos (Align DS655). An external power supply was used to provide the operating power of 6 V.

All servos have a theoretical motion speed from 60° in 0.05 s at 6 V, but under load speed is likely to decrease. The servos are controlled synchronously, i.e. one single control signal is transmitted to all four servos. This results in a symmetrical tension in the direction of the bars. This tension/stretch can be defined by the servo angle. The surface can be stretched up to 12 mm (at 130°), in comparison to the neutral position where the flexible surface is flat (at around 2 mm stretch, 65°). This results in a system stretch range between 2 mm and 12 mm.

Within the interface, the flexible haptic layer can in general be used in two modes: static or dynamic. In static mode the servos build up a surface tension before the pen reaches the surface and maintain this tension regardless of user actions. In contrast, the dynamic tension is changed by user events. In this case, two events can generally cause tension changes: pressure events (e.g. pressing a button) or sliding events (e.g. exploring an element). Note that an increase in tension on the flexible surface actually moves the pen upwards, away from the display, as the pen is pushed upwards (see Figure 4.14).



Figure 4.14: Feedback types afforded by our system based on different forms of actuation and the flexible or rigid state of the system. From left to right: firmness, upward force, vibration and sound (including vibration), and rigid.

Membrane properties

A further characteristic of flexible materials is that they serve as a membrane: Transmitted vibrations can be easily felt when the pen touches the vibrating material, as the surface is under sufficiently high tension. To explore the potential of vibrations, a small loudspeaker and a vibration motor were attached to the flexible haptic layer next to the tablet (see Figure 4.13). The fixed membrane centre of the loudspeaker is connected to the flexible surface using a rubber ring on the loudspeaker membrane. In this way, it is not only possible to reproduce audio, but the flexible surface also functions as an extended speaker membrane and offers the possibility of displaying tactile cues in accordance with the audio feedback that can be perceived through the pen. The speaker is driven by a tiny 3 W mini amplifier mounted underneath the bottom plate and connected to the tablet line-out. Obviously, audio from that loudspeaker can normally be heard, but not if the users wear headphones. For tactile feedback independent of audio, a Precision Microdrives 12 mm/3 mm pancake vibration motor (from 0-1.4 g (amplitude) and 0-300 Hz (frequency)) was fixed onto the flexible material.

Calibration

A calibration is required to determine the characteristics of the flexible layer. This includes several components that have to be considered. First of all, the relationship between servo motor and tension in Newton must be determined. For this purpose a pressure distribution of the transparent elastic layer has been computed. Therefore a pressure sensor (Interlink 402) with a small disk similar to the pen-tip diameter in a mechanical contraption was mounted above the tablet. The contraption allows to adjust the depth of the sensor, for calibration this was exactly 6 mm depth. During the generation of the stress-strain curve, symmetry properties were used. Therefore, 3 locations (1, 2 and 3 - see Figure 4.12) had to be measured. The three stress-strain curves of the three locations are also depicted in Figure 4.12. It can be seen that each of the three measuring points has a slightly different curve. We measured pressure at the full tension range of the flexible material and calculated both the stress-strain curves and pressure accordingly.

Since the pen also has a pressure-sensitive tip, the next step is to calibrate the pen. To do this, we mounted the S-pen in the mechanical contraption and measured the pen-tip pressure again under the full tension range, at the middle position (number 3) at 6mm protrusion depth. The pen-tip was also matched with a force profile using the aforementioned method. With these calibration steps it is possible to determine which force is applied to the pen as well as how deep the pen has penetrated the flexible material at all tension levels. Furthermore it is possible to convert the servo positions to the applied force by using the stress-strain curves. Position 3 was used as main feedback area in our user studies. To avoid mis-calibration or bias, the top plastic layer was fixed during both calibration and studies.

4.3.2 Experiment - Perception Study



Figure 4.15: A schematic representation of the psychophysical study conducted to determine the perception threshold.

To look into further aspects of research question (RQ1), we explore the potential of our flexible layer to convey different haptic properties. Therefore we looked closely at the perception of firmness (by varying surface tension through stretching the material). In doing so, a just noticeable difference (JND) experiment was performed. A schematic representation of the study can be found in Figure 4.15. In our case, we wanted to know how well users can perceive and differentiate between different tensions of the flexible haptic layer. In a subsequent subsection a potential application is presented, it shows in various ways how different feedback can be mapped to particular interaction components.

Prior to the main study, we conducted a pilot study to confirm that all parameters were set correctly and to ensure that the feedback from the system was reliable and that all components were performing robustly. Unlike to the first study in this chapter (see Section 4.2), where the index finger was used to interact with the touchscreen, the following experiment focused on pen interaction. Furthermore, we see a press event, for example when using a button, and a sliding event, for example in drag and drop gestures, as the main interaction metaphors of the interface. For this reason, these two metaphors will be examined more closely and it is hypothesised that:

H1: Interaction metaphor (slide, press) has an impact on the perception threshold

Participants

12 subjects (one female) with an average age of 28.67 years (SD = 5.94) and a range from 20 to 39 years participated in the experiment. Most had normal vision; 4 people who wore glasses or contact lenses. The majority had at least some experience with pen-based interaction, 9 participants had used them sometimes (75%) and 2 frequently (16.7%). All the participants took part in the study voluntarily. No money was paid for participation. Furthermore, all participants were informed before the experiment that they could withdraw from the study at any time without consequences.

Apparatus

Both the experiment and the complete interface (see Figure 4.16) are designed for landscape usage. Throughout the study, users wore noise-cancelling headphones playing white noise to filter any potential external audio cues. A wooden box covering all parts except the screen was installed to avoid visual cues such as moving side bars. The study participants were able to place their hands on this box to enable an ergonomic posture. All users operated the system in a sitting position.



Figure 4.16: Apparatus, with sound and visual cover (Left), top and side/top view of setup showing art application (Right).



Figure 4.17: Three representatives of the conducted staircase protocol. When a line descends, the difference is correctly detected, when a line ascends, the difference is incorrectly detected.

Design and Procedure

Study 1 followed a standard staircase protocol for calculating the JNDs, in which users had to compare two different tension stimuli, stating if they felt different or the same [JT13]. This adaptive method was used to identify at which threshold the users could still feel a difference, by iteratively increasing or decreasing the tension based on the users last response. Users performed the protocol for both pressing and sliding separately, as both are considered as typical metaphors for the interface; both pressing and sliding tasks were repeated twice. The base tension ("standard value") was chosen at 2 mm stretch (65° servo angle), where the surface is flat. The staircase procedure started at an stretch of 10 mm (110°) and went down in steps of 5° until the user has entered a false response a reversal point. This can either be when the user does not notice a difference between the "standard value" and the "reference value" although there is a difference, or when the user perceives a difference but the two stimuli are the same. After the first reversal point the step size was reduced from 5° to 1° to get a more accurate result. We chose 110° (1.63N) as the starting point, as it offers a tension clearly different from 65° (0.74N). The procedure was stopped after 50 steps, or 12 reversals. These equal distance servo steps can be used as the relationship between stress differences (N) and servo angles is linear, as can be seen in Figure 4.12. In order to make the procedure more clear, we have selected three representative to visualise the applied staircase protocol (see Figure 4.17).

Besides haptic feedback, visual feedback was additionally provided. Therefore, a button was shown at the centre (position 3, see Figure 4.12) that either had to be pressed or

which users had to slide the pen over. For sliding, a marker was moved along the button that the user should follow, starting at 1.4 cm to the left of the button and ending 1.4 cm right of it. This velocity guided method, similar to the interface presented in the previous section, is intended to ensure that all users perform the activity at almost the same speed. Pressing and sliding was performed at 6 mm surface depth. A colour indicator was provided (button or moving marker turned green), once the correct pen depth was achieved. We implemented this by reading the pen tip pressure and comparing it to the calibration measurements.

In both tasks, sliding and pressing, all users had to lift the pen entirely from the surface. This guarantees that users do not receive feedback when switching to the second stimulus. Furthermore, when switching between the simulated buttons, the pressure was set to neutral before setting the tension again, to avoid giving users visual cues based on the motion of the flexible surface. After pressing or sliding, users had to indicate if they could detect a difference or not, by pressing a button on the screen. During training and the main experiments, the user's hand holding the pen rested on the ledge provided by the support box. After the study, a questionnaire (see Questionnaire 2) recorded subjective ratings for the system, with a 5-point Likert scale (from 1, disagree to 5, agree).

Results

All participants could to a certain part well distinguish between the provided tensions within the press and slide tasks. Based on the collected staircase data, the probability of correctly identified tension differences between the two sequentially provided stimuli were determined. These probabilities were used to estimate both psychometric curves, one for pressing and the other for sliding (see Figure 4.18). The estimation was done using the python toolbox for psychometric function estimation from the University of Tübingen ¹.From the psychometric curve the point of subjective equality (force at probability(0.5)) and the JND ((force at probability(0.75)-force at probability(0.25))/2.0) can be derived, according to [JT13]. For our interface a JND threshold for pressing ($\Delta 0.18$ N) and for sliding ($\Delta 0.15$ N) could be defined. This results in a Weber fraction for pressing =21% and for sliding =18%. Both fractions are in line with the range for firmness (stiffness) experiments (15% to 22%) defined by Jones. According to our hypothesis (H1), the participants were able to distinguish better between tensions during sliding than during pressing. Therefore, the hypothesis can be accepted because the interaction metaphor has an influence on the perceptual threshold.

¹https://github.com/wichmann-lab/psignifit/wiki (last accessed 19th April 2021)



Figure 4.18: Psychometric curves (forces in N). Pen press/slide depict results of the pilot study.

The analysis of the questionnaire indicated that the interface was easy to use and to learn for sliding (M = 4.25/SD = 0.45; resp. M = 4.42/SD = 0.51) and for pressing (M = 4.58/SD = 0.51; resp. M = 4.5/SD = 0.67). Participants were neutral about the white noise sound they heard during the experiments (M = 3.10/SD = 1.50), however, they reported that the servo motors could not be heard and therefore the goal of applying white noise sound was successfully achieved. Moreover, participants did not feel frustrated by the task (press M = 3.58/SD = 1.31; slide M = 3.67/SD = 1.07), felt confident operating the system (press M = 4.08/SD = 0.79; slide M = 4.0/SD = 0.85), and could sense cues well with the pen (press M = 3.67/SD = 1.07; slide M = 3.58/SD = 0.9). In general, ratings were quite similar for both experimental conditions and ranged from neutral to positive.

Due to the flexible surface we offer a continuously variable, compliant feedback, which is force-wise in a continuum between the soft mid-air feedback, such as that provided by ultrasonic sound waves [CSL⁺13], and the more firm, solenoid-based or vibration feedback provided by haptic pens, such as [CST15]. The feedback range can best be understood when looking at the different sensations felt when touching human skin in different places on the hand (often referred to as the "doneness test" when grilling steaks) - it shows both the softness and the ability to regain the original shape. Although further research is needed, we assume our feedback that is variably compliant and soft can potentially be associated with different types of materials, ranging from soft fabrics to rubber or silicone.

4.3.3 Application scenarios

Alongside the analysis of perceptual aspects that can be helpful for interaction with GUI elements, we also see great potential for possible applications. For example, the elasticity of the material used is suitable for simulating haptic properties, for example, tools. Therefore, we will discuss how the pen can be actuated, and how these kinds of actuation - including flexible and rigid state transitions - can be used within a sample application, an art application. Afterwards, we discuss the higher-level functionality that can be derived.

The presented interface can generate a variety of different feedback, including force feedback, tactile and auditory feedback. Figure 4.14 (A-E) looks closely at how the pen can be operated. If there is a minimum tension, different degrees of hardness can be simulated (A), which the user can feel when the pen is pressed down. It is also possible to generate an upward force (B), by raising the tension quickly the pen will move away from the tablet display. By adding the vibration motor and the speaker touching the flexible material, vibrations, sound and combined feedback could be provided by the interface (C, D). In cases where the tablet-pen interaction should be performed as usual, the flexible surface can be relaxed to a degree that the pen touches the screen, resulting in a rigid feedback. This means that the interface can also handle the physical constraints that are common when interacting on a rigid screen: the movement is limited to a 2D surface (E). These cues are similar to cues provided in mobile devices (e.g. [KIP13], [LOM⁺11], [VDH14]), but are here directly coupled to the feedback provided over the flexible haptic layer. Note that the membrane vibration can also be felt (though slightly dampened) while the pen touches the rigid screen.



Figure 4.19: Art application: rigid and variable compliant (flexible) states simulate different tools. The paint-by-numbers application displays upward force to warn users when passing over a boundary of an area.

Art application

To demonstrate the potential of the interface an art application was designed and implemented (see Figure 4.19). With this application the versatile possibilities of the system can be illustrated clearly. It enables the user to draw and paint with different media (oil and watercolour painting with different brush types) by adjusting the "feel" of the different tools (e.g. brush type versus pencil) and surface properties (e.g. paper roughness) - the application actively uses transitions between flexible and rigid states. Moreover, we studied novel widgets by introducing, for instance, a haptic slider. In the application the slider allows the user to adjust the settings for tool like brushes and surface properties like paper roughness. A benefit of the haptic slider is to feel the effect immediately and directly. By changing the position of the slider we assign the new value to the tool property to be manipulated. Thus, the properties of the tool can be adjusted and simultaneously perceived through the feedback provided.

The novel haptic layer plays a central role in the application by switching between the "full flexible mode" (which represents watercolours and oil brushes) and the "rigid mode" in which the flexible surface touches the display. In contrast to the flexible mode, the latter provides a pen-on-paper feeling. All tools actively use the pressure sensitive tip to change the line/stroke thickness. Through the speaker and vibration motor, which are attached to the flexible surface, we can simulate the roughness of the paper structure during a painting process. Finally, we demonstrate an active form of guidance using a painting-by-numbers example. This painting method, often used by children, has coloured areas with strict boundaries where the paint only needs to be applied within the borders. As soon as a user moves over a defined boundary with his/her pen, a dedicated feedback is provided.

In particular, the art application illustrates how we can map pen actuation to functionality. In the following a more abstract view and discussion of the system will be provided. Based on the material properties of used flexible surface the haptic exploration is bound to the simulation of soft deformable materials, similar to [MKG⁺14], but more controllable. Mueller et al. focus on push and pull gestures on a large elastic display. Their interface uses the similar elastic properties as ours to provide haptic feedback. But compared to the one presented here, Mueller et al.'s can be actively actuated. Moreover, the coupling of both (a) the speaker and vibration motor and (b) the rigid surface of the touchscreen allows the generation of cues that map well to application requirements. Cues can be associated with standard widget elements, such as buttons and sliders, as introduced in previous paragraphs. In our current implementation, the system provides different kinds of haptic feedback for widgets. Selection of widget elements is supported in the flexible haptic layer by measuring pen-tip pressure, or by pressing the pen down to touch the rigid surface.

Finally, audio and vibration can be used to provide selection cues, similar to [WRS17]. Tactile or flexible cues can augment the interaction with tools that exhibit elastic properties. As we showed in our application, such cues can enhance the user experience and potentially also performance with such tools. Such cues provide indications of the real-world haptic properties that help the user understand how the tool performs while using it. Besides improving performance or user experience using haptic cues, the richness of haptic feedback – such as brush properties – stimulates artists, and could therefore support creative processes by bringing more realism into the interaction [BSLM01]. At least, the system can be used for guidance. Simulated rims [KIP13] or textures, but also other sudden changes of feedback, such as an abrupt upward force, can be used to display a warning or to guide hand motion. As an example, the pen can be forced upward a bit once the user starts to press a delete-all button, to indirectly warn the user about the implications.

4.3.4 Discussion

The second interface emphasizes the potential of haptic feedback with a unique approach designed for touchscreen interaction and thus contributes to answering RQ1. Additionally, the proposed interfaces differs essentially from previous haptic pens. Such pens often require physical contact with a rigid screen in order to explore physical aspects. Further pen interfaces provide basic force feedback while operating in the air. It is a challenge to transfer our combined haptic feedback into a pen with a small form factor: most pens only support a subset of our functionality and are often tethered. This means compared to pen interfaces discussed in Chapter 2 is that our pen do not require any additional components. The feedback is transmitted using the elastic material. This makes the pen light, requires no additional power and the fact that the feedback is outsourced allows increased force to be applied to the pen. All this enables, a haptic feedback [CSL⁺13], and the firmer haptic feedback in rigid screen interaction, e.g. through the haptic pens mentioned above.

Although this interface offers a broad spectrum of haptic feedback, it only support a subset of all potential haptic features. For instance, weight, temperature and volume are not supported in the current system. By allowing components to be integrated into the pen, it is easy to integrate temperature feedback into the pen. It can be realized using a Peltier element. However, if the pen is to remain without additional components, warm air can be generated externally using a coil and provided to the user through a small fan, similar to [WLT⁺20]. In contrast to temperature, weight could be difficult to simulate without tethering. Volumetric feedback can currently only be rendered in a very small range. With the system presented in this section, larger intervals could, of course, be mapped to the defined range, but this in turn means that the spatial resolution of the feedback becomes decreased. In addition, with the introduced interface only a limited range of variable compliant stiffness can be simulated (about 0.7N to 1.6N). With the present technical design it would be difficult to simulate much stiffer materials, as the servos will not be able to actuate the used material. Also, the current surface material blocks direct finger interaction. Yet, as long as the material is transparent, other flexible surface materials could simulate different ranges of softer flexible materials. Perceiving textures through the flexible surface material likely has some limitations as the material absorbs vibration differently under different tensions (especially when it touches the rigid display), which warrants further study. Finally, while the sound of the servos could not be heard during experiments due to the headphones, it might have to be blocked by other means for situations where headphones are not an option.

At least, pen latency should be discussed. In the current implementation pen latency led to some limitations. Although the latency is similar to that of other conventional pens (e.g. Wacom Cintiq), it is higher than that of the latest systems (Apple Pencil 50 ms or MS surface, where there is almost none). Although the haptic feedback in the interface was decoupled from the visual feedback, still, the latency was hardly noticed, especially in our art application, as painting movements were rather deliberate and slow. Furthermore, there is only low servo latency for continuous static feedback, such as while using a brush. This may be similar to other haptic exploration tasks, such as pressure sensing and especially contour exploration, which are typically performed more slowly [LK09]. In a future iteration of the system, we plan to use a lower-latency tablet.

4.3.5 Conclusion

In order to obtain further details of the research question RQ1 a multilayer feedback approach for pen-based tablet interaction was introduced. To our knowledge, the interface introduced in this section is the first system that demonstrate a combined elastic, tactile and rigid feedback in a single interface. Within pen-based technologies, haptic feedback, especially force feedback, tends to be difficult to achieve without connecting the pen to a force mechanism, a issue that we overcome with FleXurface. Through an art application, we showed how the different feedback possibilities afforded by the system can enhance interaction through its multilayer approach, illustrating the potential of the system.

Taking into account our Aim 1 and Objective 2 (see Section 1.2), our sub-question (How can elastic materials be combined with touchscreen interaction?) and the resulting perception aspects and limitations, the following sub-contributions could be achieved with the interface:

- A novel haptic pen-based touchscreen interaction approach that allows the exploration of various haptic features, for instance firmness, textures, material characteristics and interaction properties was proposed. Multilayer haptic interaction approach uses two interaction layers a flexible haptic and a rigid screen layer and the user can seamlessly transition between flexible and rigid states with the untethered pen.
- Perception-based aspects of the flexible haptic layer have been studied to show the limitations of the interface. Therefore, we **estimate perceptual properties** of the flexible layer through a user studies. Therefore, we performed a just noticeable difference (JND) experiment to assess how well users can perceive tension (firmness) differences.

Previous work has also shown that tactile cues may improve performance with widgets [BCB07], [FS01], [LMBS07]. As we do not have to substitute haptics (kinesthetic cues) through tactile feedback, we assume that our multilayer approach can further enhance performance for such tasks and support higher accuracy. Though previous work shows initial positive indications $[FLO^+12]$, this latter aspect deserves further study, to show to what extent variable compliant haptic feedback would be truly better than vibration-only feedback. Furthermore, as we did not alter the visual appearance of buttons when they were pressed, visual feedback did not modulate firmness perception. Previous work $[KTC^+12]$ indicated that haptic firmness discrimination performance could be improved when visual feedback is provided along with haptics. While haptic cues work within the non-visual sensory channel, they also afford eyes-off interaction, another area that deserves follow up. In addition, it will be interesting to assess to what extend the system can improve expressiveness [BSLM01] – while initial feedback with our art application was positive, we are confident our system can deliver to this extent, which we aim to address in a more elaborate study.

Notwithstanding that the current interface has mainly been used to study the design space of the multilayer approach, miniaturization is an interesting option to explore similar feedback mechanisms within the mobile domain. One potential solution is to make use of deflection rollers to wrap the material more effectively around the display with smaller servos or to use linear actuators which are also quieter. Finally, while the pen interaction has shown good performance, finger input is a further interesting direction to explore. To support direct finger interaction, other flexible materials, such as those that include, for instance graphite particles, could be used.

4.4 Pin-based Feedback on the Back-of-the-Device

In the previous sections of this chapter, it was shown how force feedback can enhance touchscreen interaction. Therefore, two individual haptic user interfaces have been introduced. In this section we will explore our third prototype to investigate further details of RQ1, for which purpose HapticPhone was developed. In this section, we will explore how a smartphone can be enriched with haptic feedback and what impact pin-based haptic feedback has on touch accuracy. Our approach uses the back of the smartphone for interaction, similar to fingerprint sensors. In order to achieve our first aim (Aim 1) we have realized our objective 2 and thus research on the third sub-question in this context.

Since the appearance of first prototypes about a decade ago, researchers exploring the possibilities of back-of-device (BoD) interaction. Motivated by efforts to minimize screen occlusion through fingers, early prototypes like LucidTouch [WFB+07] made use of pseudo-transparent displays. At the back, users could interact with the full screen content using touch input without occluding content. One hope was that back interaction could compensate for poor pointing performance, especially for the selection of smaller objects.

Instead of touch input on the back, we explore the potential of pin-based force feedback at the back to enhance thumb-based touch interaction on the front-of-the-device (FoD) screen. Adding force feedback to smartphones could have great potential, as many force events currently are substituted through tactile (vibration) or visual-only feedback mechanisms, which affects the perception of those events [KWBT91]. Consider pressing a button: in real life, we receive physical (force) and tactile (surface) feedback while pressing the button down. With current smartphones, audio-tactile feedback ("click") and a change in button colour indicates when a button is pressed, which is not compliant with real-world interaction. Based on our previous work, we assume that force feedback could affect performance positively, in particular for thumb-based interaction that often suffers from occlusions. Moreover, as we will investigate in our studies, force feedback might improve performance in tasks that, in the real world, do not depend on such feedback.

Studies have shown that tactile feedback can increase performance in target selection tasks [AS94], [CCKB15]. However, the exploration of force feedback in smartphones is rare. A major cause is that adding force feedback actuators to the front display has many physical form factor constraints, which makes physical construction challenging. For example, mounting an actuator on the front would occlude part of the display, and likely hinder input. In contrast, the main research problem our approach faces is how well force feedback at the BoD works.



Figure 4.20: Illustration of the prototype, a touchscreen for thumb interaction is shown on top and the force feedback pins in landscape orientation are depict below.

4.4.1 Implementation

We designed and developed a BoD approach (see Figure 4.20), a device for exploring the haptic design space at the back of smartphones. We use BoD feedback to overcome physical form-factor challenges for adding force feedback to the front of the device and explore this novel design space by looking into the human and technical aspects of our interface.

As showed in Section 4.2 and Section 4.3 and also discussed in the literature review - force feedback could improved interaction [JKT⁺16]. Therefore, our main design goal was to combine BoD interaction and haptic feedback with thumb-based interaction. Continuous, assisting force feedback can provide a secondary cue for improved touch performance and increase the user experience in terms of subjective performance. To do so, we have attached a force feedback mechanisms to the back of the device that relays feedback to touch-input on the screen at the FoD. As such, our design couples front touch-based screen interaction and BoD feedback.

Technical Design Rationale

To overcome physical form-factor constraints - such as screen occlusion, input hindrance, and an unergonomic grip - at the front of smartphones, we explore the feasibility of adding force feedback to the back of a smartphone



Figure 4.21: Photos from different iteration stages all pursued the same goal of providing haptic feedback on the back of a smartphone.

We designed the new system for both one-handed (portrait) and two-handed (landscape) interaction. Input is performed by either the left or the right thumb or both on the FoD, while haptic feedback is relayed to the index fingers at the BoD. After several hardware design iterations (see Figure 4.21), that looked into ease-of-use, grip stability, and reachability of the feedback elements, and informed by the results of Le et al. [LMBH18] on appropriate finger locations for one-handed BoD interaction, we chose a three pin layout. The arrangement and dimensions are shown in Figure 4.22. This layout allows users to use the outer pins comfortably while interacting in landscape mode, just as the middle or top pin can be used for one-handed portrait interaction.

The remaining fingers grasp the ergonomically shaped bottom part of the device, to stabilize and balance the device during interaction and feedback. To improve the grip on the device, we attached a rubber based grip tape around the edge of the unit. The grip has also been designed to afford comfortable device usage in both portrait and landscape mode. However, one-handed usage may suffer from limitations, as previous work has shown that one-handed interaction can lead to more fatigue, less precision, and a less secure grip of the device [BLC⁺12].

Force feedback is provided by three actuated pins at the back of the device (see Figure 4.22), at the locations where the left or the right index finger naturally rest. Three dimples in the housing help to keep the fingers at the ideal locations, where force feedback is provided.

Hardware

Our interface comprises three high voltage (7.4V) micro servos (BMS-22HV), each measuring 23.0 x 12.0 x 25.4mm, running with a speed of 50ms per 60° rotation at no load. Each servo has a pulse width of 1200μ m and a maximum torque of 0.245N/m. We estimate that the servos have a resolution of 0.2° as follows: a pulse width of 1200μ m for 120° results



Figure 4.22: Technical drawing of the top and side view of our interface to show dimensions and the chosen pin layout.

in 1200 steps for 120°. With a dead band width of 2μ s this results in 600 controllable steps for 120°. All in all, the servos can be controlled with a resolution of 0.2°, which corresponds to a radian measure of 0.049mm with an arm length of 14mm(see Figure 4.23 left).



Figure 4.23: Modified scotch-yoke mechanism to translate rotational into linear movements. An exploded diagram to introduce the used components (right) and a front view of a fully retracted and extended pin (left).

For each of the three pins a modified scotch-yoke mechanism was developed to convert the servos' rotational movement into a linear movement (see Figure 4.23). Through two guiding rails on both sides of the base of a pin (highlighted in grey and red) and through an additional sliding mechanism (highlighted in blue and red), a linear movement can be created. For smaller pin movements, e.g., amplitude 1mm, our approach achieves a frequency of 50Hz. In idle mode, when no counter-force is applied, the power consumption



Figure 4.24: System latency measurements (in milliseconds) for force, tactile and display, shown as a bar plot. The actuators were operated either with 6V or with 7.4V and latency was measured with 0mm or 1mm pin height.

of the prototype is about 10mA. With a strong counter-force the consumption is about 500mA.

The design enables a feedback range along a single axis from 0 up to 5mm. All hardware components are mounted in a self-contained unit at the back of an Android 7 Huawei P9 Plus mobile, which also supports pressure-based input. All other parts, such as the case or the scotch-yoke mechanism, are 3D printed, which ensures easy reproducibility. The entire prototype measures 157.6 x 80.6 x 28.5mm, weighs about 200g and has a resolution of 1920x1080. A 16-bit Adafruit Feather (M0 Basic Proto) micro-controller, directly connected via USB (OTG) to the smartphone, handles the communication between feedback elements and the application, which means that touch input is forwarded with low latency to the pins.

The current prototype relies on a cabled solution. However a completely self-maintained unit can be created using an Arduino Nano and communication over Bluetooth. Servo motors could be powered with a buck-boost converter, which can increase or decrease the input voltage and which can permit the use of small battery packs.

Figure 4.24 lists the latency of the systems hardware components in milliseconds. In an experimental setup, latency of the touchscreen (visual), the vibration (tactile), and the proposed feedback (force) were measured. Force feedback was measured under multiple conditions, with 6V vs 7.4V and pin heights of 0mm vs 1mm to show the influence of the described parameters on latency. A two pole relay was used to trigger a touchscreen using a micro-controller, similar to Deber et al. [DAJ⁺16]. After triggering the touchscreen a photodiode (Tru Components 5013M1C) was used to measure the time between touchscreen event and visual feedback. To obtain tactile feedback latency, the photodiode was replaced by a vibration sensor (Phidgets 1104). To obtain force feedback latency, an electric circuit was mechanically closed. This mechanism allows to measure the elapsed time of

the touchscreen event and its associated feedback, known as latency. For each feedback modality, 100 measurements were performed automatically. Figure 4.24 shows the average measured latency and the standard deviation of the measurement data. Opinions differ on the influence of latency on performance [JH05], [KBL14]. The modality latencies are slightly above those recommended by Kaaresoja et al. discussed in the literature review (see Section 2.2.1). Nonetheless, we believe that a performance comparison of the three components is valid for an exploratory study.

Implementation

In our implementation, we support two different types of touch input, namely positionand force-based interaction. The latter makes use of the pressure-sensitive touchscreen of the smartphone.

In contrast to conventional tactile feedback through vibration on the screen, the introduced force feedback can support several clearly distinguishable levels of intensities, as our psychophysical study will demonstrate. These intensities can be rendered in a continuous and discrete way. For the proposed haptic feedback two main software components are important. Firstly, the implementation on the smartphone, which takes care of the orientation of the smartphone and offers developers the possibility to address the three pins individually for their application. The smartphone communicates with a micro-controller via I²C. The second component, the software on the micro-controller, receives data from the smartphone and passes it directly to the motors.

To support people with different thumb lengths, our implementation enables customization, where the front interaction areas shown on the screen can be scaled to match the thumb reach. With this, users are able to define their interaction radius according to their thumb size. This approach ensures that all graphical interface elements are reachable on the screen. In landscape view the screen is thus separated into a right and a left interaction area, meaning that the left and right thumb have their "own" interaction areas.

4.4.2 Experiment 1 - Psychophysical Perception Study

We conducted this study to determine the relationship between stimuli and sensation at the index finger using HapticPhone. The study examines haptic perception aspects and constraints of the BoD system. We performed a just noticeable difference (JND) experiment to address which force signals can users easily distinguish. This study is an important step towards answering RQ1, as the accuracy of differentiation will later influence touch accuracy. In addition, we studied the relationship between thumb and index finger movements through a second task (spatial compliance). This investigates if it possible to transfer movements of the index finger to the thumb and how the users perceive such movements.

Participants

12 users (3 female, age M = 30.7/SD = 5.8) volunteered to participate in this study. All were right handed. Participants had various experiences with force feedback, ranging from no to regular experience. All participants used smartphones regularly. Each participant completed all 60 trials, which took on average 12 minutes. The index fingers of the subjects were 88.0mm (SD=10.8mm) and the thumbs were 62.0mm (SD=5.9mm) large on average.

Apparatus

Users operated the mobile system as described in the system design section, while being seated comfortably at a desk. Users could rest their arms on the desk, and were allowed to take small breaks between tasks. During the whole user study, users wore noise-cancelling headphones that played white noise to prevent users from hearing the servo actuation, as the servo sound could provide some information about the force feedback.

Tasks and Procedure

The psychophysical perception study consists of two tasks. The first task investigates the just noticeable difference of force stimuli afforded by the pins, while the second task looks at spatial compliance. Both tasks were performed in a counterbalanced order. Participants were asked to use their dominant hand for interacting with the system. This ensured that the dominate thumb was used for touch interaction and the index finger of the same hand for the relayed feedback. To do so users were asked to place their index finger on one of the outer two pins, while holding the device two-handed in landscape mode. In addition, users were advised not to apply force to the pin.

In contrast to FleXurface, where we performed the JND study using a staircase protocol, this time we used the principle of constant stimuli with a 2AFC, similar to Geschneider [Ges97]. This decision was made due to the fact that in our pin-based approach we wanted to predefined the values, as they were strongly limited to the system. This protocol specifies that n times a randomly chosen stimulus from an appropriate interval must

be compared to a standard stimulus. This can take place either spatially or temporally shifted.

In the experiment the standard stimulus S was set to 1mm while the comparison stimuli C_n were one of 1.2, 1.4, 1.6, 2.0 or 2.4mm. Stimuli were defined by the distance the pin was moved towards the index finger, i.e., away from the device. Each participant completed 10 repetitions. Each repetition was fully randomized, resulting in a total of 60 trials. For each trial, users were prompted to press a start button, to enable them to prepare themselves for the task. Each stimulus was presented for 2 seconds, with a pause of 1 second in between. The order of stimuli was chosen randomly, meaning the standard stimulus was presented either first or second. Users then had to determine which of the two stimuli was higher. Before the experimental tasks, users were allowed to do three practice trials. All training trials were marked visually so that the users knew when the actual study started.

The other task focused on the psychophysical perception of depth and height, which combines visual feedback and touch interaction (thumb) on the screen with BoD force feedback on the index finger. With this task we explored the spatial compliance between index finger and thumb, where we assume that the thumb can "perceive" height data from the index finger. This explores whether users perceive a valley at the thumb when the distance between thumb and index finger increases or will they perceive a hill and vice versa? This is important in order to use the feedback metaphor appropriately. For example, when creating the feeling that the thumb is sinking into the display, the correct feedback should be applied. In this task, participants were asked to explore a 3D Gaussian-like shape with the thumbs on the touchscreen. Since the screen has 2 physical dimensions, the 3rd dimension was mapped to the index finger, like a height map.

The maximum feedback (height/depth) was set to 2.4mm whenever the thumb reached the maximum of the shape displayed on the screen. While visual feedback remained constant over all trials, force feedback was designed so that either the distance between the index finger and the thumb became smaller or larger. Because the task was simple and easy to understand, only 8 trials were performed in random order per user. After each trial, users were asked if they felt that the 3D Gaussian-like shape was directed either into (valley) or out (hill) of the display.

depth perception					
participant	PSE [mm]	JND [mm]	WF [%]		
1	1.20	0.11	8.89		
2	1.28	0.07	5.88		
3	1.06	0.04	3.90		
4	1.23	0.13	10.80		
5	1.21	0.11	9.43		
6	1.21	0.12	9.62		
7	1.29	0.12	9.62		
8	1.23	0.10	7.82		
9	1.21	0.06	4.91		
10	1.15	0.17	14.74		
11	1.10	0.08	7.01		
12	1.14	0.04	3.83		
Mean	1.18	0.10	8.29		
SD	0.06	0.04	3.43		

Table 4.2: List of the results of the psychophysical perception experiment

Results

The JND experiment was analysed using the psignifit MATLAB toolbox², similar to $[JKT^+16]$. Three relevant values were determined: point of subjective equality (PSE) and stimulus values at 25% and 75% probability. With these three values the JND and the Weber Fraction (WF) were determined. The results of the psychophysical perception study are summarized in Table 4.2. The (index finger) depth perception threshold of our system is on average 0.1mm. All psychophysic curves of this study can be found in Figure B.1. Our results are in line with what Jang et al. [JKT⁺16] reported, who found an average JND of 0.15mm.

The evaluation of the answers shows that in the second part of the study 8% of all subjects felt a valley whenever the distance between index finger and thumb decreased and 91% felt a hill whenever the distance increased. Since users were asked what they perceived with their thumbs on the touchscreen this indicates that users seem to experience the sensation that their thumbs sink into the smartphone when the distance between index finger and thumb increases and vice versa.

²https://github.com/wichmann-lab/psignifit/wiki (last accessed 19th April 2021)

4.4.3 Experiment 2 - Touch Accuracy Study

This study mainly examined touch accuracy of the BoD feedback mechanisms, comparing force with tactile and visual-only ("without") feedback. An overview of the conducted study is provided in Figure 4.25. Since thumb interaction offers a relatively smaller interaction area due to physical constraints for reaching compared to using the index finger, a higher accuracy might offer the possibility of comfortably controlling denser interaction elements on a smaller area. To determine touch accuracy we used two tasks, a drag-n-drop and a selection task. The drag-n-drop task is designed to determine the accuracy of a constant force feedback stimulus and to compare this with other feedback metaphors. The selection task pursues the same goal, however, the intensity of the force feedback was varied here. In a questionnaire we additionally explored user experience in terms of subjective performance of completion time, precision and quality of the feedback. Since the proposed system employed thumb-based interaction, multi-touch metaphors were excluded from the studies. For both parts, the drag and drop and the selection task, it was hypothesized:

H1: The proposed feedback mechanism will increase touch accuracy.

H2: Haptic feedback will increase the subjective performance of the participants.

Participants

12 users (4 female, age M = 32.0/SD = 5.4) volunteered to participate in the laboratory experiment. Again, all participants were right handed. Participants had varied experience with force feedback, ranging from none to regular experience. Each participant completed all of the 108 trials, which took an average of 20 minutes. As in the previous studies, participation was voluntary and the subjects could stop at any time without any effect.

Apparatus

The apparatus for landscape interaction was similar to the one used in the first study reported above. In addition, we investigated interaction in portrait mode in this study, where the thumb of the dominant hand was used to handle the touch input and the corresponding index finger receives the force feedback of the middle pin. Users were allowed to use the other hand to keep the smartphone stable.


Figure 4.25: A schematic overview of the touch accuracy study. The diagram illustrates the apparatus, procedure and design of the two task.



Figure 4.26: Exemplary representation of the target sizes in the drag-and-drop study. Users had to drag the orange disk onto the grey one.

Design and Procedure

Participants had to complete two tasks, a drag-and-drop and a selection task. Since the proposed BoD approach is comparable with more common feedback metaphors, users were asked to use all three different feedback metaphors (force, tactile, visual-only) to gain an understanding of each feedback modality and their differences.

Can interaction of the thumb be influenced by controlled movements induced on the index finger? To investigate this question, this study focuses on the interplay of relayed feedback from the index finger at the BoD to the thumb with the aim of increasing accuracy of thumb interaction on the front of the device. In addition, since the offset between fingers could also affect touch accuracy, we also investigated the influence of the offset in the axis orthogonal to the touchscreen between the thumb position on the touchscreen and index finger on the back. All tasks were performed in landscape and portrait mode, as both modes are common in every-day smartphone interaction.

After the study, users completed a questionnaire (see Questionnaire 3), in which we queried them about the three parameters accuracy, completion time and quality for each feedback modality using a 7-Point Likert scale, ranging from "strongly disagree" to "strongly agree", through statements such as: I could perform the task precisely, I could complete the task quickly, and The quality of the feedback was excellent. Finally, the participants had to fill out the SUS questionnaire [Bro96] to get feedback about the usability of the introduced interface.

Drag-and-Drop Task

In this study, participants were asked to perform a drag-and-drop task by moving a disc as accurately as possible onto another (see Figure 4.26). We used a within-subject design to examine touch accuracy and user experience, through a (3x3x2) factorial design with 3 independent variables: screen orientation (levels: landscape and portrait), target disc size (levels: 1.0, 1.5, 2.0mm) and feedback type (levels: force, tactile and visual-only). The factors disc size and feedback type were fully randomized over the trials, whereas the factor screen orientation was counterbalanced. The dependent variables were touch accuracy and user experience. Touch accuracy was measured with an error rate in mm, representing the distance to the centre of the target disc. The user experience was measured through the questionnaire described above.

To examine touch accuracy, users had to drag one disc (the action disc) and drop it onto another (target) disc. As we specifically wanted to investigate the accuracy of the event, the participants were asked to do this as accurately as possible. Throughout the entire study the size of the action disk (diameter=4mm) was the same. The size of the target disc varied (diameter=1.0, 1.5, 2.0mm), with the largest target size chosen so that users still had a chance to see or estimate where the disc is located, without completely covering the target. In contrast, the smallest target was completely covered by the thumb when both centres intersect. Both discs appeared at random, but reachable, locations. Once participants assumed that the centre of the action disc was above the centre target disc, participants had to submit their position result by pressing a button with the other thumb.

We designed the visual feedback to be similar across different feedback types, while the perceived stimuli at the index finger/thumb were different. To initially select the action disc, users had to select it for at least 400ms. Thereafter, participants were able to drag the disc with their thumb. In the 'visual-only' mode there was only visual feedback. In the other conditions, tactile respectively force feedback was enabled whenever the centre of the action disc was inside the target disc, which means that either standard vibration was turned on or a 2mm force feedback event was rendered on the BoD.

Selection Task

In the selection task a within-subject design was used to examine touch accuracy and user experience. We used a 3x3x2 factorial design, with 3 independent variables: screen orientation (levels: landscape and portrait), cell size (levels: 3, 5, 7mm) and feedback type (levels: force, tactile and visual-only). Each user thus completed 36 trials. The factors cell size and feedback type were fully randomized over the trials, whereas the factor screen orientation was counterbalanced, so that participants completed all trials for the same screen orientation in one block. This avoids users having to constantly rotate the screen, which could lead to unnecessary fatigue and increased stress. The dependent variables

1	30	20	4	27	26
11	15	29	28	25	35
9	6	33	17	7	22
32	21	3	18	14	19
8	12	24	31	0	5
23	13	10	34	2	16

1	30	20	4	27	26				
11	15	29	28	25	35		1	30	20
9	6	33	17	7	22		11	15	29
							9	6	33
32	21	3	18	14	19		32	21	3
8	12	24	31	0	5		8	12	24
				_			23	13	10
23	13	10	34	2	16	•			

Figure 4.27: Exemplary representation of the different target matrix sizes in the selection task. Users had to select a specific cell in the matrix. Cells have alternating intensities (tactile) or heights (force).

were touch accuracy and user experience. Touch accuracy was measured with an error measurement in mm, corresponding to the distance of the touch centre to the centre of the selected cell. User experience was assessed with the questionnaire described above.

We used a matrix of abstract targets, as a representation similar to common interaction targets, such as menus, toolbars, tool palettes or links. Users were asked to search and select an element in a 6x6 matrix, in landscape as well as portrait mode (see Figure 4.27). The centre of the matrix was located exactly above the feedback pin. For each cell of the matrix a unique number from 0 to 35 was randomly generated. The number to be selected was visually marked. Users could then use the thumb of their dominant hand to select the cell as best they could, as close as possible to the centre of the cell. To submit the target position participants had to press a button with the other thumb. We chose three cell sizes of 3, 5 and 7mm for this study, based on earlier research on appropriate target sizes for thumb interaction [PKB06], [HKAI16].

Depending on the feedback type, we implemented different approaches to search for and select a cell. For visual-only feedback, as the name suggests, no additional cues were rendered. For tactile feedback, we provided vibration via the in-built functionality of the touchscreen device. This means that the feedback was rendered to the whole device, including the front screen. Since Android does not natively offer the possibility to change the intensity of the vibration, we used a binary approach, i.e., only on or off. Thus, vibration feedback was either switched on and off, alternating with the cells, with yellow cells receiving vibration feedback (see Figure 4.27). This enables the user to sense the transitions between cells. For force feedback we also alternated feedback with the cell

matrix cell, but increased the intensity of the feedback towards the centre of the (whole) matrix, up to a maximum of 2.4mm in discrete steps, depending on the size of the matrix. For example, the step size was 0.8mm for the 6x6 matrix. We chose to increase the feedback towards the middle of the matrix so that users can not only sense the transitions from cell to cell (local position), but also determine the (global) position in the matrix based on the intensity.

Before each experiment, users were asked to explore each of the feedback modalities for about a minute. Completion time was measured per trial and logged for both tasks.

Results

Drag-and-Drop As the data were not normally distributed, we used a Friedman-test to analyse the data of the drag-and-drop task. The completion time in milliseconds per trial and feedback are listed in Table 4.3. The time difference between the conditions was not significant.

Table 4.3: Mean time and standard deviation per trial for the drag-and-drop task in seconds (standard deviation in brackets).

	portrait	landscape
visual-only tactile feedback force feedback	5.32 (4.42) 6.66 (5.78) $6.12 (5.03)$	5.12 (3.95) 6.01 (5.20) 5.69 (3.61)

Through the error distance measure we identified that that users performed the task significantly more precisely with tactile and force feedback for both landscape ($\chi^2 = 11.03$, df = 2, p = 0.004) and portrait mode ($\chi^2 = 10.53, df = 2, p = 0.005$), see Figure 4.28. A pairwise comparison identifies a significant effect between tactile and visual-only (landscape p = 0.099/portrait p = 0.042) and force and visual-only (landscape p = 0.013/portrait p = 0.006). There was no significance between tactile and force feedback, according to the Nemenyi *post-hoc* test. Regarding the first hypothesis (H1), it can be concluded that haptic feedback on the back can improve touch accuracy compared to no feedback ("visual-only"). There was no difference between tactile and pin-based BoD feedback in this study.

The results for the questionnaire data (see Figure 4.29) identify that force and tactile feedback were perceived to be more precise. A pairwise Wilcoxon-test identified a significant effect for the accuracy ratings. Participants felt more accurate with force (M = 6.0/SD = 0.9; W = 8, p < 0.001) and tactile feedback (M = 5.3/SD = 0.8, p < 0.001)



Figure 4.28: Box-plot for touch accuracy in the drag-and-drop study. For smaller target sizes force and tactile feedback have a significantly lower error than visual-only.

Figure 4.29: Subjective drag-and-drop performance in terms of precision, completion time and feedback quality.

W = 17.5, p = 0.001) compared with visual-only feedback (M = 3.5/SD = 1.3). The subjective scores of completion time reveals a similar trend. Thus users felt faster with force (M = 5.6/SD = 0.8; W = 17, p = 0.001) and tactile feedback (M = 5.3/SD = 0.8; W = 25, p = 0.006) compared with visual-only feedback (M = 3.9/SD = 1.2). Previous shows that hypothesis H2 is true for the drag and drop task. The assessment of self-performance increases with haptic feedback.

In conclusion, in both landscape and portrait mode tactile and force feedback were on average 1.3 times more accurate compared to visual-only feedback. This corresponds to a mean difference of about 1mm in touch accuracy between tactile and force feedback. The results also indicate a higher variance for the average accuracy without feedback compared to other forms of feedback. This can be interpreted as that the users felt more confident in their selection with tactile and force feedback than without feedback. Additionally, any potential offset between touch and pin position seems to have had no observable influence on accuracy and time per trial in the study.

Selection We were primarily interested in touch accuracy for the selection task. As the proposed BoD force feedback condition involved several intensity levels, this meant that participants were able to determine not only their local but also the global position in the matrix. Therefore, we assumed that the touch accuracy would be higher with force feedback compared to the visual-only or tactile conditions. As the data were not normally distributed, we applied a Friedman-Test instead of an ANOVA. We conducted a Nemenyi-Test to calculate pairwise comparisons between feedback groups in the post-hoc analysis. For sake of completeness, the means and standard deviations of the completion times per trial are listed in Table 4.4.

Table 4.4: Mean time and standard deviation per trial in seconds for the selection task (standard deviation in brackets).

	portrait	landscape
visual-only tactile feedback	$\begin{array}{c} 3.28 \ (4.16) \\ 3.23 \ (1.72) \\ 2.66 \ (2.41) \end{array}$	3.24 (1.96) 2.64 (1.40) 2.20 (2.67)
force feedback	3.66(3.41)	3.30(3.67)

We found a significant effect of touch accuracy in landscape mode ($\chi^2 = 7.75$, df = 2, p = 0.021). A pairwise comparison with the Nemenyi post-hoc test between visual-only, tactile and force feedback showed a significant difference between force and no feedback (p = 0.016), but no significance between tactile and no feedback. The analysis of the selection accuracy in portrait mode indicates also a significant effect ($\chi^2 = 14.53$, df = 2, p < 0.001). The post-hoc test for portrait mode reveals a significant difference of touch accuracy for force and visual-only feedback (p < 0.001) and tactile and visual-only feedback (p = 0.010). This demonstrates that BoD force feedback can increase touch accuracy in both conditions. Similar to the first task, the proposed feedback approach improves touch accuracy compared to no feedback. However, unlike in the first task, a higher touch accuracy was also found when compared to tactile feedback. Thus, hypothesis H1 is true.

This means users were able to select smaller targets with higher accuracy with force feedback than without. Users were able to select all cell sizes with a probability of 66.6% (force feedback), 63.9% (tactile feedback) and 59.7% (without feedback) in landscape mode. Considering only the smallest cell size (3mm) in landscape mode, users could select the cells correctly with a probability of 45.8% (force feedback), 37.5% (tactile) and 12.5% (without feedback).

In portrait mode the results look similar. Participants were able to select cells of all sizes with a probability of 68.1% (force feedback), 70.8% (tactile feedback) and 59.7% (without feedback). Considering only the smallest cell size, participants were able to select the cells with a probability of 37.5% (force feedback), 41.6% (tactile feedback) and 20.0% (without feedback). Figure 4.30 depicts the results for all cell sizes.

The questionnaire data was evaluated using a pairwise Wilcoxon-test, which indicates that both accuracy and time per trial were significant (see Figure 4.31). Participants felt more accurate with force (M = 5.9/SD = 0.8; W = 6, p < 0.001) and tactile feedback (M = 5.3/SD = 0.8; W = 15.5, p < 0.001) compared with visual-only feedback (M = 3.3/SD = 1.3). Additionally, users felt also faster with force (M = 5.7/SD = 0.9;



Figure 4.30: Percentage of successfully selected cells per size and mode. The influence on accuracy without feedback decreases dramatically as soon as the thumb covers the target.

Figure 4.31: Subjective selection performance in terms of precision, completion time and feedback quality. For all three parameters participants liked force feedback most.

W = 23.5, p = 0.004) and tactile (M = 5.4/SD = 0.8; W = 28, p = 0.009) feedback than with visual-only feedback (M = 3.9/SD = 1.5). Haptic feedback also leads to a higher subjective performance in the selection task, which means that hypothesis H2 can also be accepted for the second part of the study.

The improved selection accuracy of smaller targets is an important finding, extending previous research in BoD interaction, such as [WFB⁺07], showing that haptic feedback is also a valuable method to overcome occlusion problems. When looking at the smallest cell size of 3mm, force and tactile feedback can on average increase accuracy by a factor of 2.5 compared to visual-only feedback. This means that the presented feedback mechanism can improve selection tasks, including text selection, considerably.

Participants filled out the SUS questionnaire to evaluate system usability. They rated our new system on average at 81 with a standard deviation of 13. According to Bangor et al. [BKM09] this means that the average is between "good" and "excellent" usability.

4.4.4 Application scenarios

To show how HapticPhone supports higher-level tasks, we implemented three applications that extend the basic (low-level) functionality demonstrated in experiments (see Figure 4.32). These applications demonstrate other instances where force feedback at the BoD is also useful.



Figure 4.32: Exemplary application scenarios for the proposed BoD force feedback approach: (top), a volume viewer, a racing game, a map viewer, all supported by force feedback on the back of the smartphone. The different pin lengths are an example of how data can be mapped to the pins.

Pressure input. We implemented a *3D Volume Viewer* application based on stacked images that can be explored through pressure input. In the first iteration, the applied pressure was directly transferred to the intensity of the servo motor. In this case force feedback was used to enhance scene navigation, as force feedback indicates where the user is in the scene: feedback is spatially compliant, as it is based on displacement with more/less pressure. Specifically, the user can directly feel if the currently viewed slice is further up or down in the 3D volume stack, aligned with the displacement axis. Thus, we assume that force feedback can be useful to afford more precision for finding a given layer in a volume.

Height perception. We created a *Map Exploration* application that supports simultaneous exploration of multivariate, map-related data. With normal map viewers, exploring multivariate data in parallel can become difficult. Thus, complex visualization methods are normally needed for such scenarios. In our application, we pass different geographic data to different feedback elements, e.g., output elevation profiles via a servo and map environmental pollution data to tactile stimuli. This choice effectively implements multichannel feedback to provide feedback about different aspects of the data. The application builds on the JND results, which showed that users can interpret fine differences through force feedback. We assume this accuracy can assist users in finding, e.g., a location on a map that is desirable in multiple dimensions.

Gaming experience. Finally, in a 2D Racing Game, we explore a combination of pressure input and height perception. Here, the car is steered through the touch position and input pressure is mapped to speed, while force feedback gives indication of the speed, by mapping the speed to the force pin. While force feedback has been shown to improve fun and immersion [LB13], we specifically targeted subjective control accuracy here.

4.4.5 Discussion

In our first study, we assessed if relayed feedback at the back of a device can support touch-based interaction on the front. Doing so, we explored different back-to-front feedback mappings to investigate both perception (psychophysical limits) and performance. In our second experiment, we showed that relayed feedback can indeed improve both objective and subjective performance. In some instances results are comparable to tactile feedback, while in others (selection accuracy of small targets) objective and subjective performance and preference are better, which demonstrates the potential of our approach.

Here, we discuss important factors that need to be considered when applying back-to-front relayed feedback in applications.

Design space In our device design and studies, we only explored a subset of the whole design space afforded by interaction on the back of a smartphone. A smartphone provides many different input and output modalities at different locations, which can be combined to create different mappings. e.g., a microphone, speaker, buttons, or notification LED are normally located at the front, while a fingerprint scanner, torchlight, or camera, are typically located at the back of the device. Consequently, the design space has many dimensions, as the same hardware mapping can be used to elicit different back-to-front feedback N:M mappings, especially for force feedback at the BoD.

Direct vs. indirect mapping Feedback can be directly or indirectly mapped. While we studied only directly mapped feedback, where a direct connection between the type of action and feedback exists, indirectly mapped back-to-front actions are also possible. Moreover, similar to tactile or auditory warnings, back-of-device feedback can communicate information about other processes than the one the user is involved in, for example to draw attention to a notification. Additionally, the presented prototype could also be used to enable in-pocket or other eyes-off interaction.

Spatial compliance We explored spatially compliant and non-compliant feedback. Our feedback mechanism provides displacement along a single axis, afforded by the mechanical constraints of the servo. When users press down on the screen, force feedback that moves the index finger away from the phone is spatially consistent, which matches the direction of the touch input action. Our analysis of the results indicates that users could interpret such spatially consistent feedback well, also because such consistency was preferred by

users. Users stated that spatially inconsistent feedback, for example mapping up/down dimple displacement to sliding left/right, was more difficult to interpret.

Range and resolution Remapping values to work within the provided (force) range is an issue for spatial consistency, in particular when fine-grained feedback is required. The resolution of actuation is both a technical and human issue. Technical limitations may affect the physical range and levels of feedback one can provide. With our current device, the range was a physical displacement of 5mm, with 160 steps. Through the just-noticeable difference experiment, we showed that users can detect differences of around 0.1 mm, showing the differentiable range is high.

4.4.6 Conclusion

In this section, we presented a concept of back-to-front relayed feedback on smartphones and discussed the results of our evaluations. HapticPhone is a continuation of the two prototypes presented in the previous sections, with the claim to make haptic feedback more mobile and compact. According to research question (RQ1) we showed the possibility of haptic feedback being implemented together with a smartphone; additionally, we showed how to improve touchscreen accuracy. Our design and studies focused on a novel smartphone interface that provides servo-actuated force feedback at the back of the device, which overcomes physical form-factor constraints for providing force feedback on the front. The prototype and the evaluation answer our sub-question 3 and serve to achieve Aim 1. In contrast to Jang et al. [JKT⁺16], who have also researched haptic feedback for smartphones, we use the smartphone's rear for feedback and not its edges. Our approach allows direct feedback, as thumb and index finger are connected more naturally. Through numerous hardware iterations we have also noticed how important a stable grip and ergonomic aspects are. Experience has shown us that both aspects are very important when designing a compact device. With the acquired knowledge, further development and refinement could take place in this area in particular. The questions could be, how can different finger lengths be supported and how can a stable grip be guaranteed.

Through our evaluations, we showed that relayed feedback can improve interaction on the frontal display. Based on the results, we discussed several factors affecting the relayed back-to-front feedback paradigm. Beside our developed prototype we would like to summarize our main findings:

• Psychophysical analysis revealed that perception of force stimuli using the presented mechanisms resulted in a Weber Fraction (8.29%) that is comparable to other

devices, confirming that feedback works appropriately and affords the perception of fine differences.

• Both selection and drag-and-drop task show encouraging results. In most situations, force and tactile feedback performed about equally well. For both selection and drag-and-drop, as expected, feedback yielded much better objective performance than without. Accuracy for small targets was increased by factor of 2.5 (for selection) and 1.3 (for drag-and-drop). Especially for selection of smaller targets, force feedback can produce better results than tactile, extending previous findings that showed the potential of back-of-device interaction for selection of smaller targets [WFB⁺07]. Users preferred force feedback over tactile or no feedback, feeling more accurate, yet not faster. Thus, in terms of subjective performance, force feedback was the best option.

As a next step, we intend to look into eyes-off interaction and how to easily code informations using a pin-based approach, can these pins also be used as input. Another direction could be, looking at factors affecting relayed feedback points to the potential for many different types of devices, for example those that can flip or move. We did not discuss these options, as this is beyond the scope of back-to-front feedback relay. Finally, as HapticPhone already is a fairly compact design, we are considering if it can be offered as a compact extension to existing smartphone setups.

4.5 Summary

All three sections in this chapter have shown how versatile the approaches for augmenting touchscreens with haptic feedback can be. All three objectives were achieved. ForceTab, our first prototype actuates the whole touchscreen through a motion platform. With this prototype, we were able to show the limitations and advantages of haptic exploration through three sub-studies (Objective 1). With regard to our hypotheses, it was shown that the users can clearly distinguish between three different resistance levels. Furthermore, it was shown which shapes can be perceived via the implemented feedback mechanism. It is not possible to perceive complex shapes without visual support. Larger shapes or a slower exploration speed increases the recognition rate. When finding and differentiating a specific region, it can be said that smaller differences are harder to identify and that the shape has an influence on the accuracy. Finally, the approach of finding a specific region with haptic feedback is very well suited, and the users were able to achieve a good result regardless of the shape.



Figure 4.33: Sketch of implemented haptic feedback approaches. From rigid to compliant to pen-based haptic feedback. All three approaches extend the interaction with a touchscreen (black). Yellow and blue symbolised possible actuations.

With our second prototype, the second objective was realised. A prototype was developed that uses an elastic material above a touchscreen to provide haptic feedback. The material's tension can be changed using servo motors, thus creating different levels of resistance, enabling a novel haptic exploration. In the most relaxed state, the touchscreen can also be included in its original rigid form. We were able to identify the perceptional threshold for a press and a slide event through a psychophysical study. The results of the study were incorporated into an application that has been presented.

Finally, HapticPhone was presented. There were two challenges to overcome: firstly, to combine haptic feedback with a mobile touchscreen (smartphone) and secondly, to increase the touch accuracy with the help of the designed feedback approach (objective 3). As with the previous prototypes, a perception study was first performed in order to identify the limitations. Subsequently, two standard tasks (selection and drag-and-drop) were explored to show the influence the implemented haptic feedback. The third sub-question could thus be answered, a smartphone prototype that supports a pin-based haptic feedback approach has been developed. Additionally, both of our hypotheses in this section could be accepted. The presented approach can increase the touch accuracy as well as the subjective perception of the performance.

The creation and evaluation of the three prototypes presented have shown the multifaceted possibilities of the interplay between haptic feedback and touchscreen interaction. Research question 1 was addressed in different ways with all three prototypes. The results of all studies are included in the answer RQ1. Several improvements could be shown, limitations identified and several approaches demonstrated, reflecting the versatility. The haptic feedback range regarding touchscreen interaction was extended by 1) an actuated tablet providing rigid haptic feedback, 2) a variable compliant feedback approach using a stylus and 3) a back-of-the-device approach providing pin-based haptic feedback. The studies, conducted in Section 4.2 provided an understanding of the capabilities and limitations

of comparable platforms such as those used by Kim et al. [KHSK14]. The system of Sinclair et al. [SPB14] showed a better recognition rate in 1-DOF, but compared to our study, users could see the visual display device, which strongly supports the identification of shapes [Gru08]. The interface presented in Section 4.3 shows further possibilities and exceeds previously presented pen interfaces [CBMB16], [WRSS16]. Finally, a third feedback mechanism for smartphones has been introduced (Section 4.4), which makes use of the back of the device. The variety also becomes visible when comparing the two psychophysical experiments. Looking at the participants' comments, it becomes clear that the differentiability of flexible material was more complicated compared to our pin-based feedback. From a technical point of view, both experiments are difficult to compare, since one was done with a pen and the other provided feedback directly on the index finger. Looking at the haptic exploration and perception described by Lederman [LK09], FleXurface could instead be put into the category pressure (hardness), while ForceTab fits into the category global shape; however, there is no clear category for HapticPhone. Additionally, the wrist also plays a role in perception. Just as ForceTab and FleXurface allowed partial wrist movements, HapticPhone only moved the fingers.

To conclude, in this chapter, we have explored three different approaches to augment touchscreens. We have studied performance and perception-based aspects. In the next chapter (see Chapter 5), we will continue exploring haptic interfaces and address the second research question of this thesis. Instead of touchscreens, we will explore haptic interfaces in virtual and remote environments. Specifically, two prototypes will be presented that investigate perceptual aspects.

Chapter 5

Foot Haptic Interfaces for VE and RE

5.1 Introduction

While the previous chapter (see Chapter 4) dealt with the augmentation of touchscreen interaction with haptic feedback, thus mainly focusing on hand-driven interaction. Doing so, three prototypes were presented, all pursuing a goal: introducing novel ways of haptic feedback. Accordingly, we explored aspects such as touch accuracy, the perception of shapes and recommended a set of application scenarios. In this chapter, additional haptic user interfaces are introduced, the focus shifts from user interfaces designed for the hands to interfaces focusing on the feet. Therefore, unconventional interface concepts will be considered. While the first three prototypes were confined to touchscreen interaction, in this chapter, we will explore how haptic feedback on the feet can affect realism within virtual and remote environments. In the following, remote environments are considered as telepresence as well as virtual reality scenarios. Since in both cases, the user immerses in either an artificial or remote environment. In line with the previous chapter, further haptic user interfaces are presented. Specifically, this chapter addresses the second research question (RQ2):

How can foot haptics for navigation tasks be designed to improve self-motion perception and spatial awareness in virtual and remote environments?

Similar to the previous chapter, we will pursue objectives (Objective 4 and 5) in this chapter, which will help to achieve Aim 2. Additionally, for this purpose, two sub-questions were formulated in Section 2.3.2 in order to answer research question 2. The wide spectrum of haptic feedback design remains and is also visible in the following sections.

There are many ways to provide haptic feedback, and of course, we can haptically stimulate any part of the body that is capable of doing so. However, the feet were chosen because we see great potential in foot-based haptic feedback. Firstly, they are rarely involved in the interaction between human and computer and are, therefore, unlike the hands, "free". Secondly, the feet are also haptically stimulated in everyday life - for example, when walking. Thus, for some applications, it can be useful to experience similar stimuli as in real life. Several haptic feedback interfaces in each of our application areas have been presented in the Chapter 2. For example, research on the simulation of ground textures was introduced [NNTS12], [PFC⁺10], to make the user experience more realistic.

However, the literature review has also shown that there are still open questions. Therefore the following chapter is dedicated to answering RQ2. In order to do so, objective 4 was implemented with our fourth prototype. An approach was developed that maps spatial data of a remote environment onto the user's feet to increase spatial awareness. To sense the remote environment, a telepresence robot is extended with a for this purpose developed sensor ring. The fourth sub-question (RQ 2.1) was to investigate *how foot haptics support the user's awareness when interacting in remote environments.* Of course, there are many approaches that use haptic feedback combined robots, however, they are mainly used for assembly support [HWMZ91], when handling dangerous objects or to navigate unmanned vehicles [JS19]. All of them have only one main task, unlike in our system where the social component is more important than the manipulation of a robot. With our interface, proximity (vibrations) and collision (small impacts) cues are passed from the remote robot to the feet. More specifically, to the side of the feet, as these reflect the main directions (forward, backward, left and right). We will present two studies that address the challenges and direct our attention towards Aim 2 to address RQ 2. Our fifth and final prototype is designed for use in virtual realities. It has been designed to implement cue combinations using two interaction metaphors to investigate how can haptic feedback be designed to enhance self-motion perception in virtual environments? (RQ2.2). Through an exploratory study with five different sessions, we will implement objective 5. Which will further contribute to answering RQ2. It additionally explores how foot-based haptic feedback can be used to improve realism and related aspects of perception in a standing and sitting condition. As mentioned previously, this chapter addresses research in the areas: virtual reality and telepresence. For both cases the prototypes itself will be presented, similar to the previous chapter, the design of the studies performed and the results achieved will be presented and discussed.

5.2 Foot-based Haptic Proximity Cues in VE und RE

Telepresence robots are mobile platforms allowing people to interact remotely. A mobile telepresence robot typically contains a video conferencing system, and it can be controlled from the remote. They can enable people to participate remotely in a variety of spaces and situations, including remote attendance at academic conferences [NVPH16], [RN17], home schooling [NO17], and remote office work and meetings [RTM12]. Research has identified benefits for mobility in remote spaces and situations, including an increased sense of social presence [YJNS18]. However, despite these advantages, it is not always straightforward when using a telepresence robot and being aware of the surrounding environment, especially in crowded places such as conferences and social meetings, where large numbers of people move around, often unpredictably [RN17], and where people often try to follow or actively participate in conversations.

In order to support users to immerse stronger into a remote environment and to become more aware of spatial structures, we have developed a haptic feedback approach. The prototype could also help users to navigate the robot more precisely while at the same time increase spatial awareness and confidence of the user. When using the interface, users have to simply place their feet inside the system and receive haptic feedback on their feet when approaching obstacles in the remote space (Figure 4.32). The main objective of the user interface is to provide additional non-visual feedback on the physical space in the remote environment to the users so that they can better identify and understand obstacles and make improved navigation decisions. Another aim is how this type of feedback can be designed to enhance the feeling of spatial presence, since a greater presence cloud sometimes lead to more focus and investment in an activity or space. We specifically chose the feet in order to keep the hands free for other tasks, including using a keyboard/mouse or gaming controller to drive the robot. Additionally, the orientation of the feet helps to provide directional feedback, as feet, unlike the hands, normally reflect the direction of movement. Therefore, users should be able to infer the direction an obstacle is at based on where on their feet they feel the feedback. Even more so, the feet might naturally bump into things when moving about a space in person, and are not typically used at all to interact with telepresence robots.

Two user studies were conducted to explore the potentials and limitations of the designed user interface. The goal of the first study was mainly to gain an early understanding of how it affected users' behaviours in a virtual environment as well as their understanding and awareness of the environment. Moreover, this first study sought to explore the potential impact of the system on spatial presence, i.e. the feeling of "being there" [WS98]. While the first study was a strong controllable environment, the second study, focuses more into real environments or scenarios, respectively. Here, the goal was to study how people would use and experience the system while driving the robot in an environment with other people and obstacles.

5.2.1 Implementation

In many telepresence configurations, both visual as well as audio channels are already used for the key activities (e.g. talking to people, looking at objects, etc.). Consequently, it is important aspect of the introduced haptic feedback is that an unused sense for spatial perception in included to ensure that other sensory channels remain free for other activities. Therefore, we have chosen the feet as target of the interface, as it is often the first part of the body that is touched when a person walks into something or is bumped by something and the feet can be seen as the natural place for providing this kind of feedback. It may also be one of the best places on the body to provide directional (i.e., left, right, forward, backward, etc.) feedback, as it is easy to place sensors all around the users' feet and keep them relatively stable (e.g., versus their hands, which may move frequently throughout use) when users operate the system while being seated. In addition, the feet normally point forwards or in the direction of walking, and thus they make a good virtual reference point.



Figure 5.1: The ring consisting of 12 ultrasonic distance sensors attached to the Beam+ telepresence robot enables 360° coverage with a radius of up to 4m.

Figure 5.2: The foot platform consisting of six movable actuator blocks. In the detail drawing the vibration motors in red (proximity feedback), the solenoid is shown in green (collision).

In our experiments we used an augmented version of a Beam+ telepresence robot created by Suitable Technologies. The Beam is 134 cm tall with a 25.4 cm (10 inch) LCD monitor and a maximum speed of 2km/h. Two wide-angle HDR cameras are attached to the robot. One camera points towards the floor to provide a navigational view (e.g., it shows other people's legs and feet), while the other points forward to show other people (e.g., their bodies and faces) and the environment. We extended the telepresence robot with a purpose-built sensor ring (see Figure 5.1) consisting of 12 ultrasonic range sensors (HC-SR0). Our prototype system is composed of a high-density wood and aluminium foot platform to which actuator blocks are attached (see Figure 5.2). Users remove their shoes and place their feet within the foot platform. Based on the distance sensors on the robot's sensor ring, users receive directional vibration cues when an object is sensed, and vibration frequency increases as the object gets closer to the robot. Once an object is in a predetermined collision range, a solenoid is triggered and provides a soft but noticeable impact to the user's foot. This is similar to a collision with a wall, but before actually hitting an obstacle, so users have enough time to avoid a collision. An overview about all components is depict in Figure 5.3.



Figure 5.3: A schematic representation of the whole interface and its components. An iteration from sensing to robot control is shown.

Sensor Ring Design

The distance range afforded by the sensors on the ring is between 2 and 400 cm at a frequency of 40 kHz, with a horizontal sensing cone of 30° . As such, the robot can detect objects in a radius of approximately 400 cm around the robot. The selected sensors are sufficient for the requirements of the interface, as we initially aim to collect feedback in both following exploratory studies. For higher precision or faster response times, an optical sensor, for example on a laser basis, would probably be recommended.

The sensors were connected to a Raspberry Pi 3 (Rev3 Model b+). In order to cover 360° and to avoid signal overlap, all 12 sensors were grouped into 3 disjunct classes. This allows controlling all 3 groups sequentially with a delay of 50 ms, so that the entire area can be scanned within 150ms with an accuracy of up to 3mm according to the specification of the sensors. A microcontroller (Arduino Uno Rev3) manages the introduced grouping and collects distance data from the sensors, and passes it to a Python script on the

150

Raspberry Pi. All distance data are then sent via Wi-Fi to the remote-control PC using OpenSoundControl. Both the Arduino and Raspberry Pi are self-contained, powered by the robot's battery using the 5v connections provided.

Feedback Design

The foot platform contains a series of actuator blocks as illustrated in Figure 5.2, individually adjustable by moving the bars to support most foot sizes (between 21.0cm and 30.0cm). An actuator block consists of a vibration motor (Precision Microdrive 304-116), a solenoid (ROB 10391, 36V), and a coil spring mechanism to provide a movable linkage required for collision feedback. For a more comfortable fit, a 1cm thick of compressible firm foam is attached to the foot-directed surface. A 5mm cylindrical vibration motor was integrated into the foam to provide tactile proximity feedback. Our prototype platform consists of three actuator blocks per foot, mounted at the front, the back and the outer edge of each foot. In addition, the six locations of the actuator blocks reflect potential collision points of the real robot. Both inner sides of the feet were not equipped with actuation blocks, as there is no potential collision point at these locations. Two Arduino Mega were used to control all actuator blocks, one for each foot. In addition, a graphical user interface displays the distance data of the sensors and allows the configuration of the parameters, such as proximity range.

Since 12 sensors were required to cover 360°, a mapping was necessary to map the 12 sensors to the six actuator blocks on the foot platform. Both the left and right sensor groups were explicitly assigned to the corresponding actuator blocks. Each front and back actuator block shared a front and rear facing sensor, plus a position-dependent neighbour that was added either to the right or left block. In order to guarantee unique feedback to the users, no cross-fade occurred, the feedback group that recognized the closest obstacle was activated.

The design of the foot platform was made to be easily adaptable to different foot sizes. The design allowed us to easily and quickly adjust the vibrotactile contact points so that they touch the user's feet on all sides (front, back, left, right), thus allowing the user to feel the feedback on all sides. While we could have instead designed a different form factor (e.g., a pair of shoes), we chose the platform design because it was a good balance between (a) making sure all of the motors touch at the right points (for directional feedback), ensuring proper feedback, and (b) being easy and quick to adjust to the various foot sizes of our participants. The form factor was not ideal (i.e., big and bulky). However in our studies we mainly focused on the higher-level idea of providing gradual directional haptic feedback in telepresence situations instead of the underlying technology.

Two major modes for proximity feedback are supported in our implementation: *continuous vibration feedback* that increases the intensity and *pulsing feedback* that increases frequency in relation to the distance when an object is sensed. The closer an object gets to the robot, the higher the intensity or frequency of the vibration, respectively. Our implementation supports a linear as well as a polynomial proximity function, and an adjustable range for activating proximity and collision feedback. In addition, the duration of the collision feedback and a maximal intensity for proximity can be adjusted. Both amplitude and the frequency of vibration are determined by the hardware and ranges from 0-1.4g (amplitude) and 0-300Hz (frequency). This means that the bump of the collision is clearly distinguishable from the vibration. Values chosen for the current studies were based on pilot testing.

5.2.2 Experiment 1 - Virtual Environment Navigation

The aim of the following two studies is to gain an understanding of how the introduced interface affects users behaviour and their perception of spatial presence. The key difference between both experiments is that the first was conducted in a virtual environment, so that all parameters were well controllable. In contrast, the second study was designed to evaluate the system in real world scenarios. This means that in our first study we used the haptic feedback platform for the feet, but the telepresence robot was simulated in a virtual environment (see Figure 5.4). For an initial exploratory evaluation this has several advantages: firstly, a prototype can be developed quickly and adapted if necessary. Secondly, if the feedback has a negative effect on the robot's navigation, neither the environment nor the robot can be damaged. Both visual feedback and robot control were adapted to the real use of a Beam+. For example, as in the second study, the users use a game controller to operate the robot. The first study was mainly designed to test the following hypotheses:

- H1: Proximity cues on the feet increase spatial awareness while navigation in VR.
- H2: Users will complete the task faster with the proposed feedback approach.
- H3: With haptic feedback, users will make fewer collisions.



Figure 5.4: A diagrammatic representation of the study design, structure and procedure as well as the methods of analysis.

Participants

A total of 19 participants were acquired either via social media postings, posters on our university campus or an online study registration system for the students of our department. The participants were 19-41 years old (M=24/SD=5), 12 women, six men and one non-binary gender. Most participants were students, while the others were casual workers. All the participants took part in the study voluntarily and no money was paid for participation.

Apparatus and Procedure

As the first study was designed to have a greater influence on the environmental parameters, it has been performed in a simulated virtual environment. To do so, the participants stood in front of a large television screen and navigated a virtual robot through a 3D scene. In order to simulate the real distance sensors, a ray-casting approach was implemented to determine the distance between the robot and the environment, as well as mappings for collision and proximity ranges were computed.

We used an within subject design. All participants had to operate the robot two times along a similar, but not the same, track, in a counterbalanced order - once with and once without the proposed feedback. Both tracks were defined by a start and an end position. The display we used was an 147 cm-wide TV screen (in 16:9 format) standing on a 75 cm table. Participants stood about 1m from the screen, and stood in the same position in both conditions. Controlling the robot was also the same in both conditions (with and without feedback), and users were limited to only use the analogue stick of a PlayStation 3 controller for navigation. During navigation a sideways shifts (strafing) was excluded, so that navigation only allowed forward or backward movements and turns to the left or right. Based on pilot testing, we used a collision range of 20cm and a proximity range of 2m with a linear proximity function. Maximum intensity of the tactile feedback was set to 100% (about ± 1 G peak amplitude; about 240 Hz) and the solenoids were driven with 24 V.

Before each condition the participants were asked to test-run the particular interface. This training was intended that the user get used to the system and to get an understanding of how the system, especially the feedback works. The test-run scenario was a small virtual room that contained objects. As soon as the participants felt confident in using the system they were asked to began the task. Per condition all participants were given an overhead map of the virtual environment and they were told where the final position was located.

They were not told where their starting position was (i.e. they had to find out where they were in the surroundings). To further look into spatial presence aspects, the virtual environment includes walls, static obstacles and moving avatars. In both conditions, a click sound played whenever the simulated robot collided with an object or an avatar. The sensation of the subject's spatial presence was measured with a slightly modified version of the Witmer & Singer questionnaire [WS98]. The modified questionnaire asked 19 questions on 7-point Likert scale and measured six different aspects of spatial presence (see Questionnaire 6), which were combined to give a general understanding of spatial presence. The six individual factors were: realism, possibility of action, quality of the interface, possibility of investigation, self-assessment of performance.

The modified questionnaire is the same as the original, except that three questions on sound and two questions on haptics (questions 20-24 in the Witmer & Singer questionnaire) have been removed. The three sound questions have been removed as the virtual environment did not have much sound. The two haptics questions have been removed for two reasons: haptics was not part of the experience for the no-feedback condition, and we were interested instead in the impact that haptic feedback had on the other six individual factors measured in the presence questionnaire.

As mentioned above, the order of both feedback conditions and environment layout were counterbalanced. After completing the study tasks, we conducted short semi-structured interviews with participants to get their thoughts about the interface and its feedback. In total, each navigation task took approximately 5-10 minute, while the survey took 10-15 minutes to complete, and interviews lasted about 5 minutes. The quantitative data was analysed using dependent (repeated-measures, within-subjects) t-tests. A Shapiro-Wilk test was used to test the normality of the data, and Wilcoxon Signed-ranks test were used in place of t-tests when the data was not normally distributed. All data were analysed using the statistical functions of SciPy an open-source software for mathematics, science, and engineering ¹.

Results

Quantitative Findings The Witmer and Singer Presence Questionnaire was chosen for analysis as it is important to know to what extent the users felt involved into the remote environment. In accordance with the questionnaire, the following six factors have been assessed: presence, realism, possibility to act, interface quality, possibility to examine and self-evaluation of performance. All six factors were questioned for both study

¹SciPy.org https://docs.scipy.org/doc/scipy-0.14.0/reference/index.html

conditions (with haptics, no feedback). All results are shown in Figure 5.5; the box-plots represent the means (circle), medians, first and third quartile, minimum and maximum plus detected outliers. A significant effect was found for two factors (presence and realim). Since one participant answered only some of the items on the questionnaire, the following results refer only to the 18 participants who answered all questions.



Figure 5.5: Summary of the individual ratings of the Wittmer and Singer Presence questionnaire for both conditions of study one. There was a significant effect for those highlighted in orange.

The overall ratings of the participants for spatial presence were significantly higher with haptics (M = 5.26/SD = 0.82) than without (M = 4.93/SD = 0.91). A t-test showed that participants had a stronger sense of presence with the proposed interface (t(17) = 2.42, p = 0.03). A similar result was achieved by the sub scores relating to the sensation of realism(with haptics: M = 4.94/SD = 1.10, no feedback: M = 4.21/SD = 1.45). Users felt the remote environment with haptic feedback significantly more real than the environment without feedback (t(17) = 3.13, p = 0.006). For the remaining four factors, there were no significant findings, however, for sake of completeness, a list of the means and standard deviation follows, that are also depict in Figure 5.6. Subjects scores for a slightly higher possibility to act with haptics (M = 5.53/SD = 0.99) than with no feedback (M = 5.35/SD = 1.21). A similar trend has also been observed for the factors: interface quality (with haptics M = 5.42/SD = 1.41, no feedback M = 5.29/SD = 1.53) and possibility to examine (with haptics M = 5.19/SD = 1.07, no feedback M = 4.95/SD = 1.25). Self-assessment of performance does not follow the preceding tendency (with haptic

M = 5.75/SD = 1.24, no feedback M = 6.01/SD = 0.93), participants felt to perform better without the introduced interface.



Figure 5.6: Ratings of the individual questions that showed a significant effect.

When analysing the questionnaire in detail, a significant effect can be observed for three individual questions (6,10 and 13). Question 6: "How compelling was your sense of objects moving through space?", participants found that moving objects were more intensively with than the presented interfaces (M = 5.78/SD = 1.31) than without (M = 4.33/SD = 1.70). A Wilcoxon test shows a significant effect of the responses (Z = 70, p = 0.006). Similarly, question 10: "How compelling was your sense of moving around inside the virtual environment?", participants also found that the sense of moving around inside the virtual environment?", participants also found that the sense of moving around inside the virtual environment was significantly more compelling with haptics (M = 5.22/SD = 1.62) that without (M = 4.45/SD = 1.78; Z = 55.5, p = 0.021). Showing that a higher sense of self-motion can be achieved through the interface. Lastly, question 13: "How involved were you in the virtual environment experience?", participants reported higher involvement with haptics (M = 5.58/SD = 1.11) than without (M = 4.44/SD = 1.61; t(17) = 3.64, p = 0.002). Looking at the hypotheses, we can accept H1. The approach can significantly improve the overall spatial presence and create a more realistic experience when navigating in virtual space.

Besides the analysis of the above-mentioned presence questionnaire, additional performance parameters were also collected during the study. These included the total time per task and condition and the number of collisions during the task. No significant effect was found for either parameter total time nor number of collisions. Nevertheless, a brief summary of the results is given in the following. On average participants independent of the condition needed in total M = 352.84 s with a SD = 191.51 s. Surprisingly, the participants needed 14% more time compared to the average. When looking at the collisions per minute it can be seen that with haptics (M = 0.453/SD = 0.31) there were slightly less collisions than without feedback (M = 0.48/SD = 0.59). Even the two slightly different rooms had no influence on the total time and collisions under the respective conditions. Finally, the influence of the order of conditions was examined, showing that the second run was independent of the condition, clearly faster with a similar number of collisions. The participants needed longer in the first session (with haptics M = 478.61 s/SD = 183.30 s; no feedback M = 334.90 s/SD = 163.76 s) than in the second (with haptics M = 317.11 s/SD = 250.13 s; no feedback M = 275.37 s/SD = 126.38 s). The hypotheses H2 and H3 that with our approach the participants are faster and make fewer collisions must be rejected. Surprisingly, users took longer with haptic feedback than without. Two possible reasons for this are that the feedback has to be understood and interpreted before interacting or that the users are more cautious because of the feedback.

Users' Perceptions In line with the quantitative analyses, the qualitative analysis shows that the participants enjoyed the interface and found it helpful for most of them, thus fulfilling its purpose. Participants described the vibration as providing a tingling sensation at its strongest and being only a little noticeable at its weakest. They described the collision feedback as a light tap which was sometimes surprising, but never painful. Due to the haptic feedback, a greater attentiveness or caution in controlling the robot in the remote environment could be observed.

"The vibration caused me to be cautious, just as in the real world..." (P9)

A comment suggests a similar effect towards the awareness of objects in the remote environment.

"[The interface] made me more aware of how close objects were..." (P8)

A further participant reported:

"Vibrations made me want to stay away [from objects]..." (P2)

In this case, the sensation of awareness or caution has been slightly too strong, the aim of the interface is to make the user more aware of the environment, but not to block users from "normal" action. However, it was not only objects that were given more attention by the interface, but also the avatars were perceived with a more focused sensibility.

"I was trying not to hit people and objects..." (P3)

Participants said they consciously and subconsciously moved around the space more cautiously when using our feedback than without, often being more hesitant in their actions. While this hesitance allowed participants to be more careful not to bump into things, it also prevented some participants from exploring more.

"I was less inclined to explore a bit [with haptic feedback]. Because of the size of the doorways, I would always get a buzz when walking in." (P12)

Likely due to being worried about bumping into something. Participants also found that haptic feedback increased their anxiety and stress.

"Vibrations made me feel anxious at times (definitely more than having no vibrations).... It's good for alerting of the odd object, but when I'm moving through a narrow walkway, it feels more stressful than helpful" (P2)

There are further comments in a similar direction, however, none of them express that haptic feedback is too stressful, but that it is simply too much. Therefore, some felt that the feedback distracted attention from the actual task.

"I get a lot of warnings when I'm close to something, but sometimes too much..." (P2)

"It felt harder doing the task [with haptic feedback], but that's probably because my feet was getting constant feedback. My feet were vibrating almost all the time, so that was a little bit distracting." (P12)

However, most agreed that even with the distraction that the haptic feedback caused, it was still useful in certain cases. Rather, they felt that it should only be used in crucial circumstances (e.g. when the perception of a nearby object could prevent a collision). Participants generally found the feedback most useful when they were moving or when they were addressed by a moving object while they were stationary.

Many participants said that their involvement and immersion in the activity with foot haptics has increased, which has made participants feel more responsible for their behaviour and more aware of the consequences of their actions.

"Haptic feedback made me feel more engaged." (P6)

"Moving without the haptic feedback definitely felt emptier. And there was less awareness of proximity to objects as there was no move vibration." (P5) "It felt great to experience real consequences while exploring [the] virtual environment." (P9)

Summary The findings from our study showed that our approach could significantly improve the overall spatial presence and create a more realistic experience when navigating in a virtual space. This was promising as it demonstrated the potential for the proposed interface. It provides additional spatial awareness and increases the sense of presence and participation when operating a telepresence robot. It was also found that our interface caused some users to be more careful with their environment, which we found positive. However, under certain circumstances the feedback was distracting and caused anxiety. This indicated that it would be important to adjust the amount and frequency of feedback and consider adjusting the feedback intervals. We were also interested to see if these results could be transferred to practice. We were also interested to see how the prototype would affect user behaviour and interactions with other people when a telepresence robot is operated in a real environment. Our results and further aspects of RQ2 led us to our second study.

5.2.3 Experiment 2 - Real Environment Tasks

Motivated by the results of the first study and driven by our Research Question 2 we are interested in researching the interface within real telepresence environments that are less controllable. In the second study, participants are asked to visit a university campus with the introduced telepresence robot (see overview in Figure 5.7). They will encounter a range of activities that include many real-life situations that people would experience when using a telepresence robot either at work or at a conference; e.g., talking to people, finding places, navigating through corridors, searching for objects. The objective was to understand how user behaviour with foot haptics would look like compared to that no feedback. Similar to the first experiment we tested the hypothesis:

H4: Proximity cues on the feet increase spatial awareness while navigation in VR.

Participants

A total of 17 participants were recruited through snowballing (word of mouth), social media (Twitter and Facebook) and posters placed around our university. Fourteen participants had never used a telepresence robot before, and two participants had used a telepresence



Figure 5.7: A diagrammatic representation of the study design, structure and procedure as well as the methods of analysis.

robot, but without any haptic feedback. Among the participants were eight women and nine men aged 19 to 55 (M = 25/SD = 8). The professions covered a range of occupations and activities (e.g. university students, consultants, administrative staff). All participants had smartphones and used video conferencing technologies (e.g. Skype). All participants were paid 15 dollars for participation.

Apparatus and Procedure

The second study was designed as an exploratory study, as we were mainly interested in how haptic feedback effects the behaviour of the participants in real world situations. We also wanted them to be able to compare their experience with haptics and with the experience without. Based on the results of experiment 1, we set the feedback parameters to a collision range of 20 cm and an proximity range of 1 m with a linear approach function (experiment 1 had a collision range of 20 cm and an approach range of 2 m). The one-meter length was based on measurements of all rooms where the participants were asked to perform the tasks. The maximum intensity of the vibrators was set to 60%(approx. ± 0.6 G peak amplitude; approx. 150Hz) and the magnets were driven at 24V (experiment 1 had 100% vibration intensity and 24V). We assumed that the new settings would lead to a lower level of distraction (lower intensity) and less stress due to too much feedback (lower proximity range). Furthermore, it was important for us to ensure that the robot's navigation was as real as possible. For this reason, the participants sat in front of a 55 cm wide monitor with their feet placed on the foot platform and used a PlayStation 3 controller to control a beam telepresence robot, similar to a normal telepresence user would do. A diagrammatic overview of the second study can be found in Figure 5.7.

There were four different tasks, the tasks were chosen in a way that allowed studying a wide range of everyday tasks. These included tasks to provide opportunities for social interactions (e.g., asking people to open a door, press an elevator button, and ask a question from an advisor) as well as manoeuvring in both confined (e.g., narrow hallway, book-store aisles) and wide-open spaces. External factors such as the number of people in the places where each task took place and physical obstacles on the way were taken into account when designing the tasks. In contrast to the first study, we are aware that the parameters of real situations change from task to task. In contrast to the first study, we are aware that the parameters of real situations change from task to task. However, in order to keep the changes low, we have scheduled each session at the same time of day. The participants received relatively few instructions so that they could "go their own way", i.e. the variant that seems most suitable for each participant. A schematic reference map was given to the participants, on which the locations of the four tasks were marked.

Similar to the first study, each task was completed twice, in the opposite direction - once with feedback and once without.

Task 1 Navigate through a narrow hallway and find the poster board. Then look at the poster which has "The Code" in its title and take a screenshot of that poster.

Goals driving in narrow spaces, move along a wall, search for patterns

Task 2 Find out what kind of courses you can take. Go to room 2747 or 2745 (on the same hallway) and ask a student advisor to show the course table for undergraduate program.

Goals entering a room, interact directly with another person

Task 3 Drive through an "accessibility door", ask passers-by to push the handicap-door button to open it. Then go the book-store and find the textbook for IAT ARCH 100.

Goals interact with passers-by, navigate and search in a store

Task 4 Move toward the tables in the mezzanine as shown on the map. Find this person [headshot shown on an iPad] and say "hi". Then park the robot next to the person.

Goals search for something at distance, navigate around many obstacles

After completing the above four tasks for the first time, the participants completed two questionnaires. The first was the same presence questionnaire used in the first study (based on [WS98]), but we added the questions about sound (questions 20-22 in the original questionnaire) again because this experiment included sound (see Questionnaire 6). We still excluded the questions on haptics and touch (questions 23 and 24 in the original questionnaire). The second questionnaire was the NASA Task Load Index (TLX) [HS88] to measure the perceived workload on six scales. In addition to its effects on spatial awareness, we were also interested in finding out whether our interface affected the participants' feeling of how hard they were working or mentally concentrated. Participants were asked to rate mental stress, physical demand, temporal stress, performance, effort and frustration using 7-point Likert scales from "very low" to "very high".

Before the second session the feedback condition (with haptics, no feedback) is switched, and participants have to perform similar four tasks but in reverse order. Of course, we did not modify the tasks fundamentally, only the information to be sought was exchanged. For example, participants had to find a different person in the crowded table configuration, look for a different course textbook in the book-store, etc. After completing the final four tasks, participants drove back to the start point of the study. They then completed the spatial presence and NASA TLX questionnaires again for the second condition. The order of conditions (with haptics or no feedback) was counterbalanced across participants.

Before each condition participants are asked to practice. The study began as soon as they said they were used to the interface condition.

Similar to the first study, we conducted semi-structured interviews after completing the tasks. The first phase of the interview focused on understanding their previous experiences using telepresence technologies and video conferencing systems. The second phase of the interview focused specifically on their experiences using the robot with the introduced haptic interface.

Data Collection and Analysis

All eight tasks took about 40 minutes in total (~20 minutes for each condition), while the questionnaires lasted 10-15 minutes in total. We recorded all our interviews on audio with the consent of the participants and made detailed notes. The interviews lasted about 15 minutes and all interview data was transcribed. We recorded a video of the user driving the robot by recording the screen showing the robot's two cameras

A pattern analysis of our interview data was performed in order to identify key issues and categorise them. All videos were reviewed to gain additional understanding of what was happening. The quantitative data were analysed in the same way as in the first experiment. P10 was not able to complete all eight tasks due to battery problem, so we have not considered P10's quantitative data in the analysis. In contrast, we have decided to include P10's qualitative data in the analysis.

Results

Quantitative findings Participant scores for overall spatial presence were similar with (M = 5.09/SD = 0.49) and without the use of haptic feedback (M = 5.06/SD = 0.69). Unfortunately the analysis of the overall spatial presence and the partial aspects, such as realism, possibility to act, quality of interface, possibility to examine, self-evaluation of performance, did not show a significant effect. Both conditions reached similar results in all aspects of the questionnaire (see Figure 5.8). For completeness, the mean values and the standard deviation for all sub-scores are listed in the following: realism (haptics: (M = 5.06/SD = 0.74); no feedback (M = 5.05/SD = 0.97)), possibility to act (haptics:

(M = 5.27/SD = 0.58); no feedback (M = 5.23/SD = 0.77)), interface quality (haptics: (M = 5.17/SD = 1.08); no feedback (M = 5.15/SD = 1.17)), possibility to examine (haptics: (M = 4.15/SD = 1.09); no feedback (M = 4.44/SD = 0.97)), self-evaluation of performance (haptics: (M = 5.59/SD = 0.59); no feedback (M = 5.47/SD = 0.72)).

When looking the Hypothesis H4 and the results of the individual questions of the questionnaire, we found that, unlike in the first study, in which our approach increased the feeling of spatial presence and involvement in the environment, none of these effects were transferred to the telepresence situations we tested and H4 have to be rejected.



Figure 5.8: Summary of the individual ratings of the Wittmer and Singer Presence questionnaire for both conditions of study two.

In terms of the perceived workload measured by NASA TLX, participant ratings were similar with (M = 36.90/SD = 18.09) and without haptic feedback (M = 36.60/SD = 16.15). Thus, the questionnaires did not show any noticeable difference in either presence or perceived workload when driving the robot with haptic feedback compared to without. A comparison of the respective sub-aspects of the two conditions (haptics and no feedback) can be found in Figure 5.9.

User's perception The main themes that could be identified from the interviews are confidence in haptic feedback, behaviour to avoid collisions and contextual differences that influenced the feedback.



Figure 5.9: Overview of the individual sub-scores of the NASA-TLX questionnaire for the both conditions of the study.

During the qualitative interview, 11 out of 17 participants reported that they found haptic feedback useful for better understanding the remote environment. They also found that they could feel if they came too close to obstacles and if someone was approaching from behind or beside the robot, especially if they could not see them on the screen. The haptic interface is a useful background warning system that provides additional information to what they have already seen, noted three participants. Two participants found it most useful when driving in an unfamiliar environment for the first time, but did not have a strong preference for using it once they had learned more about the remote location and got used to driving there. Furthermore, some participants believed that the interface helped them avoid collisions with objects and people. For example, it could be observed that some participants left more distance to objects or persons with haptic feedback than they did with no feedback.

"What it [haptic feedback] did was that in the first trial it trained me" (P1)

"It was pretty useful because obviously, I was getting close to obstacles." (P10)

"The vibration was kind of useful. Like looking at the rear mirror in a car, it kind of provided a different layer on top of what I already have and used to." (P2)
The haptic feedback also had an effect on the paths taken by the participants while driving the robot. For example, in tasks 1 and 2, where they had to drive through a narrow passage and ramp (see Figure 5.11, left), some participants drove in the middle of the ramp to get the least vibration, while some drove on the side to leave more space for others. This may be due to a increased understanding and perception of the environment. However, some users felt frustrated by the continuous tactile feedback.

"On the ramp, the robot was not in danger but it kept vibrating; it makes me ignore the feedback." (P5)

"In small places, like small corridor where there is nothing I can do, so the vibration feedback was making me annoyed and after a while, I stopped paying attention to it. There was nothing I could do, so I was just driving at the center." (P5)

In contrast to the above, the proximity feedback, i.e. the vibrations, does not feel disturbing. The levels of intensity chosen appear to be just right for some participants.

"Other vibrations were great, because it starts off slow, and then slowly increases as you get closer in the radius of objects in your way. That felt really great because it tells you are slowly getting closer." (P1)

"The intensity worked pretty well because it didn't feel like it was overwhelming to me and to the point that I was always distracted by it." (P2)

The feedback had clear effects on the behaviour of participants. For some, it meant changing their behaviour immediately and, for others, it meant questioning the feedback and thinking through what they should do next. Nine out of 17 participants took immediate action in order to avoid hitting objects and people whenever they received collision feedback; they would immediately slow down or stop the robot when the feedback occurred. This immediate reaction to collision feedback mostly happened in confined spaces where there were many obstacles within the collision range (see Figure 5.11, right).

"The kick really helped out in providing feedback that you're getting way too close that you'll hit it." (P10)

It was observed that some participants chose to keep more distance between the robot and the objects and people (see Figure 5.10). Although this behaviour was seen in a



Figure 5.10: A difference observed between haptic feedback and no feedback.



Figure 5.11: Driving the robot down a ramp (left) and driving in a confined space (right).

few cases, it was not consistent among all participants. That is, some participants, on occasion, performed the opposite - keeping less distance to the person and more distance to the object. This behaviour suggests that there might be other social and environmental factors that affect behaviour.

Other participants were not sure whether the feedback meant that they had collided with an obstacle or that they were getting too close to objects, but had not yet hit them. Thus, they could not clearly differentiate what in particular the feedback was alerting them to. In these cases, subjects would either continue on or investigate. For example, some subjects used the bottom-facing camera to learn and understand when the collision feedback was being triggered, and this was seen as being useful. In some cases, an immediate change in direction of the robot as a response to the feedback led to colliding with other nearby objects.

"I don't know if I hit any of those chairs or not, because when it kicks, does it mean that I'm really close or does it mean that I'm hitting the thing?" (P1)

Participants see a correlation between the context, i.e. the nature of the environment (e.g. limited vs. wide open, the number of obstacles or people in the environment) and the individual tasks. The feedback depends strongly on the environment, the task and also the navigation skills of the operator. In order to adjust the feedback in an optimal way, the minimum width of all rooms in which the participants were supposed to perform the tasks was measured before the study started, and thus the proximity range was set at 1 m. This meant that feedback would be given if the robot was within 1 m of an object.

This decision was taken to ensure that the participants received relatively consistent feedback throughout the tasks they were performing. However, it was found that this configuration was not practical for some of the tasks. For instance, the configuration was seen by participants to be useful for manoeuvring in the book-store and the advisor's office because the spaces were more constrained and participants wanted to avoid bumping into things. However, the same configuration was less useful when driving in spaces with consistent width and straight paths, such as hallways and ramps where the walls were within 1m. In these cases, participants received constant haptic feedback while there was nothing they needed to avoid (see Figure 5.11, left).

5.2.4 Discussion

Two different studies were conducted to evaluate the presented interface; both were exploratory. The first study examined the interface in a controlled virtual environment, whereas the second study dealt with real situations and was therefore less controlled. There were also differences in the technical implementation, whereas in Study 1 the feedback platform was "directly" connected to the virtual world, a network interface had to be added in the second study. Furthermore, in the first study the sensing of the environment was solved using software and in the second study a sensor ring was added to the robot. All this can of course lead to unwanted side effects such as higher latency or a misinterpreted object through the hardware ring.

User Behaviours and Reactions Our studies have shown that users find value in receiving haptic proximity feedback on the feet while driving a telepresence robot, and that such feedback is both easily understandable and can be usable for future design. After a short period of learning, all participants were able to appropriately interpret the feedback and were able to react to it in a proper way. In study 2, participants felt that their behaviour had changed when using the interface in telepresence situations compared to without. In both studies, the designed feedback had a positive effect on user behaviour in terms of improved awareness, encouraging participants to be more cautious in the distant environment. However, there were also participants who found the feedback disturbing and distracting. This is mainly due to the fact that in some cases the feedback was too intense or too continuous. In the following, the results will be discussed, with a focus on user behaviour and reactions and on use in different contexts.

Nonetheless, our findings suggest that there is value in continuing to explore haptic feedback options for telepresence robot operation as there is a range of uses for it. Such use may depend on the individual behaviour of the person using the robot; e.g., some people may desire feedback that is stronger vs. weaker, or in some types of environments and not others. There are design options for exploring ways in which users can easily control the mapping of haptic feedback to their understanding of the environment surrounding them. For example, haptic feedback could, at least initially, be combined with additional visual feedback on the screen so that participants can learn and understand which situations trigger which types of feedback. This could be reduced for more advanced drivers. The haptic feedback could also be used as a training mechanism for new drivers, and possibly removed for experienced drivers. This reflects our findings that some drivers only found value for the feedback while they were getting used to driving the robot.

While we have shown that haptic feedback can provides a heightened sense of presence in virtual environments (study 1), we cannot conclude that this is also the case for complex telepresence situations in which the user has to talk to people and respond to more demands from the remote environment (study 2). Thus, if the only goal is to provide a heightened sense of presence, our haptic feedback approach alone (at least in its current design) may not be the be-all-end-all tool to achieve this, at least for complex telepresence situations with many people and lots of obstacles. It may be worthwhile to explore, as future work, a more individual feedback per user which could be learned, and a more refined feedback that adapts to the environment, for example automatically adjust the velocity of the robot or proximity range.

Contextual Variations It was clear from the study that with systems such as the introduced, context is important. There were times when feedback was not necessary and participants were able to understand the environment from the camera perspective. At other times the feedback was too intense and continuous, and participants would have wished it had been adjusted accordingly. Our results also show that a future system will need to adapt more to the user and the context.

Naturally, providing context-dependent feedback solely on the spatial configuration of the scene is challenging. For example, in crowded locations (such as conferences), movements can be highly unpredictable [RN17]. Designers could potentially explore novel feedback mechanisms that consider context by addressing the range and density of objects surrounding the user, as well as the movement speed and time-to-contact with objects.

5.2.5 Conclusion

Both studies were to be formative in nature, allowing to explore design ideas around foot-based haptic feedback systems and to work on better understanding what works well and what doesn't. Study 1 was more controlled, and real world situations were not necessarily replicated, e.g. dynamic obstacles and human positions. Study 2 explored a wider range of situations and user reactions. As a result, not all participants were encountered with exactly the same tasks and scenarios, as the study was conducted in a public area of a university campus.

In addition, participants in both studies are not experienced to navigate a telepresence robot, most have done it for the first time, as opposed to experienced drivers who may already have developed navigation strategies when driving a robot. Furthermore, it is planned to investigate how the current results could be generalised to an extended use of the system, a wider range of users and capabilities with telepresence robots, and to other scenarios such as conferences and remote working environments. Further work is needed to understand how users might respond to a longer exposure to haptic feedback on their feet.

While we focused on understanding how directed haptic feedback affects the user's perception of the remote environment, there was less focus on methods of controlling the robot, but it would be useful to explore this in future work. For example, it might be worthwhile to investigate how to get both feedback and control on its feet. There are several ways of doing this, for example, controlling the robot based on leaning or based on walking-like metaphors.

5.3 Combining Haptic Modalities using a Foot-Platform in VR

In the previous section (see Section 5.2), a system was presented that explored the effect of haptic feedback on the feet when controlling a telepresence robot. It was shown that under certain conditions the user's presence can be increased. Furthermore, it was also found that haptic feedback depends strongly on context, especially when operating in realistic everyday situations. In order to continue addressing research question 2 (see RQ2), a further interface will be presented in this section, that also deals with the influence of haptic feedback on the feet. Similar to the previous prototype, the user will again be immersed in another environment, this time in a virtual environment, and it will again be examined how perception can be changed or even intensified.

Natural walking, in comparison to reality, is often not possible when navigating in larger virtual environments and computer games. The interface and research presented in this section explores how conventional navigation methods can be augmented with haptic feedback to enhance factors such as self-motion sensation, also referred to as vection. More specifically, gamepad-based and leaning-based navigation metaphors will be studied. For this purpose a custom-designed foot haptics prototype was designed and developed. Several multimodal stimuli will be presented to explore cues related to walking, reaching from vibrotactile cues (via vibrotactile transducers and bass-shakers under participants' feet) to auditory cues (footstep sounds) to visual cues (simulating bobbing head-motions from walking).

Although gamepads are popular interfaces for navigation in virtual reality (VR), they hardly provide the same self-motion cues associated with walking in real world environments. The ability to move freely in space while wearing a head-mounted display would offer appropriate physical cues of motion, however, it is often not feasible because of limited space, safety, cost, technical complexity or fatigue from long-term use. In contrast, there are leaning-based interfaces, some implementations are based on the Wii Balance Board [VSBH10], [JL12], recognizing the weight shift and using it to navigate. Some other approaches are based on optical methods to detect movements in the upper body or even the whole body [GPI⁺15], [MPL11]. However, regardless of the technical implementation, both approaches allow for long-distance movements and navigation without the above mentioned limitations, such as limited space. Compared to gamepad interfaces where the human body remains primarily passive and vestibular and proprioceptive cues are missing. Leaning-based interfaces can improve navigation performance [HNW⁺14] and provide a more immersive and embodied experience, as they allow at least some vestibular cues, which can improve self-movement perception (vection) [KRTK15], [RSP13].

A perfect locomotion interface should reflect all conditions, such as vestibular and proprioceptive cues from walking, air movements of our ears and haptic and auditory signals of our feet touching the ground, just like in reality. While the aim of the present interface was not to develop a perfect interface, it allows a number of questions to be examined. Specifically, the focus was on the effect of haptic feedback on self-motion and the overall experience, which is also important for answering research question 2. It has previously been shown that visual induced perception of illusory self-motion can be enhanced by appropriate auditory (e.g. dynamic sound fields) and vibration cues [LR14]. However, the impact of these cues on active locomotion conditions will be assessed in the following. Therefore, a comparison between a seated user navigating with a gamepad and a user navigating using a standing, leaning-based interface is investigated.

In order to get a more detail understanding, a prototype that provides fine-grained multisensory cues for standing leaning-based interfaces, extending work by Marchal et al. [MCV⁺13] and Feng et al. [FDL16] who focused on supporting seated users. The interface provides audio-visual cues as well as foot-based stimuli ("foot haptics") that partly substitute real-world cues. The focus of the proposed system is on "foot haptics", which are implemented through a dense grid of vibration motors, a bass shaker and a loudspeaker under each foot. The cues are physically co-located, similar to cues that are perceived when walking in the real world. It will enable to add various auditory, visual and vibrotactile cues related to walking, which could affect both standard joystick navigation (seated users) and leaning-based navigation (standing users).

5.3.1 Implementation

According to RQ2, the objective is to explore how foot haptics can improve the perception of self-motion and performance in leaning-based as well as gamepad-based navigation. For this purpose a prototype has been designed to augment virtual scenarios with foot-based feedback. In the next paragraphs a detailed overview of the implementation of the interface will be outlined, followed by a report describing the study conducted and the related results.



Figure 5.12: Feedback to each foot consists of A) a loudspeaker mounted in a solid case, B) a bass-shaker, C) eight vibration motors included in a flip-flop and D) the core frame of a Wii balance board.



Figure 5.13: Foot pressure distribution zones of a standing user showing the vibrotactors locations, and the five phases during a half gait cycle.

System

The haptic interface (see Figure 5.12) consists of four feedback components that are placed under the feet. All components are set in a wooden housing made of medium density fibreboard. In addition, a Wii Balance Board has been mounted under the housing. The feedback is provided by various actuators (a speaker mounted in a speaker housing, a bass shaker and a grid of vibration motors). Moreover, walking like visual cues are provided by a head-mounted display connected to the system, an Oculus Rift DK2.

The overall system (see Figure 5.12) consists of a feedback components mounted underneath the feet, placed on top of the Wii balance board load sensors and electronics in a wooden (medium-density fibreboard) case. Cues are provided by actuators mounted underneath each foot: a loudspeaker mounted in a speaker case, a bass-shaker, and a grid of vibration motors. Furthermore, additional visual cues are provided through a head-mounted display connected to the system, an Oculus Rift DK2. To ensure that the feedback given to one foot is barely perceptible to the other, the two boxes have been separated by 1 cm thick insulation of solid foam. The aim of this prototype is to provide walking-like cues, including visual (head vibrations), acoustic (footstep noises) and vibration cues (rolling pressure and ground impact). The entire system runs in real time on a graphics workstation implemented in Unity3D. All vibration motors are controlled by two Arduino Megas, while the loudspeaker and bass shaker are driven by two amplifiers. **Navigation** As mentioned in the introduction, two navigation metaphors are supported. Within the seated version, navigation is done using a Microsoft X-Box controller. In the standing version, navigation is performed by leaning the upper body. A Wii-Balance board allows to determine the centre of pressure, which in turn can be used to navigate in virtual environments. The implementation supports forward, backward, and sideways motions, as well as turning during forward motion.

Visual Head Movement In line with the head bobbing concept of Grossman et al. [GLA⁺88], a proprietary head bobbing algorithm has been implemented in Unity3D to simulate the horizontal and vertical oscillating movement of the head during real walking. In doing so, the translation is described by two sine waves, one representing the movement on the X-axis and the other the movement on the Y-axis. To simulate typical head movement, the vertical wave must have approximately twice the frequency of the horizontal wave. Followed an iterative design and pre-tests were used to fine-tune the frequency and amplitude for the up and down movement of the head synchronously with the speed of the user's movement.

Audio Realistic walking sounds were used as a source for the audio cues. The orgin of the sounds was chosen at the feet, as in real life, similar to $[PFC^+10]$. Therefore, two loudspeakers (Visaton FR10 20W) were installed in the aforementioned wooden housing; both loudspeakers were powered by an amplifier (Samson Servo 200). Walking sounds are defined by two main characteristics: speed and surface texture. Based on the speed of movement, the walking sound duration is reduced or increased so that it corresponds to the walking phase duration in which the feet touch the ground and the associated airborne phases in which the feet do not touch the ground. In this way, the sounds are kept synchronised with the ground contact phases influenced by the walking speed, starting from the moment the heel hits the surface. A solid (wood) and aggregates (gravel) were selected, as we assumed that both can be distinguished quite well [GMV⁺08] and represent typical surface textures.

Vibrotactile Motivated by previous research that used plantar vibrations to improve self-motion perception [FDL16], [TMM⁺13] and navigation cues [VBV⁺12], a vibration approach and system has been developed that stimulates different parts of the foot soles. The aim was to reproduce the sensation of a rolling movement of the foot as best as possible. To do this, the interface uses eight vibrating actuators per foot (see Figure 5.13): seven below the metatarsus and toes (Precision Microdrives: Model No. 304-116; 5 mm;15 000 rpm) and one below the heel (Precision Microdrives: Model No. 307-103; 9 mm; 13 800 rpm). All vibration motors are positioned so that users with different foot sizes can still perceive the stimuli. All vibration motors are glued to a rubber surface which is stretched over a solid foam sole in which small holes are drilled to accommodate the vibration motors. In this way, each vibration motor can vibrate properly against the foot sole instead of absorbing and damping a heavy load when the users stand on it. This means that the feedback provides similar results under different conditions, as postures (such as standing or sitting) affect the pressure on the soles.

A bass shaker (Visaton EX-60) is used to amplify the heel impact on the floor. The bass shaker is mounted on the foot support platform underneath the heel and stimulates this part of the foot the most. Although the current design cannot isolate the vibrations in order to solely impact on the heel, there is a noticeable drop in power towards other parts of the foot, similar to the effect of ground impact when walking in real life. The vibration motors and the bass shaker are synchronised to simulate the different phases of walking. The gait pattern can be defined by stride phases and length, as well as by differences in pressure and duration of the foot sole impact (the "ground contact phase") experienced during the different phases of the foot rolling in a gait phase.

Each vibration motor can be controlled individually (see Figure 5.13), thus allowing the simulation of walking, slow or fast running, which in turn is again defined by the ground contact phase and its duration. When normal walking, either the left or the right foot is always stimulated, with some overlap. With increasing speed and a changed swing phase associated with the movement of the limbs the phases in which none of the feet receive (tactile) stimuli also increases. These phases coincide with an increase in stride length, frequency and ground contact. A basic vibration was used for the different surface textures. The sound also becomes louder when speed increased. This is achieved by increasing the basic volume: walking (10%), slow (25%) and fast running (40%).

5.3.2 Experiments

In order to test our hypotheses and to find further answers to our research question, we conducted five sub studies in a single experiment. The aim was also to create a better understanding of the potential approaches and interaction of our implemented cues. Besides the post-questionnaire (see Questionnaire 6), which deals with overarching questions, there are five hypotheses, one per session. In the hypotheses the general term perception is used, which stands for the four specific perception aspects queried in Questionnaire 5. It is hypotheses that:

- H1: Surface textures are perceived more strongly when using a leaning-based surface.
- H2: Audio and vibrations affect perception the most.
- H3: The perception of cue combinations differs in the navigation metaphors.
- H4: Minimal walking in place will positively influence perception.
- H5: Adding all haptic cues increases perception while navigating freely.

Participants

Twelve participants (25-48 years old, mean age 29, one female) have been recruited to participate in the experiment. Participants were recruited either through a mailing list or randomly on campus. Seven participants reported they played games daily or weekly, and the rest less frequently. All users had normal or corrected-to-normal vision. On average, the whole study took about one hour to complete. As in the previous studies, users received a briefing and a debriefing and participants were informed that they could quit at any time without consequences. Participation was voluntary, and no money was paid for participation.



Figure 5.14: User in standing leaning (left) and seated (middle) pose. Test environment with follow-me object (right).

Apparatus

The following five sessions were performed with the prototype introduced earlier. Participants were seated during all gamepad conditions and stood in the leaning conditions (see Figure 5.14). In the following either the gamepad or the leaning condition will be mentioned, assuming that the leaning condition is performed standing while the gamepad

is performed seated. The virtual camera was placed at the same height in both conditions, independent of the conditions. This ensured that visual perceptions were as identical as possible over all settings. When answering the questions, participants were asked to sit down again.

Design and Procedure

The experiment consisted of five individual sessions, all participants had to attend all sessions (within-subjects design). A diagrammatically overview can been seen in Figure 5.15. All sessions were performed in the same order. The focus of the study was more on participant perception and user experience than on task performance. Thus, even if there was a transfer of learning between sessions, this should not critically affect the observed results.

After each trial in each session participants were asked questions about self-motion perception. The in-study questionnaire (see Questionnaire 5) was based on an 11 point Likert scale (0-10), whereas 10 points were total agreement and 0 total disagreement. Users to directly answer, after each trial, the following four questions. All questions were displayed in the HMD and were rated orally while the experimenter note the answers.

- "I had a strong sensation of self-motion" (vection)
- "I could judge velocity travelled well" (velocity)
- "I could judge distance travelled well" (distance)
- "I was not aware of my real environment" (*involvement*)

After each session, the HMD was removed and participants were asked to answer questions, which were displayed in a form on a computer screen. The questions included user comfort, ability to focus on the task, perceived navigation performance, ease of learning, fun, ability to use the interface for a longer time, intensity of the vection and ease of use Questionnaire 6. Additionally, after the first session, the participants scored the degree of confidence in walking on the idifferent surface textures. The participants were allowed to take a short break and to remove the HMD whenever motion sickness was an issue. Before and after, the total experiment participants reported whether they were fresh and relaxed, and the degree of their motion sickness. Finally, it was asked which condition (seated or standing) had a more positive influence on self-motion.

In all sessions participants were asked to navigate along a clearly visible curved trail. Six paths were created through a natural environment populated with trees at the side of each



Figure 5.15: A diagrammatic view on the overall study as well as an overview of the individual sessions and their analysis.

trail (see Figure 5.14, right). All the trails had a similar curvature profile, and the trees were selected to provide some motion-cues in the peripheral visual field. To guarantee similar speed profiles despite active navigation, participants in sessions 1-3 had to follow a marker (a clearly visible blue ball, see Figure 5.14, right) that moved in front of them. The marker was moved at different speeds, starting with walking, followed by slow/fast or fast/slow running, changing every three seconds. The speeds were chosen to correspond to the profiles of walking, slow/fast walking and fast/fast running. Each trial lasted about 10 seconds. Pilot studies showed that 10 seconds was sufficient to experience vection and to learn and evaluate the different interfaces and cue combinations. In sessions 4 and 5 the participants were asked to move freely on the path, so that no marker was displayed.

As many users may not know what self-motion perception exactly means, a calibration phase was performed before the actual experiment, which all participants have to took. During the calibration users have to lean forward and navigate through a star field simulation which provides a strong self-motion perception. This experience should be used as reference for the evaluation of vection intensity. The study began immediately after the calibration. A description of the five sessions is provided in the next paragraphs, which are also illustrated diagrammatically in Figure 5.15:

Session 1: How well can multisensory cues be associated with different surface textures?

Description: Participants were asked to practice both navigation interface by following a marker (blue sphere). All cues (tactile, bass shaker, audio and head bobbing) were activated. The path was divided into two zones (wooden or gravel ground). Participants had to travel in each zone twice, resulting in two trials in total.

Session 2: How do audio-tactile cue combinations affect vection while leaning?

Description: The focus in this session is on cues combinations and the effect on self-motion perception and involvement using our leaning interface. Therefore a 2 x 2 x 2 factorial design was applied; each participant completed two repetitions, resulting in a total of 16 trials. The independent variables are tactile (on/off), audio (on/off) and bass-shaker (on/off). The repetitions were blocked, meaning that all cue conditions were completed before they were repeated. Head bobbing was on in all conditions.

Session 3: What influence do multisensory cues have on vection when comparing gamepad and leaning?

Description: In this session research focused on perception of self-motion and involvement of the implemented cues when comparing the leaning and gamepad interface. The study was based on a 2x2x3 factorial design; the independent variables were navigation interface (leaning/gamepad), head bobbing (on/off) and foot-haptics (off/audio/all). This results in a total of 24 trials with two repetitions, again blocked.

Session 4: What effect has minimal walking in place on leaning locomotion?

Description: The potential of adding a minimal walking in place (WIP) metaphor to the leaning interface is explored in this session. To ensure that participants received complete feedback during the study, all participants were advised not to move their feet during minimal WIP, the heel should rest on the platform, and users were asked only to pretend walking. Participants could freely navigate along the predefined path using the leaning interface either with or without added minimal WIP, repeated twice, totalling four trials. The independent variable was WIP (on/off). We allowed to practice WIP once before starting the actual trials. Participants were encouraged to synchronize their WIP with the foot haptics.

Session 5: Does foot haptics affect perception during free locomotion?

Description: Finally, the effect of foot-haptics while free navigate was studied. Participants were allowed to move freely along a path and were instructed to vary their speed dynamically. By doing so, we aimed to to explore the relationship between feedback and speed. Thus there was an independently variable: feedback (on/off). In this case, either all components were active or no feedback was provided. With one repetition there were four runs per participant. Although the exploration time in sessions 4 and 5 was not limited, the participants generally did not take more than 20 seconds.

Results

According the data from the post-questionnaire (see Chapter A) it can be concluded that in almost all aspects the common gamepad interaction was preferred by the users. Reviewing the data from all five sessions and comparing the three interfaces (gamepad, leaning and walking in place), a clear ranking of the three interfaces becomes noticeable (see Table 5.1). The gamepad interface has achieved the highest score, followed by the leaning-based interface and at least the walking in place interface. Interestingly, with regard to self-motion, the leaning-based interface received a considerably higher rating than the gamepad interaction. In general it can also be said that the gamepad and leaning-based interface was very positively received by the participants. The walking in place metaphor, on the other hand, did rather poorly. In terms of motion sickness, it was in average quite low, the average motion sickness scores received M = 3.92/SD = 2.64, only four of the 12 participants experiencing high levels (> 5). In the following paragraphs the results of the individual sessions are presented in detail.

	Interface		
	gamepad	leaning	walking in place
comfort	M = 9.0/SD = 1.6	M = 7.3/SD = 1.7	M = 6.1/SD = 2.2
focus	M = 8.9/SD = 1.3	M = 7.8/SD = 1.7	M = 3.6/SD = 2.7
navigate	M = 9.0/SD = 1.2	M = 7.6/SD = 1.5	M = 3.6/SD = 2.8
learn	M = 9.2/SD = 1.2	M = 7.8/SD = 1.5	M = 3.4/SD = 2.9
fun	M = 8.3/SD = 1.2	M = 8.1/SD = 1.8	M = 3.8/SD = 2.6
long-time	M = 8.2/SD = 1.5	M = 6.8/SD = 2.4	M = 3.7/SD = 2.5
self-motion	M = 7.7/SD = 1.3	M = 8.1/SD = 1.4	M = 4.0/SD = 2.8
usability	M = 8.5/SD = 1.6	M = 7.4/SD = 1.7	M = 3.2/SD = 3.0

Table 5.1: Overall ratings of the studied interfaces (likert scale 0-10); complete questions can be looked up in Questionnaire 6.

Session 1 offered the possibility to become familiar with the introduced interfaces. In addition, the perceptibility and influence of two surface textures (wood and gravel) was explored. Data from the in-study questionnaire (see Questionnaire 5) were we looked into self-motion, judgement of the travelled velocity and distance and also at the level of involvement showed that independent of the surface texture the level of involvement was significant higher when navigating with the leaning-based interface (gamepad M = 7.12/SD = 1.63/leaning M = 7.76/SD = 1.24; Z = 28; p < 0.038) (see Figure 5.16). Thus, the hypothesis (H1) for the first three factors (vection, velocity and distance) must be rejected. For the factor involvement, the hypothesis is correct. It can be assumed that the surface textures are perceived more strongly in the standing, leaning-based interface, and thus the sensation of involvement is stronger.



Figure 5.16: In-study questionnaire showed an significant effect of involvement, highlighted in yellow.

The post questionnaire (see Questionnaire 6) showed that participants found the feedback from both surface textures more convincing when using the leaning-based interface (wood M = 7.08/SD = 1.62; gravel M = 6.42/SD = 1.89) instead of the gamepad navigation (wood M = 6.75/SD = 1.06; gravel M = 5.92/SD = 1.80) (see Figure 5.17, 2nd row). In terms of the quality of feedback, users felt that walking on wood was received better than walking on gravel.



Figure 5.17: Users could better concentrate on the task and better navigate. Users also scored the learnability of the gamepad interface higher compared to the leaning-based interface. Sensation of surface textures received higher scores while standing.

When looking at the data from the post questionnaire it becomes visible that in this session the interface has a significant effect on focus, navigation and learnability (see Figure 5.17, 1nd row). This means that the users were able to focus better on the task when using a gamepad (gamepad M = 9.08/SD = 1.38/leaning M = 7.92/SD = 2.23; Z = 0; p =0.034). Furthermore, the users also found the navigation using a gamepad significantly better compared to the leaning-based interface (gamepad M = 8.92/SD = 1.38/leaning M = 6.92/SD = 1.62; Z = 0; p = 0.006). Finally, the learnability of the gamepad interface has also convinced the participants (gamepad M = 9.08/SD = 1.38/leaning M = 7.67/SD = 1.50; Z = 2; p = 0.017). A significant influence on the remaining factors of the questionnaire (comfort, fun, long-time usage, self-motion and usability) could not be determined. However in average the users had slightly more fun with the leaning-based interface (gamepad M = 8.42/SD = 1.16/leaning M = 8.83/SD = 1.47) and also a higher self-motion perception (gamepad M = 7.75/SD = 1.71/leaning M = 8.42/SD = 1.38). Interestingly, participants found the leaning-based interface on average better for long-term usage (gamepad M = 8.17/SD = 1.19/leaning M = 8.33/SD = 1.61). It is probably not surprising that in many cases the gamepad variant was rated better by the users, as most of them already know this kind of interface very well.



Figure 5.18: Audio and vibration are the most relevant parameters with regard to the perceptual factors studied.

In session 2 the influence of the implemented cue combinations on perception was studied. The focus was on perceptual factors such as the sensation of self-motion, the perceived speed and distance and the level of involvement. According to Figure 5.15 the emphasis was on the leaning-based interface. The visual feedback (head-bobbing) was always enabled. All results of the in-study questionnaire are also depicted in Figure 5.18. It turns out that audio and vibration feedback are the driving factors. A 3-way ANOVA with repeated measures showed significant effect on self-motion perception when either adding audio (F(1, 11) = 43.78, p < 0.001) or vibration (F(1, 11) = 5.01, p = 0.026) is

enabled. Participants could further judge travelled velocity better when either adding audio (F(1,11) = 20.14, p < 0.001) or vibration (F(1,11) = 5.31, p = 0.022) is enabled. The vibration feedback has no significant influence on the participants' judgement of the distance travelled (on: M = 8.01/SD = 1.28/ off:M = 7.71/SD = 1.30) and their presence (on: M = 7.34/SD = 1.14/ off:M = 7.12/SD = 1.24). However, audio feedback underneath the feet can significantly affect the perception of distance (F(1,11) = 13.72, p < 0.001) and presence (F(1,11) = 26.77, p < 0.001). H2 can be accepted, vibrations and audio are the dominating factors. Surprisingly, the feedback of the bass-shaker has no significant effect. Different combinations, such as audio and vibration, did not have an influence on perception. Audio seems to be more effective, as it achieved a significant result for all factors studied, while the vibration feedback only increases the perception of self-motion and velocity.



Figure 5.19: A significant effect was found for all variables when comparing the two navigation metaphors.

In session 3 the two interfaces were compared. For this purpose different feedback modalities were turned on and off. In this way, the influence of the modalities (audio, haptics and head-bobbing) on our perception parameters could be measured. Looking at the ratings of the in-study questionnaire, all three conditions have a significant effect on self-motion perception. A 3-way ANOVA with repeated measures found that the leaning-based interface (F(1,11) = 4.19, p = 0,043) has a significant impact on vection among all participants. But also feedback (F(2,11) = 14.42, p < 0.001) and head-bobbing (F(1,11) = 5.39, p < 0,022) can significantly influence the perception of self-movement (see Figure 5.19). With regard to our hypothesis (H3), a pairwise comparison of the two interfaces and the remaining three perceptual parameters (velocity, distance and involvement) shows a significant effect. A Wilcoxon signed-ranks test resulted in velocity (Z = 447.0, p = 0.027), distance (Z = 370.0, p = 0.024), and involvement (Z = 236.0, p = 0.031).

Further results show a similar trend as in the previous two sessions, haptic feedback below the feet seems to be the most important component. It was found that participants could significantly better judge velocity (F(2, 11) = 8.43, p < 0.001) and distance (F(2, 11) =6.48, p = 0.002) with haptic feedback. Furthermore, the feeling of presence can significantly improve through haptic feedback (F(2, 11) = 9.44, p < 0.001). Looking at the findings of the post questionnaire comparing both interfaces, it becomes clear that the gamepad solution is significantly preferred by the users. The ratings showed higher scores for the gamepad versus leaning-based interface for comfort, concentration, ease of navigation, learnability, prolonged usage, and overall usability (see Figure 5.20). While all ratings for the leaning interface were fairly high, there is a clear need for improvement to bring them close to the gamepad level. This of course also influences the hypothesis H3 and would mean that if the quality of the leaning-based interface is increased, H3 can be better tested.



Figure 5.20: Gamepad interaction was clearly preferred by users.

Session 4 dealt with the comparison of walking in place in combination with the leaningbased interface and with the pure leaning-based interface. Preliminary tests suggested that the addition of a walking-like metaphor could increase perception factors like self-motion perception. Unfortunately, there were some problems with the implementation and the realisation of the metaphor. The walking in place metaphor made it difficult to navigate by shifting the weight, furthermore the synchronization of the stimuli had to match with the steps performed by the participants, which also proved to be difficult. All except two participants had problems to complete the task correctly. This in turn led to poor results for the WIP condition, which scored significantly lower than the leaning-based interface in both the in-study (see Figure B.2) and the post-study questionnaire (see Figure B.3). The results of the two questionnaires can be found in the appendix. All factors, self-motion perception, judgement of velocity and distance as well as the level of involvement were worse in the WIP condition. Since the results from the post questionnaire were just as bad, the results from session 4 cannot be taken into account. To conclude, the idea of implementing WIP with the presented interface must be further developed, and thus, in this case, it makes no sense to explore our hypothesis (H4) further.



Figure 5.21: When enable all haptic cues, perception will increase for all studied parameters.

In the last session (session 5) the impact of foot haptics in free exploration was studied by using either no foot haptics or full foot haptics. The focus was again on the leaningbased interface, and the trend was further confirmed - there was a positive effect of foot haptics on perception. In detail, this means that adding combined haptics (bass shaker, audio and vibration) to the feet can significantly increase self-motion perception (haptics off M = 6.54/SD = 1.66/on M = 8.54/SD = 0.95; t(11) = -3.88, p = 0.003), the estimation of velocity (haptics off M = 6.75/SD = 1.30/on M = 8.04/SD =1.22; t(11) = -2.95, p = 0.013) and distance (haptics off M = 7.21/SD = 1.46/onM = 8.33/SD = 1.23; t(11) = -3.0, p = 0.012), and the level of involvement (haptics off M = 6.29/SD = 0.59/on M = 7.88/SD = 1.21; t(11) = -4.42, p = 0.001) (see Figure 5.21). In general, the ratings were slightly higher than in the previous sessions. This is consistent with the participants' statement in the post-session interview that they were better able to focus on the effect of the cues when they did not have a main task, for example, to follow a moving object. In summary, hypothesis (H5) can be accepted. The combination of all haptic cues significantly increases all examined perception parameters. In a concluding step, data from the different sessions were analysed in order to draw further conclusions. The focus was on completion time and the influence of different stimuli on self-motion perception. Although users were asked to follow a blue sphere, it was interesting to see if there was a difference in completion time. For this purpose, the participants' time spent per trial was logged. Another interest was the influence of the four stimuli (vibration, audio, bass shaker and head bobbing) on self-motion, which was asked in the second part of the post-questionnaire. Looking at the first sessions in which the users had to follow a marker, and considering the influence of the interface on completion time, a one-sided hypothesis test shows that when using the gamepad interface (M = 48.96 s/SD = 8.14 s) the participants completed the courses significantly faster than when using the leaning-based variant (M = 52.39 s/SD = 7.57 s; W = 54.0, p = 0.028). The influence of interface on completion time was confirmed again in session 3, where participants were significantly faster with the gamepad (M = 44.05 s/SD = 9.30 s) than with the leaning-based interface (M = 53.78 s/SD = 16.63 s; W = 1440, p < 0.001).

The influence of cue combination on perceptual parameters was investigated in session three. When analysing the completion time in this session, audio (on M = 49.82 s/SD = $15.02 \text{ s}/\text{off} \ M = 49.95 \text{ s}/SD = 16.54 \text{ s}$) and vibration (on M = 51.00 s/SD = 16.83 s/offM = 48.77 s/SD = 14.60 s) did not show significant differences. Surprisingly, with the assumption that the paths could be completed faster with feedback, the participants completed the tasks significantly slower with bass-shaker feedback (on M = 51.40 s/SD = $17.14 \text{ s}/\text{off} \ M = 48.37 \text{ s}/SD = 14.16 \text{ s}; \ W = 1760.0, p = 0.038$). The main results of the completion time are also shown in Figure 5.22.



Figure 5.22: Main results of the first three sessions in terms of completion time. The graph shows that in sessions 1 and 3 the interface had a significant influence on the completion time. Session 2 showed that users with an active bass shaker were noticeably slower.

As mentioned at the beginning, the impact of the individual feedback components, independent of the task and interface, on self-motion perception needs to be studied. For this purpose, the answers from all sessions were summarised. A pairwise comparison of the four feedback modalities showed that audio (M = 8.43/SD = 1.97) and vibration (M = 8.28/SD = 1.87) feedback clearly improve the feeling of self-motion compared to bass-shaker (M = 7.94/SD = 2.10) and head-bobbing (M = 7.84/SD = 2.48), Figure 5.23 illustrates these results. Finally, the recognition of the stimuli was queried, and the users' ratings showed that all feedback modalities could be easily distinguished from the others (audio: M = 8.83/SD = 1.52; vibration: M = 8.50/SD = 1.55; bass-shaker: M = 8.83/SD = 1.95; head-bobbing: M = 8.17/SD = 1.95)

Overall - effect of feedback modality on vection



Figure 5.23: Effect of our four feedback components on vection over all sessions. The figure shows that audio and vibration feedback have a stronger influence on vection.

5.3.3 Discussion

Focused on RQ2 and our hypotheses we have we presented a prototype combining leaningbased navigation with a foot-haptics mechanisms. Results showed that both self-motion perception (vection) and involvement/presence could be significantly enhanced by adding walking-related vibrotactile cues (via vibrotactile transducers and bass-shakers under participants' feet), auditory cues (footstep sounds), as well as visual cues (simulating bobbing head-motions from walking). Moreover, participants' self-reported ability to judge self-motion velocities and distances travelled was enhanced by adding footstep sounds and vibrotactile cues. Interestingly, all these observed benefits of adding walking-related cues occurred independently of whether participants controlled self-motion via joystick while seated or via leaning while standing. This suggests a more general benefit of adding walking-related cues that might generalize to further locomotion paradigms and interfaces, with many potential application areas.

Together, the outcomes support the assumption that haptic and proprioceptive cues experienced during natural walking can at least to some degree be substituted for by other feedback channels such as vibrotactile feedback, and can be further supported by audio-visual cues. This outcome is in line with previous studies, such as the system and study by Terziman et al. [TMM⁺12], showing similar effects for seated users.

A key finding of ours is that leaning while standing improved self-motion perception significantly compared to seated users using a joystick, even though participants had extensive experience using joysticks but no experience using leaning-based interfaces. This extends prior work showing that passive (but not active) seated leaning on a manual gaming chair could enhance self-motion sensations [RF12].

5.3.4 Conclusion

Motion sickness was an issue for some users. While this might at least in part be attributed to the long duration of the experiment inside a head-mounted display, as even with breaks it took around one hour, further research is needed to investigate which factors might have contributed and how motion sickness could be reduced. Because of the markerfollowing procedure, we could only ask participants to introspectively rate their ability to judge velocities and distanced travelled. Future work is planned to investigate if this self-assessment also translates to improved behavioural measures of distance/velocity and more complex navigation behaviour. Pilot data suggests that seated leaning can indeed reduce distance underestimation for VR locomotion. However, the current results suggest that compared to seated joystick usage, standing leaning interfaces, in particular when combined with minimal WIP might require additional cognitive/attentional resources, and would benefit from additional practice and further interface improvements.

In the future, we intend to extend the base system by looking into the potential of including limited haptic feedback to the feet, for example to provide collision feedback. We are also interested in the addition of other motion cues, such as wind and barely perceptible wind sounds that occur when someone is moving through the physical world. Furthermore, we will investigate how we can generalize the system to better include rotations, for example by using torso-directed locomotion [BKH98].

Despite the need for further system improvements, the current results already highlight the potential of sensory substitution and incorporating walking-related auditory, visual, and vibrotactile cues for improving user experience and self-motion perception in applications ranging from virtual reality and gaming to telepresence and architectural walk-throughs.

5.4 Summary

Both prototypes presented showed in several ways how perceptual aspects could be improved in virtual and remote environments. For this purpose, both approaches used the feet for haptic feedback, once the sole and once the sides of the feet. The objectives and sub-question should help to answer RQ2. Objective 4 was successfully implemented with the fourth prototype. A haptic foot platform was presented that maps spatial data of a remote environment captured by a telepresence robot to increase spatial awareness. To answer the sub-question (RQ2.1: How can foot haptics support users awareness when interacting in remote environments?), two separate studies were conducted. While the first study was conducted in a simulated virtual environment, the second study was conducted in a real environment with real tasks. A quantitative evaluation showed that in the first study, the added feedback could significantly increase spatial awareness. The users moved more carefully with haptic feedback than without. Unfortunately, this did not apply to our second study with real tasks. In summary, we can say that the feedback approach is promising and we see great potential in it. However, it still needs further development for real tasks, especially with regard to the adaptation of different environments. The algorithm for calculating haptic feedback must be designed differently for open spaces or crowded places to exploit the haptic feedback in the best possible way. Nevertheless, it could be shown that the feedback approach, to implement proximity cues with vibrations and collision cues with light impacts, works.



Figure 5.24: Sketch of both implemented foot haptics approaches. It can be seen that different locations of the foot have been stimulated (blue). While vibrations are supported by both prototypes, collisions are represented with a slight impact and walking-like cues augmented with audio feedback (yellow).

With regard to our last, the fifth prototype, and the associated sub-question (R 2.2), how can feet-based haptic feedback be designed to enhance the self-motion perception in virtual environments? it can be concluded that perceptual parameters such as self-motion perception could be increased. In detail, we were able to achieve further results beyond the confirmation of the improvement of perception factors [TBS13], [WTC⁺20] already known in literature. As the study was highly exploratory, five different sessions were conducted to investigate various combinations of conditions. This included the combination of different haptic feedback modalities as well as combinations of our two interaction metaphors. Overall, the haptic feedback provided to the foot was perceived more intensively in the standing version than in the sitting version. In the analysis, we examined four main factors: self-motion perception, the judgement of the distance and speed travelled, and the users' involvement. Furthermore and in general, audio and vibration cues have in most conditions displayed a significantly higher effect on the four perceptual aspects. Objective 5 has been achieved, and we have introduced a foot platform that allows us to improve self-awareness, but also to improve other parameters, for example, the involvement of the users.

To conclude, Figure 5.24 shows, as we did in the previous chapter, the two basic ideas of the implemented approaches. Two prototypes were developed that can provide haptic feedback in different ways to the feet. These two designs also demonstrated how spatial awareness and self-motion perception could be increased, and contributed significantly to answering research question RQ2. In summary, it can be said that in our studies the haptic feedback achieved better effects while standing. Furthermore, the users were satisfied with the feedback on the feet, it was only perceived as uncomfortable when there was too much. In the next chapter, we want to summarise the main chapters, highlight the main contributions, but also the limitations, and conclude with ideas for future work.

Chapter 6

Conclusion

There is no doubt that the visual and auditory sensory channels are currently the most essential and efficient interface between humans and computer $[OVV^+16]$. Rendering physically correct images in real-time and generating realistic sounds is still a challenge. However, compared to our other senses, i.e. the sense of touch, smell and taste, research is more advanced in the context of human-computer interaction. Nonetheless, when considering the possibilities of haptic feedback, it has already been proven that a mouse or a keyboard or, as has become increasingly popular in recent decades, small touchscreens have a considerable impact on human-computer interaction $[OSC^+17]$. In this thesis, we have explored further potentials, which we would like to summarize in this chapter.

6.1 Summary

Since the domain of multimedia, mulsemedia, multimodal and multisensory interaction and especially haptic feedback (see Chapter 1) is quite broad [Sut03], [GTLG14], we have narrowed our focus to two areas. These areas have been used to show concepts and approaches to provide haptic feedback. In doing so, we looked at two key aspects - in particular: performance and perception, as we consider them essential and dependent. In short, often perceptual methods are required to adjust or even improve performance parameters, and when users interact efficiently with a computer, the user experience will increase.

Two focus areas are reflected in the overall structure of the thesis. Thus, the literature review (see Chapter 2), as well as our two main chapters, were divided into these areas. In the methodology chapter (see Chapter 3), we discussed different directions and methods that are important in order to address our two research questions mentioned in Chapter 2 and to achieve our objectives as specified in Section 1.2. Since all our research questions tend to suggest an empirical and artefact contribution, we introduced individual approaches, that look into the design and development of new prototypes (artefact contributions), or into the evaluation of such prototypes (empirical contributions) [WI16].

Five prototypes from two domains were presented. Each of the prototypes was evaluated with at least one study, and several specific applications were implemented and presented as well. While the two main chapters deal in detail with the combination of haptic feedback with touchscreen interaction [FS01], virtual [WTC⁺20] and remote environments [NVPH16], the whole thesis has to be seen as a framework that makes both topics part of coherent research. For each of these five interfaces, we have shown that there is potential for haptic feedback to improve performance, perception and user experience.

In Chapter 4, we proposed three unique methods through which haptic feedback can augment touchscreen interaction to show versatility. Our first prototype was ForceTab, an actuated touchscreen driven by a small motion-platform to provide force feedback to the fingers on the screen. The combination of flexible haptic feedback above and rigid feedback on a touchscreen was explored with our FleXurface prototype. With our third prototype, we investigated performance and self-perception aspects using back-of-device techniques to provide a pin-based stimulus to the finger. For all three prototypes, a perception-based study was performed to explore the possibilities and limitation of haptic feedback applied to touchscreens. The dependence of perception and performance became clear in this chapter. For example, in Section 4.4 a perception study was first carried out and the results were used to increase the touch accuracy.

Telepresence robots and virtual environments are not without their challenges. Therefore, in the second main chapter (see Chapter 5), we presented two further prototypes, both for which address the sensory augmentation. All prototypes in this chapter, focuses on haptic feedback provided directly at the foot in order to increase self-motion perception (vection) in virtual environments and to enhance spatial awareness in telepresence scenarios. In both scenarios studied, the users are either in an artificial or in a remote space. In this case, the visual interface can be seen as a window to a remote space or as a camera attached to a virtual character. Both chapters and corresponding areas clearly show an overlap in the objectives and the way of achieving these objectives. The thesis aimed to explore and improve perception and performance aspects, which was achieved in touchscreen interaction and virtual and remote environments. Furthermore, the objectives were always achieved with the help of haptic feedback. For this purpose, the hands and feet were addressed. As could also be seen, the design of the interfaces was adapted to the respective body part. While we focused on more primary and concrete tasks within the finger-based prototypes, we provided more secondary, background feedback in our foot-based prototypes. However, both served in their individual way to support the user.

6.2 Contributions

In addition to the research questions raised in Chapter 2, two aims and five objectives were formulated in Section 1.2. In summary, three objectives cover the augmentation of touchscreens with haptic feedback and two objectives looking into the areas VE and RE. All of them helped to answer our two research questions. In detail, each of the main chapters answered a particular research question that served to achieve one or more of our five objectives. In the following, the main contributions are presented. In the summaries of the main chapters, it was already shown, that all objectives have been achieved.

However, before considering the contributions of the two main chapters, we would like to mention that the results would not have been achieved without the individual prototypes. Therefore, we consider the designed and developed hardware and software concepts as part of our contribution. Furthermore, in the course of this work, a procedure and methodology have progressively been established that allow us to develop and evaluate these kinds of novel interfaces. Our interfaces are mainly based on sensors and actuators that not only allow to interact with a user conventionally but also to stimulate a further sense in order to involve the user more realistically in the interaction. We have described this methodology in Chapter 3. As a result, the following contributions have been achieved with the introduced methodology.

The above leads to the key contributions of this thesis. We contribute in two areas, all focusing on haptic feedback to enhance interaction performance and perception. In Chapter 4, we explored ways on how to augment small and tab-sized touchscreens and thus addressed RQ1, which lead to the following contributions, parts can also be found in [MKHG17], [KBT⁺19] and [MEK⁺19]:

Contribution 1:

The limitations and possibilities of haptic feedback were explored by: (i) an actuated touchscreen, (ii) elastic layer above the screen and (iii) a pin-based approach mounted on the back of a touchscreen. ranging from force and pressure feedback, to stiffness and compliance sensations, to pin-based skin and finger stimulation. Using the designed haptic interfaces, it was possible to show the variety of haptic interaction capabilities; moreover, it was also shown how the implemented haptic interfaces could be used to enhance touchscreen interaction. While tactile feedback combined with touchscreens has been widely researched, the approaches introduced in this thesis has gone beyond that and has contributed by exploring novel approaches. It was also shown that the areas of perception and performance are very closely related. Perceptual aspects are essential for answering performance issues.

In the second area, we looked at how to improve perceptional aspects in virtual environments and in telepresence systems. With the main focus on RQ2, we have found the following contribution which parts were published [KMT⁺16] and [JMM⁺20]:

Contribution 2:

With the use of two further haptic prototypes, we contributed to explore and improve perceptual aspects. We were able to show that both self-motion perception and involvement in virtual environments can be significantly improved by adding walking-related haptic cues on the feet. In this context, it was shown that our leaning-based interface has a positive effect on perception. Furthermore, we studied novel cue combinations and showed how they affect perception. Similarly, within fifth novel approach, it has been contributed that proximity and collision feedback can lead users in telepresence scenarios to change their behaviour due to an increased sense of responsibility and to become more cautious when moving in remote environments. In a particular study, it was demonstrated that spatial awareness can be increased.

In summary, it can be said that all objectives have been obtained as the contributions listed above illustrate. We have presented haptic user interfaces for two in to areas that explored perception, performance or experience. Novel haptic feedback was implemented for touchscreens, with which we explored performance and perception limitations. We have shown the potential of feedback with the feet to increase spatial awareness in telepresence systems and to intensify immersion in virtual environments. In general terms, it has been shown that haptic feedback, when used thoughtfully, contributes positively to perceptual aspects and influences performance parameters in the both of the studied domains.

6.3 Limitations

In retrospect, there are a number of limitations for this research and creating haptic feedback has its challenges. Since five unique haptic interfaces have been developed in the context of this thesis, we would like to outline the limitations we have experienced in our studies. Firstly, it has been particularly difficult to calibrate and precisely measure the exact characteristics, such as vibration or pressure force, with our resources. Especially with built-in vibration motors or when interacting with elastic material, it was not very easy to ensure that every user gets the same feedback or stimulus. Although the data collected showed that users perceived the feedback similarly, there is no guarantee that it was precisely the same for all of them. Secondly, it was difficult, especially with haptic devices, to take into account all ergonomic aspects for all users. Take the HapticPhone as an example, a smartphone with three movable pins on the back: in the study, most users reported that the pins were easy to reach, and the interaction felt comfortable. However, since the pins were attached to fixed locations, it was difficult for users with smaller or very large fingers to imagine using the prototype over a longer period of time. This example also shows which individual characteristics have to be considered when designing haptic feedback. When comparing this individuality with the rendering of images or the generation of sound, it becomes clear that there is an essential difference between the considered media and the corresponding sensory channel.

Finally, the sampling and the request of self-assessments should be reflected. We have conducted eight studies with an average of 14 participants; which leads to approximate 4000 trials in total. On the one hand, the disadvantages of convenience sampling have already been discussed in the methodology and, on the other hand, increasing the number of samples would provide more accurate results [BACM11]. However, for exploratory research and to point towards new directions, our findings are still valid and can be generalized to a considerable degree to the sample population.

Besides, we have asked for a self-assessment in several studies - we would like to emphasize that there may be biases in such subjective surveys [Ban77]. The wow-effect of new technologies can also influence users to make spontaneous emotional decisions, which can also lead to biases. However, since the analysis of the questionnaires indicated a clear direction and the variance of the data was mostly small, this effect seems to have been rather small in our studies. Nonetheless, further work is needed to increase the validity and generalization potential of the research reported here.

6.4 Future Directions

We see great potential in an increased involvement of the user in the interaction with a computer, also referred to as human-centred design [OSC⁺17]. From our point of view, this means that communication must take place as naturally as possible. This also includes the integration of further sensory channels. We see particular potential in the fields: telepresence, virtual and augmented environments and human-robot interaction. To overcome some of our limitations and looking towards a human-centred design, we want to suggest ideas for future work in all three focus areas. It is obvious that we also have an ongoing process of refinement and development of our methodology, which will also be applied in our future work.

Limitations in haptic touchscreen interactions, but also in haptic interaction in general, means that it is challenging to generate physically correct haptic stimuli and usability and ergonomics depend very much on the particular individual. It would be interesting to use further materials to investigate which are well suited to provide versatile and realistic haptic feedback and exploration. This could also include the development of new actuators. In order to better stimulate the individual characteristics, such as finger lengths, with haptic feedback when augmenting touchscreens, we believe it would make sense to detach haptic technology from the device and design a small device, similar to a thimble, that would be able to provide tactile, force and temperature stimuli. The challenge with this question will be how to combine the various components in a small amount of space while maintaining haptic requirements. These small thimbles would then no longer be limited to interaction with touchscreens. They could be used in more versatile directions, such as in telepresence scenarios and virtual and augmented environments.

We see further directions in foot stimulation and robot control. As in the previous paragraph, a challenge could be miniaturising all haptic components further, possibly integrating them into a shoe - resulting in a haptic shoe that provides versatile haptic feedback. Such a shoe could be used in both scenarios, telepresence and virtual environments. Furthermore, it could also be connected with a touchscreen to explore similar interaction metaphors we did above. Our approach to perceive the environment via proximity and collision feedback on the foot could be continued, and we could focus on further aspects such as perception and performance. If one looks at the sensor ring developed in Chapter 5 for sensing the environment, the question arises how the environment can be better recognised and a kind of assistant driving, which adapts to the environment or could even perform specific tasks autonomously. Such a procedure could be learnt and adapted to the user over time to provide the user with personalised and effective control.

6.5 Final Words

With the ever-increasing number of technologies in our daily lives, we are more and more overwhelmed with different interfaces. Understanding complex processes and using interfaces correctly is becoming increasingly important. This also influences efficiency, however, individuals should also feel as comfortable as possible when interacting, in order to make the interaction as pleasant as possible. In this thesis, five novel prototypes that provide haptic feedback were designed, developed and evaluated. We have shown in two fields that it is worthwhile to include haptic feedback in the interaction to provide both, a higher performance and an enhanced experience. In the area of touchscreen interaction, three prototypes were presented that illustrated the versatility of haptic feedback. Psychophysical methods were used to evaluate the prototypes. Beyond that, it was shown that touch accuracy can be increased with haptic feedback. Perceptual aspects were investigated in the areas of Virtual and Remote Environments. The haptic interfaces were developed to make the user experience as pleasant as possible. We presented that haptic feedback on the feet increases spatial awareness, involvement and self-motion perception.

To conclude, I think it's always worth looking at how interaction can be adapted to make it easier, better or maybe just more fun. When haptic aspects are involved, I would be pleased.

References

- [AB11] Christian Afonso and Steffi Beckhaus. How to not hit a virtual wall: aural spatial awareness for collision avoidance in virtual environments. In Proceedings of the 6th Audio Mostly Conference on A Conference on Interaction with Sound - AM '11, pages 101–108, New York, New York, USA, 2011. ACM Press.
- [ABKS18] Oscar Ariza, Gerd Bruder, Nicholas Katzakis, and Frank Steinicke. Analysis of Proximity-Based Multimodal Feedback for 3D Selection in Immersive Virtual Environments. In 25th IEEE Conference on Virtual Reality and 3D User Interfaces, VR 2018 - Proceedings, Reutlingen, Germany, 2018.
- [AFHW⁺18] Firas Abi-Farrajl, Bernd Henze, Alexander Werner, Michael Panzirsch, Christian Ott, and Meximo A. Roa. Humanoid Teleoperation Using Task-Relevant Haptic Feedback. In *IEEE International Conference on Intelligent Robots and Systems*, pages 5010–5017. Institute of Electrical and Electronics Engineers Inc., dec 2018.
 - [AGWF11] Michelle Annett, Tovi Grossman, Daniel Wigdor, and George Fitzmaurice. Medusa: A Proximity-Aware Multi-touch Tabletop Michelle. In Proceedings of the 24th annual ACM symposium on User interface software and technology - UIST '11, page 337, New York, New York, USA, 2011. ACM Press.
 - [APPE20] Haneen Alsuradi, Chaitali Pawar, Wanjoo Park, and Mohamad Eid. Detection of Tactile Feedback on Touch-screen Devices using EEG Data. In *IEEE Haptics* Symposium, HAPTICS, 2020.
 - [AS94] Motoyuki Akamatsu and Sigeru Sato. A multi-modal mouse with tactile and force feedback. International Journal of Human Computer Studies, 40(3):443–453, jan 1994.
 - [Bab15] Earl R Babbie. The practice of social research. Nelson Education, 2015.
 - [BACM11] Michael S. Bernstein, Mark S. Ackerman, Ed H. Chi, and Robert C. Miller. The trouble with social computing systems research. In *Proceedings of the 2011 annual* conference extended abstracts on Human factors in computing systems - CHI EA '11, page 389, New York, New York, USA, 2011. ACM Press.
 - [Ban77] Albert Bandura. Self-efficacy: Toward a unifying theory of behavioral change. Psychological Review, 84(2):191–215, 1977.

- [BB10a] Kristopher J. Blom and Steffi Beckhaus. Virtual collision notification. In 3DUI 2010 - IEEE Symposium on 3D User Interfaces 2010, Proceedings, pages 35–38. IEEE, mar 2010.
- [BB10b] Andrea Bubka and Frederick Bonato. Natural Visual-Field Features Enhance Vection. *Perception*, 39(5):627–635, may 2010.
- [BBG08] Jelte E. Bos, Willem Bles, and Eric L. Groen. A theory on visually induced motion sickness. *Displays*, 29(2):47–57, mar 2008.
- [BC09] Patrick Baudisch and Gerry Chu. Back-of-device interaction allows creating very small touch devices. In Proceedings of the 27th international conference on Human factors in computing systems - CHI 09, page 1923, New York, New York, USA, 2009. ACM Press.
- [BCB07] Stephen Brewster, Faraz Chohan, and Lorna Brown. Tactile feedback for mobile interactions. In Proceedings of the SIGCHI conference on Human factors in computing systems - CHI '07, page 159, New York, New York, USA, 2007. ACM Press.
- [BCML15] Luca Brayda, Claudio Campus, Mariacarla Memeo, and Laura Lucagrossi. The Importance of Visual Experience, Gender, and Emotion in the Assessment of an Assistive Tactile Mouse. *IEEE Transactions on Haptics*, 8(3):279–286, jul 2015.
- [BHSO16] Hrvoje Benko, Christian Holz, Mike Sinclair, and Eyal Ofek. NormalTouch and TextureTouch. In Proceedings of the 29th Annual Symposium on User Interface Software and Technology - UIST '16, pages 717–728, New York, New York, USA, 2016. ACM Press.
- [BKAP11] Frederick Bonato, Juno Kim, Robert Allison, and Stephen Palmisano. Simulated Viewpoint Jitter Shakes Sensory Conflict Accounts of Vection. Seeing and Perceiving, 24(2):173–200, apr 2011.
- [BKH98] Doug A. Bowman, David Koller, and Larry F. Hodges. A methodology for the evaluation of travel techniques for immersive virtual environments. *Virtual Reality*, 3(2):120–131, jun 1998.
- [BKLP05] Doug A. Bowman, Ernst Kruijff, Joseph J. LaViola, and Ivan Poupyrev. 3D user interfaces : theory and practice. Addison-Wesley, 2005.
- [BKM09] Aaron Bangor, Philip Kortum, and James Miller. Determining What Individual SUS Scores Mean: Adding an Adjective Rating Scale. J. Usability Studies, 4(3):114– 123, 2009.
- [BLC⁺12] Sebastian Boring, David Ledo, Xiang 'Anthony' Chen, Nicolai Marquardt, Anthony Tang, and Saul Greenberg. The fat thumb. In Proceedings of the 14th international conference on Human-computer interaction with mobile devices and services -MobileHCI '12, page 39, New York, New York, USA, 2012. ACM Press.
- [BMK⁺20] Anil Ufuk Batmaz, Jens Maiero, Ernst Kruijff, Bernhard E Riecke, Carman Neustaedter, and Wolfgang Stuerzlinger. How automatic speed control based on distance affects user behaviours in telepresence robot navigation within dense conference-like environments. *PLOS ONE*, 15(11):1–41, 2020.

- [BO10] Gary Barrett and Ryomei Omote. Projected-capacitive touch technology. Information Display, 26(3), 2010.
- [BPIH10] Olivier Bau, Ivan Poupyrev, Ali Israr, and Chris Harrison. TeslaTouch: electrovibration for touch surfaces. In Proceedings of the 23nd annual ACM symposium on User interface software and technology - UIST '10, page 283, New York, New York, USA, 2010. ACM Press.
- [BPM09] Olivier Bau, Uros Petrevski, and Wendy Mackay. BubbleWrap: a textile-based electromagnetic haptic display. In Proceedings of the 27th international conference extended abstracts on Human factors in computing systems - CHI EA '09, page 3607, New York, New York, USA, 2009. ACM Press.
- [Bro96] John Brooke. SUS: A quick and dirty usability scale. Usability evaluation in industry, 189:4–7, 1996.
- [BSLM01] Bill Baxter, Vincent Scheib, Ming C. Lin, and Dinesh Manocha. DAB: Interactive Haptic Painting with 3D Virtual Brushes. In Proceedings of the 28th annual conference on Computer graphics and interactive techniques - SIGGRAPH '01, pages 461–468, New York, New York, USA, 2001. ACM Press.
 - [Bux07] Bill Buxton. Sketching User Experiences: Getting the Design Right and the Right Design. Morgan Kaufmann Publishers Inc., 2007.
 - [BVB10] Aurore Bourrelly, Jean L. Vercher, and Lionel Bringoux. Pitch body orientation influences the perception of self-motion direction induced by optic flow. *Neuroscience Letters*, 482(3):193–197, oct 2010.
 - [Cai16] Kelly Caine. Local Standards for Sample Size at CHI. In Proceedings of the 2016 CHI Conference on Human Factors in Computing Systems, pages 981–992, San Jose, California, USA, 2016. ACM Press.
- [CBMB16] Youngjun Cho, Andrea Bianchi, Nicolai Marquardt, and Nadia Bianchi-Berthouze. RealPen: Providing Realism in Handwriting Tasks on Touch Surfaces using Auditory-Tactile Feedback. In Proceedings of the 29th Annual Symposium on User Interface Software and Technology - UIST '16, pages 195–205, New York, New York, USA, 2016. ACM Press.
- [CCKB15] Christian Corsten, Christian Cherek, Thorsten Karrer, and Jan Borchers. HaptiCase: Back-of-Device Tactile Landmarks for Eyes-Free Absolute Indirect Touch. In Proceedings of the 33rd Annual ACM Conference on Human Factors in Computing Systems - CHI '15, pages 2171–2180, New York, New York, USA, 2015. ACM Press.
- [CDVB17] Christian Corsten, Bjoern Daehlmann, Simon Voelker, and Jan Borchers. BackX-Press: Using Back-of-Device Finger Pressure to Augment Touchscreen Input on Smartphones. In Proceedings of the 2017 CHI Conference on Human Factors in Computing Systems - CHI '17, pages 4654–4666, New York, New York, USA, 2017. ACM Press.
- [CHSD12] Victor Cheung, Jens Heydekorn, Stacey Scott, and Raimund Dachselt. Revisiting Hovering: Interaction Guides for Interactive Surfaces. In *Proceedings of the 2012*
ACM international conference on Interactive tabletops and surfaces - ITS '12, page 355, New York, New York, USA, 2012. ACM Press.

- [CK17] Heather Culbertson and Katherine J. Kuchenbecker. Ungrounded Haptic Augmented Reality System for Displaying Roughness and Friction. *IEEE/ASME Transactions on Mechatronics*, 22(4):1839–1849, aug 2017.
- [CSL⁺13] Tom Carter, Sue Ann Seah, Benjamin Long, Bruce Drinkwater, and Sriram Subramanian. UltraHaptics: Multi-Point Mid-Air Haptic Feedback for Touch Surfaces. In Proceedings of the 26th annual ACM symposium on User interface software and technology - UIST '13, pages 505–514, New York, New York, USA, 2013. ACM Press.
 - [CST15] Dapeng Chen, Aiguo Song, and Lei Tian. A novel miniature multi-mode haptic pen for image interaction on mobile terminal. In 2015 IEEE International Symposium on Haptic, Audio and Visual Environments and Games (HAVE), pages 1–6. IEEE, oct 2015.
 - [CV17] Juan Pablo Carrascal and Roel Vertegaal. Effects of Tactile Feedback on the Perception of Virtual Shapes on Non-Planar DisplayObjects. In Proceedings of the 2017 CHI Conference on Human Factors in Computing Systems - CHI '17, pages 4417–4423, New York, New York, USA, 2017. ACM Press.
 - [CW11] Douglas Cunningham and Christian Wallraven. Experimental Design: From User Studies to Psychophysics. Taylor & Francis Group, 2011.
- [DAJ⁺16] Jonathan Deber, Bruno Araujo, Ricardo Jota, Clifton Forlines, Darren Leigh, Steven Sanders, and Daniel Wigdor. Hammer Timel: A Low-Cost, High Precision, High Accuracy Tool to Measure the Latency of Touchscreen Devices. In Proceedings of the 2016 CHI Conference on Human Factors in Computing Systems - CHI '16, pages 2857–2868, New York, New York, USA, 2016. ACM Press.
- [DCC97] Rudolph P. Darken, William R. Cockayne, and David Carmein. The omnidirectional treadmill: a locomotion device for virtual worlds. In *Proceedings* of the 10th annual ACM symposium on User interface software and technology - UIST '97, pages 213–221, New York, New York, USA, 1997. Association for Computing Machinery (ACM).
- [DFAB04] Alan. Dix, Janet E. Finlay, Gregory D. Abowd, and Russell Beale. *Human-computer* interaction. Pearson/Prentice-Hall, 2004.
- [DvZN⁺13] Alexander De Luca, Emanuel von Zezschwitz, Ngo Dieu Huong Nguyen, Max-Emanuel Maurer, Elisa Rubegni, Marcello Paolo Scipioni, and Marc Langheinrich. Back-of-device authentication on smartphones. In *Proceedings of the SIGCHI* Conference on Human Factors in Computing Systems - CHI '13, page 2389, New York, New York, USA, 2013. ACM Press.
 - [EE68] Douglas C. Engelbart and William K. English. A research center for augmenting human intellect. In Proceedings of the December 9-11, 1968, fall joint computer conference, part I on - AFIPS '68 (Fall, part I), page 395, New York, New York, USA, 1968. ACM Press.

- [EOEC11] Abdulmotaleb El Saddik, Mauricio Orozco, Mohamad Eid, and Jongeun Cha. Haptics Technology: Bringing Touch to Multimedia. Springer, 2011.
 - [FDL16] Mi Feng, Arindam Dey, and Robert W. Lindeman. An initial exploration of a multi-sensory design space: Tactile support for walking in immersive virtual environments. In 2016 IEEE Symposium on 3D User Interfaces (3DUI), pages 95–104. IEEE, mar 2016.
 - [FH03] Andy P. Field and Graham Hole. *How to design and report experiments.* Sage Publications, 2003.
 - [Fis04] Kenneth P. Fishkin. A taxonomy for and analysis of tangible interfaces. Personal and Ubiquitous Computing, 8(5):347–358, sep 2004.
 - [FL14] Riazollah Firoozian and Frederick F Ling. Servo motors and industrial control theory. Springer, New York, 2nd edition, 2014.
- [FLO⁺12] Sean Follmer, Daniel Leithinger, Alex Olwal, Nadia Cheng, and Hiroshi Ishii. Jamming User Interfaces: Programmable Particle Stiffness and Sensing for Malleable and Shape-Changing Devices. In Proceedings of the 25th annual ACM symposium on User interface software and technology - UIST '12, page 519, New York, New York, USA, 2012. ACM Press.
- [FLO⁺13] Sean Follmer, Daniel Leithinger, Alex Olwal, Akimitsu Hogge, and Hiroshi Ishii. inFORM: Dynamic Physical Affordances and Constraints through Shape and Object Actuation. In Proceedings of the 26th annual ACM symposium on User interface software and technology - UIST '13, pages 417–426, New York, New York, USA, 2013. ACM Press.
 - [FN10] Paul Frowen and Donald Neale. *Neale's disorders of the foot.* Churchill Livingstone/Elsevier, 2010.
 - [FPG17] Nicholas Fellion, Thomas Pietrzak, and Audrey Girouard. FlexStylus: Leveraging Bend Input for Pen Interaction. In Proceedings of the 30th Annual ACM Symposium on User Interface Software and Technology - UIST '17, pages 375–385, New York, New York, USA, 2017. ACM Press.
 - [FS01] Masaaki Fukumoto and Toshiaki Sugimura. Active click: tactile feedback for touch pannels. In CHI '01 extended abstracts on Human factors in computing systems -CHI '01, page 121, New York, New York, USA, 2001. ACM Press.
- [GCHP10] Sidhant Gupta, Tim Campbell, Jeffrey R. Hightower, and Shwetak N. Patel. SqueezeBlock: Using Virtual Springs in Mobile Devices for Eyes-Free Interaction. In Proceedings of the 23nd annual ACM symposium on User interface software and technology - UIST '10, page 101, New York, New York, USA, 2010. ACM Press.
- [GDT⁺12] Judith E. Gold, Jeffery B. Driban, Nadja Thomas, T. Chakravarty, V. Channell, and Eugene Komaroff. Postures, typing strategies, and gender differences in mobile device usage: An observational study. *Applied Ergonomics*, 43(2):408–412, mar 2012.
 - [Ges97] George A. Gescheider. *Psychophysics: The Fundamentals.* L. Erlbaum Associates, jun 1997.

- [GG11] Cheng Gu and Michael J. Griffin. Vibrotactile thresholds at the sole of the foot: Effect of vibration frequency and contact location. Somatosensory & Motor Research, 28(3-4):86–93, sep 2011.
- [Gib80] James J. Gibson. The Ecological Approach to Visual Perception. The Journal of Aesthetics and Art Criticism, 39(2):203, 1980.
- [GLA⁺88] Gerald E. Grossman, Richard J. Leigh, Larry Abel, Douglas J. Lanska, and Stephen E. Thurston. Frequency and velocity of rotational head perturbations during locomotion. *Experimental brain research*, 70(3):470–6, 1988.
- [GMPT13] Sidhant Gupta, Dan Morris, Shwetak N. Patel, and Desney Tan. AirWave: noncontact haptic feedback using air vortex rings. In Proceedings of the 2013 ACM international joint conference on Pervasive and ubiquitous computing - UbiComp '13, page 419, New York, New York, USA, 2013. ACM Press.
- [GMV⁺08] Bruno L. Giordano, Stephen McAdams, Yon Visell, Jeremy Cooperstock, Hsin-Yun Yao, and Vincent Hayward. Non-visual identification of walking grounds. The Journal of the Acoustical Society of America, 123(5):3412–3412, may 2008.
 - [Gol10] E. Bruce Goldstein. *Sensation and perception*. Wadsworth, Cengage Learning, 2010.
 - [GPI⁺15] Emilie Guy, Parinya Punpongsanon, Daisuke Iwai, Kosuke Sato, and Tamy Boubekeur. LazyNav: 3D ground navigation with non-critical body parts. In 2015 IEEE Symposium on 3D User Interfaces (3DUI), pages 43–50. IEEE, mar 2015.
 - [Gru08] Martin Grunwald. Human haptic perception basics and applications. Birkhäuser Verlag, 1 edition, 2008.
 - [GS05] Graham Gibbs and Claire Simpson. Conditions under which assessment supports students' learning. *Learning and Teaching in Higher Education*, 1:3–31, 2005.
- [GSK⁺20] Aakar Gupta, Majed Samad, Kenrick Kin, Per Ola Kristensson, and Hrvoje Benko. Investigating Remote Tactile Feedback for Mid-Air Text-Entry in Virtual Reality. In 2020 IEEE International Symposium on Mixed and Augmented Reality (ISMAR), pages 350–360. IEEE, nov 2020.
- [GSV08] David T. Gallant, Andrew G. Seniuk, and Roel Vertegaal. Towards more paper-like input: flexible input devices for foldable interaction styles. In Proceedings of the 21st annual ACM symposium on User interface software and technology - UIST '08, page 283, New York, New York, USA, 2008. ACM Press.
- [GTLG14] Gheorghita Ghinea, Christian Timmerer, Weisi Lin, and Stephen R. Gulliver. Mulsemedia: State of the art, perspectives, and challenges. ACM Transactions on Multimedia Computing, Communications and Applications, 2014.
- [GZBV16] Daniel Gotsch, Xujing Zhang, Jesse Burstyn, and Roel Vertegaal. HoloFlex: A Flexible Holographic Smartphone with Bend Input. In Proceedings of the 2016 CHI Conference Extended Abstracts on Human Factors in Computing Systems -CHI EA '16, pages 3675–3678, New York, New York, USA, 2016. ACM Press.

- [HBC⁺92] Thomas Hewett, Ronald Baecker, Stuart Card, Tom Carey, Jean Gasen, Marilyn Mantei, Gary Perlman, Gary Strong, and William Verplank. ACM SIGCHI Curricula for Human-Computer Interaction. Association for Computing Machinery, jan 1992.
- [HBH⁺16] Ken Hinckley, William Buxton, Seongkook Heo, Michel Pahud, Christian Holz, Hrvoje Benko, Abigail Sellen, Richard Banks, Kenton O'Hara, and Gavin Smyth. Pre-Touch Sensing for Mobile Interaction. In Proceedings of the 2016 CHI Conference on Human Factors in Computing Systems - CHI '16, pages 2869–2881, New York, New York, USA, 2016. ACM Press.
- [HCBK09] Eve Hoggan, Andrew Crossan, Stephen A. Brewster, and Topi Kaaresoja. Audio or Tactile Feedback: Which Modality When? In Proceedings of the 27th international conference on Human factors in computing systems - CHI 09, page 2253, New York, New York, USA, 2009. ACM Press.
 - [HGL14] Jaehyun Han, Jiseong Gu, and Geehyuk Lee. Trampoline: A Double-sided Elastic Touch Device to Create Reliefs. In Proceedings of the 27th annual ACM symposium on User interface software and technology - UIST '14, pages 383–388, New York, New York, USA, 2014. ACM Press.
 - [HH05] Vanessa Harrar and Laurence R. Harris. Simultaneity constancy: detecting events with touch and vision. In *Experimental Brain Research*, 2005.
 - [HK08] Martin Hachet and Alexander Kulik. Elastic Control for Navigation Tasks on Pen-based Handheld Computers. In Proceedings of the 2008 IEEE Symposium on 3D User Interfaces, pages 91–96, USA, mar 2008. IEEE Computer Society.
- [HKAI16] Khalad Hasan, Junhyeok Kim, David Ahlström, and Pourang Irani. Thumbs-Up: 3D Spatial Thumb-Reachable Space for One-Handed Thumb Interaction on Smartphones. In Proceedings of the 2016 Symposium on Spatial User Interaction -SUI '16, pages 103–106, New York, New York, USA, 2016. ACM Press.
- [HKK13] Akif Hacinecipoglu, E. I. Konukseven, and A. B. Koku. Evaluation of haptic feedback cues on vehicle teleoperation performance in an obstacle avoidance scenario. In 2013 World Haptics Conference (WHC), pages 689–694. IEEE, apr 2013.
- [HKLB08] Eve Hoggan, Topi Kaaresoja, Pauli Laitinen, and Stephen Brewster. Crossmodal Congruence: The Look, Feel and Sound of Touchscreen Widgets. In Proceedings of the 10th international conference on Multimodal interfaces - IMCI '08, page 157, New York, New York, USA, 2008. ACM Press.
 - [HL13] Seongkook Heo and Geehyuk Lee. Indirect shear force estimation for multi-point shear force operations. In Proceedings of the SIGCHI Conference on Human Factors in Computing Systems - CHI '13, page 281, New York, New York, USA, 2013. ACM Press.
- [HMT03] Shigeo Hiraoka, Isshin Miyamoto, and Kiyoshi Tomimatsu. Behind Touch: A text input method for mobile phone by the back and tactile sense interface. Information Processing Society of Japan, Interaction, pages 131–146, 2003.

- [HNW⁺14] Alyssa Harris, Kevin Nguyen, Preston Tunnell Wilson, Matthew Jackoski, and Betsy Williams. Human joystick: Wii-leaning to translate in large virtual environments. In Proceedings - VRCAI 2014: 13th ACM SIGGRAPH International Conference on Virtual-Reality Continuum and Its Applications in Industry, pages 231–234, 2014.
- [HOAH15] Jess Hartcher-O'Brien, Malika Auvray, and Vincent Hayward. Perception of distance-to-obstacle through time-delayed tactile feedback. In 2015 IEEE World Haptics Conference (WHC), pages 7–12. IEEE, jun 2015.
 - [HS88] Sandra G. Hart and Lowell E. Staveland. Development of NASA-TLX (Task Load Index): Results of Empirical and Theoretical Research. Advances in Psychology, 52:139–183, jan 1988.
- [HTE⁺17] Thomas Hausberger, Michael Terzer, Florian Enneking, Zofia Jonas, and Yeongmi Kim. SurfTics-Kinesthetic and tactile feedback on a touchscreen device. In 2017 IEEE World Haptics Conference, WHC 2017, pages 472–477. Institute of Electrical and Electronics Engineers Inc., jul 2017.
 - [HV08] David Holman and Roel Vertegaal. Organic user interfaces: designing computers in any way, shape, or form. *Communications of the ACM*, 51(6):48, jun 2008.
- [HVA⁺05] David Holman, Roel Vertegaal, Mark Altosaar, Nikolaus Troje, and Derek Johns. Paper windows: interaction techniques for digital paper. In Proceedings of the SIGCHI conference on Human factors in computing systems - CHI '05, page 591, New York, New York, USA, 2005. ACM Press.
- [HWMZ91] Blake Hannaford, Laurie Wood, Douglas A. McAffee, and Haya Zak. Performance evaluation of a six-axis generalized force-reflecting teleoperator. *IEEE Transactions* on Systems, Man, and Cybernetics, 21(3):620–633, 1991.
 - [HZB⁺19] Lars Erik Holmquist, Oren Zuckerman, Rafael Ballagas, Hiroshi Ishii, Kimiko Ryokai, and Haiyan Zhang. The Future of Tangible User Interfaces. In Extended Abstracts of the 2019 CHI Conference on Human Factors in Computing Systems -CHI '19, pages 1–6, New York, New York, USA, 2019. ACM Press.
 - [IKN08] Masaki Iwabuchi, Yasuaki Kakehi, and Takeshi Naemura. LimpiDual touch: interactive limpid display with dual-sided touch sensing. In ACM SIGGRAPH 2008 posters on - SIGGRAPH '08, page 1, New York, New York, USA, 2008. ACM Press.
 - [ISK⁺14] Yasushi Ikei, Seiya Shimabukuro, Shunki Kato, Yujiro Okuya, Koji Abe, Koichi Hirota, and Tomohiro Amemiya. Rendering of Virtual Walking Sensation by a Passive Body Motion. In *Haptics: Neuroscience, Devices, Modeling, and Applications*, pages 150–157. Springer, Berlin, Heidelberg, 2014.
 - [IU97] Hiroshi Ishii and Brygg Ullmer. Tangible Bits: Towards Seamless Interfaces between People, Bits and Atoms. In Proceedings of the SIGCHI conference on Human factors in computing systems - CHI '97, pages 234–241, New York, New York, USA, 1997. ACM Press.

- [IYN01] Hiroo Iwata, Hiroaki Yano, and Fumitaka Nakaizumi. Gait Master: a versatile locomotion interface for uneven virtual terrain. In *Proceedings IEEE Virtual Reality*, pages 131–137, USA, 2001. IEEE Comput. Soc.
- [JGF⁺16] Yves Jammes, Julia Guimbaud, Rémi Faure, Patricia Griffon, Jean Paul Weber, Bruno Vie, and Regis Guieu. Psychophysical estimate of plantar vibration sensitivity brings additional information to the detection threshold in young and elderly subjects. *Clinical Neurophysiology Practice*, 1:26–32, 2016.
 - [JH05] Caroline Jay and Roger Hubbold. Delayed visual and haptic feedback in a reciprocal tapping task. In Proceedings - 1st Joint Eurohaptics Conference and Symposium on Haptic Interfaces for Virtual Environment and Teleoperator Systems; World Haptics Conference, WHC 2005, pages 655–656. Institute of Electrical and Electronics Engineers Inc., 2005.
 - [JKB10] Yvonne Jansen, Thorsten Karrer, and Jan Borchers. MudPad: tactile feedback and haptic texture overlay for touch surfaces. In ACM International Conference on Interactive Tabletops and Surfaces - ITS '10, page 11, New York, New York, USA, 2010. ACM Press.
- [JKT⁺16] Sungjune Jang, Lawrence H. Kim, Kesler Tanner, Hiroshi Ishii, and Sean Follmer. Haptic Edge Display for Mobile Tactile Interaction. In Proceedings of the 2016 CHI Conference on Human Factors in Computing Systems - CHI '16, pages 3706–3716, New York, New York, USA, 2016. ACM Press.
 - [JL12] Jia Wang and R. W. Lindeman. Comparing isometric and elastic surfboard interfaces for leaning-based travel in 3D virtual environments. In 2012 IEEE Symposium on 3D User Interfaces (3DUI), pages 31–38. IEEE, mar 2012.
- [JMM⁺20] Brennan Jones, Jens Maiero, Alireza Mogharrab, Ivan A Aguliar, Ashu Adhikari, Bernhard E Riecke, Ernst Kruijff, Carman Neustaedter, and Robert W Lindeman. FeetBack: Augmenting Robotic Telepresence with Haptic Feedback on the Feet. In Proceedings of the 2020 International Conference on Multimodal Interaction, ICMI '20, pages 194–203, New York, NY, USA, 2020. Association for Computing Machinery.
 - [JP02] Norman P. Jouppi and Norman P. First steps towards mutually-immersive mobile telepresence. In Proceedings of the 2002 ACM conference on Computer supported cooperative work - CSCW '02, page 354, New York, New York, USA, 2002. ACM Press.
- [JRMT15] Steven Johnson, Irene Rae, Bilge Mutlu, and Leila Takayama. Can You See Me Now?: How Field of View Affects Collaboration in Robotic Telepresence. In Proceedings of the 33rd Annual ACM Conference on Human Factors in Computing Systems - CHI '15, pages 2397–2406, New York, New York, USA, 2015. ACM Press.
 - [JS19] Chanyoung Ju and Hyoung Il Son. Evaluation of Haptic Feedback in the Performance of a Teleoperated Unmanned Ground Vehicle in an Obstacle Avoidance Scenario. International Journal of Control, Automation and Systems, 17(1):168– 180, jan 2019.

- [JT13] Lynette A. Jones and Hong Z. Tan. Application of Psychophysical Techniques to Haptic Research. *IEEE Transactions on Haptics*, 6(3):268–284, jul 2013.
- [JWB⁺15] Brennan Jones, Anna Witcraft, Scott Bateman, Carman Neustaedter, and Anthony Tang. Mechanics of Camera Work in Mobile Video Collaboration. In Proceedings of the 33rd Annual ACM Conference on Human Factors in Computing Systems -CHI '15, pages 957–966, New York, New York, USA, 2015. ACM Press.
 - [KBL14] Topi Kaaresoja, Stephen Brewster, and Vuokko Lantz. Towards the Temporally Perfect Virtual Button: Touch-Feedback Simultaneity and Perceived Quality in Mobile Touchscreen Press Interactions. ACM Transactions on Applied Perception, 11(2):1–25, jun 2014.
- [KBT⁺19] Ernst Kruijff, Saugata Biswas, Christina Trepkowski, Jens Maiero, George Ghinea, and Wolfgang Stuerzlinger. Multilayer haptic feedback for pen-based tablet interaction. In Conference on Human Factors in Computing Systems - Proceedings, 2019.
- [KHSK14] Seung-Chan Kim, Byung-Kil Han, Jiwon Seo, and Dong-Soo Kwon. Haptic interaction with virtual geometry on robotic touch surface. In SIGGRAPH Asia 2014 Emerging Technologies on - SA '14, pages 1–3, New York, New York, USA, 2014. ACM Press.
 - [Kil10] Johan Kildal. 3D-press: Haptic Illusion of Compliance when Pressing on a Rigid Surface. In International Conference on Multimodal Interfaces and the Workshop on Machine Learning for Multimodal Interaction on - ICMI-MLMI '10, page 1, New York, New York, USA, 2010. ACM Press.
 - [KIP13] Seung-Chan Kim, Ali Israr, and Ivan Poupyrev. Tactile rendering of 3D features on touch surfaces. In Proceedings of the 26th annual ACM symposium on User interface software and technology - UIST '13, pages 531–538, New York, New York, USA, 2013. ACM Press.
- [KJWL09] Shaun K. Kane, Chandrika Jayant, Jacob O. Wobbrock, and Richard E. Ladner. Freedom to roam: a study of mobile device adoption and accessibility for people with visual and motor disabilities. In Proceeding of the eleventh international ACM SIGACCESS conference on Computers and accessibility - ASSETS '09, page 115, New York, New York, USA, 2009. ACM Press.
 - [KKL14] Andrey Kiselev, Annica Kristoffersson, and Amy Loutfi. The effect of field of view on social interaction in mobile robotic telepresence systems. In Proceedings of the 2014 ACM/IEEE international conference on Human-robot interaction - HRI '14, pages 214–215, New York, New York, USA, 2014. ACM Press.
 - [KLS09] Ki-Uk Kyung, Jun-Young Lee, and Mandayam A. Srinivasan. Precise manipulation of GUI on a touch screen with haptic cues. In World Haptics 2009 - Third Joint EuroHaptics conference and Symposium on Haptic Interfaces for Virtual Environment and Teleoperator Systems, pages 202–207. IEEE, 2009.
- [KMRL19] Abinaya Kumar, Sven Mayer, Aishwarya Radjesh, and Huy Viet Le. Improving the input accuracy of touchscreens using deep learning. In Extended Abstracts of the 2019 CHI Conference on Human Factors in Computing Systems, pages 1–6, New York, NY, USA, 2019. ACM Press.

- [KMT⁺15] Ernst Kruijff, Alexander Marquardt, Christina Trepkowski, Jonas Schild, and Andre Hinkenjann. Enhancing User Engagement in Immersive Games through Multisensory Cues. In 2015 7th International Conference on Games and Virtual Worlds for Serious Applications (VS-Games), pages 1–8. IEEE, sep 2015.
- [KMT⁺16] Ernst Kruijff, Alexander Marquardt, Christina Trepkowski, Robert W. Lindeman, Andre Hinkenjann, Jens Maiero, and Bernhard E. Riecke. On Your Feet!: Enhancing Vection in Leaning-Based Interfaces through Multisensory Stimuli. In *Proceedings* of the 2016 Symposium on Spatial User Interaction - SUI '16, pages 149–158, New York, New York, USA, 2016. ACM Press.
 - [Kos10] Robert J Kosinski. A Literature Review on Reaction Time. Retrieved httpbiaeclemsonedubpcbpLab110reactionhtm 06042010, 2010.
 - [KPP13] Roberta L. Klatzky, Dianne Pawluk, and Angelika Peer. Haptic Perception of Material Properties and Implications for Applications. *Proceedings of the IEEE*, 101(9):2081–2092, 2013.
- [KRTK15] Ernst Kruijff, Bernhard Riecke, Christina Trekowski, and Alexandra Kitson. Upper Body Leaning can affect Forward Self-Motion Perception in Virtual Environments. In Proceedings of the 3rd ACM Symposium on Spatial User Interaction - SUI '15, pages 103–112, New York, New York, USA, 2015. ACM Press.
- [KSL⁺13] Sunjun Kim, Jeongmin Son, Geehyuk Lee, Hwan Kim, and Woohun Lee. TapBoard. In Proceedings of the SIGCHI Conference on Human Factors in Computing Systems
 - CHI '13, page 553, New York, New York, USA, 2013. ACM Press.
- [KTC⁺12] Maria Korman, Kinneret Teodorescu, Adi Cohen, Miriam Reiner, and Daniel Gopher. Effects of Order and Sensory Modality in Stiffness Perception. Presence: Teleoperators and Virtual Environments, 21(3):295–304, aug 2012.
- [Kum10] Ranjit Kumar. Research methodology : a step-by-step guide for beginners. Sage Publications, 3rd edition, 2010.
- [KWBT91] Kurt .A. Kaczmarek, John G. Webster, Paul Bach-y-Rita, and Willis J. Tompkins. Electrotactile and vibrotactile displays for sensory substitution systems. *IEEE Transactions on Biomedical Engineering*, 38(1):1–16, 1991.
- [LAB⁺15] Paul Lubos, Oscar Ariza, Gerd Bruder, Florian Daiber, Frank Steinicke, and Antonio Krüger. HoverSpace: Analyses of the Perceived Spatial Affordances of Hover Interaction Above Tabletop Surfaces. In *Human-Computer Interaction*, INTERACT, pages 259–277. Springer, 2015.
 - [LB13] Pedro Lopes and Patrick Baudisch. Muscle-propelled force feedback. In Proceedings of the SIGCHI Conference on Human Factors in Computing Systems - CHI '13, page 2577, New York, New York, USA, 2013. ACM Press.
- [LDL⁺04] Johnny C. Lee, Paul H. Dietz, Darren Leigh, William S. Yerazunis, and Scott E. Hudson. Haptic Pen: A Tactile Feedback Stylus for Touch Screens. In Proceedings of the 17th annual ACM symposium on User interface software and technology -UIST '04, page 291, New York, New York, USA, 2004. ACM Press.

- [LFH17] Jonathan Lazar, Jinjuan Heidi Feng, and Harry Hochheiser. Research Methods in Human-Computer Interaction. Wiley Publishing, 2017.
- [LGB⁺16] David Lindlbauer, Jens Emil Grønbæk, Morten Birk, Kim Halskov, Marc Alexa, and Jörg Müller. Combining Shape-Changing Interfaces and Spatial Augmented Reality Enables Extended Object Appearance. In Proceedings of the 2016 CHI Conference on Human Factors in Computing Systems - CHI '16, pages 791–802, New York, New York, USA, 2016. ACM Press.
- [LGH⁺03] Charles Lenay, Olivier Gapenne, Sylvain Hanneton, Catherine Marque, and Christelle Genouëlle. Sensory Substitution : Limits and Perspectives. *Touching for knowing*, pages 275–292, 2003.
 - [LK09] Susan. J. Lederman and Roberta L. Klatzky. Haptic perception: A tutorial, 2009.
- [LKJ⁺10] Sang-Su Lee, Sohyun Kim, Bipil Jin, Eunji Choi, Boa Kim, Xu Jia, Daeeop Kim, and Kun-pyo Lee. How users manipulate deformable displays as input devices. In Proceedings of the 28th international conference on Human factors in computing systems - CHI '10, page 1647, New York, New York, USA, 2010. ACM Press.
- [LKSP04] Sangyoon Lee, Gerard J. Kim, Gaurav S. Sukhatme, and Chan-mo Park. Effects of Haptic Feedback on Telepresence and Navigational Performance. In Proceedings of International Conference on Artificial Reality and Telexistence, 2004.
- [LMBH18] Huy Viet Le, Sven Mayer, Patrick Bader, and Niels Henze. Fingers' Range and Comfortable Area for One-Handed Smartphone Interaction Beyond the Touchscreen. In Proceedings of the 2018 CHI Conference on Human Factors in Computing Systems - CHI '18, pages 1–12, New York, New York, USA, 2018. ACM Press.
- [LMBS07] Rock Leung, Karon MacLean, Martin Bue Bertelsen, and Mayukh Saubhasik. Evaluation of haptically augmented touchscreen gui elements under cognitive load. In Proceedings of the ninth international conference on Multimodal interfaces -ICMI '07, page 374, New York, New York, USA, 2007. ACM Press.
- [LOM⁺11] Vincent Levesque, Louise Oram, Karon MacLean, Andy Cockburn, Nicholas Marchuk, Dan Johnson, J. Edward Colgate, and Michael Peshkin. Frictional widgets: enhancing touch interfaces with programmable friction. In *Proceedings* of the 2011 annual conference extended abstracts on Human factors in computing systems - CHI EA '11, page 1153, New York, New York, USA, 2011. ACM Press.
 - [LR14] Ben Lawson and Bernhard Riecke. Perception of Body Motion, 2014.
- [LSMM⁺05] Robert W. Lindeman, John L. Sibert, Erick Mendez-Mendez, Sachin Patil, and Daniel Phifer. Effectiveness of directional vibrotactile cuing on a building-clearing task. In Proceedings of the SIGCHI conference on Human factors in computing systems - CHI '05, page 271, New York, New York, USA, 2005. ACM Press.
 - [LWLS11] Liping Lin, Yongtian Wang, Yue Liu, and Makoto Sato. Application of Pen-Based Planar Haptic Interface in Physics Education. In 2011 12th International Conference on Computer-Aided Design and Computer Graphics, pages 375–378. IEEE, sep 2011.

- [MCV⁺13] Maud Marchal, Gabriel Cirio, Yon Visell, Federico Fontana, Stefania Serafin, Jeremy Cooperstock, and Anatole Lécuyer. Multimodal Rendering of Walking Over Virtual Grounds. In Human Walking in Virtual Environments, pages 263–295. Springer New York, New York, NY, 2013.
- [MEK⁺19] Jens Maiero, David Eibich, Ernst Kruijff, Andre Hinkenjann, Wolfgang Stuerzlinger, Hrvoje Benko, and Gheorghita Ghinea. Back-of-Device Force Feedback Improves Touchscreen Interaction for Mobile Devices. *IEEE Transactions on Haptics*, 2019.
- [MHL⁺13] Victor Mateevitsi, Brad Haggadone, Jason Leigh, Brian Kunzer, and Robert V. Kenyon. Sensing the environment through SpiderSense. In Proceedings of the 4th Augmented Human International Conference on - AH '13, pages 51–57, New York, New York, USA, 2013. ACM Press.
- [MKG⁺14] Mathias Müller, Anja Knöfel, Thomas Gründer, Ingmar Franke, and Rainer Groh. FlexiWall: Exploring Layered Data with Elastic Displays. In Proceedings of the Ninth ACM International Conference on Interactive Tabletops and Surfaces - ITS '14, pages 439–442, New York, New York, USA, 2014. ACM Press.
- [MKHG17] Jens Maiero, Ernst Kruijff, Andre Hinkenjann, and Gheorghita Ghinea. ForceTab: Visuo-haptic interaction with a force-sensitive actuated tablet. In 2017 IEEE International Conference on Multimedia and Expo (ICME), pages 169–174. IEEE, jul 2017.
- [MKT⁺18] Alexander Marquardt, Ernst Kruijff, Christina Trepkowski, Jens Maiero, Andrea Schwandt, André Hinkenjann, Wolfgang Stuerzlinger, and Johannes Schöning. Audio-tactile proximity feedback for enhancing 3D manipulation. In Proceedings of the 24th ACM Symposium on Virtual Reality Software and Technology - VRST '18, pages 1–10, New York, New York, USA, 2018. ACM Press.
- [MNY⁺09] Nicolai Marquardt, Miguel A. Nacenta, James E. Young, Sheelagh Carpendale, Saul Greenberg, and Ehud Sharlin. The Haptic Tabletop Puck: Tactile Feedback for Interactive Tabletops Nicolai. In Proceedings of the ACM International Conference on Interactive Tabletops and Surfaces - ITS '09, page 85, New York, New York, USA, 2009. ACM Press.
 - [MPL11] Maud Marchal, Julien Pettre, and Anatole Lecuyer. Joyman: A human-scale joystick for navigating in virtual worlds. In 2011 IEEE Symposium on 3D User Interfaces (3DUI), pages 19–26. IEEE, mar 2011.
 - [MRS15] A.K. Matta, Ranga D. Raju, and K.N.S. Suman. The Integration of CAD/CAM and Rapid Prototyping in Product Development: A Review. *Materials Today: Proceedings*, 2(4-5):3438–3445, jan 2015.
 - [MS05] Laurent Mignonneau and Christa Sommerer. Nano-Scape: Experiencing Aspects of Nanotechnology through a Magnetic Force-Feedback Interface. In Proceedings of the 2005 ACM SIGCHI International Conference on Advances in computer entertainment technology - ACE '05, pages 200–203, New York, New York, USA, 2005. ACM Press.
 - [MW11] Micah M. Murray and Mark T. Wallace. The neural bases of multisensory processes. 2011.

- [Mye98] Brad A. Myers. A brief history of human-computer interaction technology. *interactions*, 1998.
- [NFM⁺19] Ken Nakagaki, Daniel Fitzgerald, Zhiyao (John) Ma, Luke Vink, Daniel Levine, and Hiroshi Ishii. inFORCE: Bi-directional 'Force' Shape Display for Haptic Interaction. In Proceedings of the Thirteenth International Conference on Tangible, Embedded, and Embodied Interaction - TEI '19, pages 615–623, New York, New York, USA, 2019. ACM Press.
- [NKW11] Tao Ni, Amy K. Karlson, and Daniel Wigdor. AnatOnMe: Facilitating Doctor-Patient Communication Using a Projection-Based Handheld Device. In Proceedings of the 2011 annual conference on Human factors in computing systems - CHI '11, page 3333, New York, New York, USA, 2011. ACM Press.
- [NNTA14] Takashi Nagamatsu, Masahiro Nakane, Haruka Tashiro, and Teruhiko Akazawa. Multi-push Display using 6-axis Motion Platform. In Proceedings of the Ninth ACM International Conference on Interactive Tabletops and Surfaces - ITS '14, pages 65–68, New York, New York, USA, 2014. ACM Press.
- [NNTS12] Niels C. Nilsson, Rolf Nordahl, Luca Turchet, and Stefania Serafin. Audio-Haptic Simulation of Walking on Virtual Ground Surfaces to Enhance Realism. In *Haptic* and Audio Interaction Design, pages 61–70. Springer, Berlin, Heidelberg, 2012.
 - [NO17] Veronica Ahumada Newhart and Judith S. Olson. My Student is a Robot: How Schools Manage Telepresence Experiences for Students. In Proceedings of the 2017 CHI Conference on Human Factors in Computing Systems - CHI '17, pages 342–347, New York, New York, USA, 2017. ACM Press.
 - [Nor13] Don Norman. The design of everyday things, volume 51. 2013.
 - [NS98] Shinji Nakamura and Shinsuke Shimojo. Orientation of Selective Effects of Body Tilt on Visually Induced Perception of Self-Motion. *Perceptual and Motor Skills*, 87(2):667–672, oct 1998.
- [NVPH16] Carman Neustaedter, Gina Venolia, Jason Procyk, and Dan Hawkins. To Beam or Not to Beam: A Study of Remote Telepresence Attendance at an Academic Conference. In Proceedings of the 19th ACM Conference on Computer-Supported Cooperative Work & Social Computing - CSCW '16, pages 417–430, New York, New York, USA, 2016. ACM Press.
 - [Oka18] Allison M. Okamura. Haptic Dimensions of Human-Robot Interaction. ACM Transactions on Human-Robot Interaction, 7(1):1–3, may 2018.
- [OOH⁺19] Jifei Ou, Daniel Oran, Don Derek Haddad, Joseph Paradiso, and Hiroshi Ishii. SensorKnit: Architecting Textile Sensors with Machine Knitting. 3D Printing and Additive Manufacturing, 6(1):1–11, mar 2019.
- [OSC⁺17] Sharon Oviatt, Bjoern Schuller, Philip R. Cohen, Daniel Sonntag, Gerasimos Potamianos, and Antonio Kruger. The Handbook of Multimodal-Multisensor Interfaces: Foundations, User Modeling, and Common Modality Combinations -Volume 1. [s.n.], 2017.

- [OVV⁺16] Marianna Obrist, Carlos Velasco, Chi Thanh Vi, Nimesha Ranasinghe, Ali Israr, Adrian D. Cheok, Charles Spence, and Ponnampalam Gopalakrishnakone. Touch, taste, & smell user interfaces: The future of multisensory HCI. In *Conference on Human Factors in Computing Systems - Proceedings*, volume 07-12-May-, pages 3285–3292, New York, New York, USA, may 2016. Association for Computing Machinery.
- [PBKW18] Denise Prescher, Jens Bornschein, Wiebke Köhlmann, and Gerhard Weber. Touching graphical applications: bimanual tactile interaction on the HyperBraille pinmatrix display. Universal Access in the Information Society, 17(2):391–409, jun 2018.
- [PFC⁺10] Stefano Papetti, Federico Fontana, Marco Civolani, Amir Berrezag, and Vincent Hayward. Audio-tactile display of ground properties using interactive shoes. In Proceedings of the 5th International Conference on Haptic and Audio Interaction Design, volume 6306 LNCS, pages 117–128. Springer, Berlin, Heidelberg, 2010.
- [PGG⁺12] Joshua Peschke, Fabian Göbel, Thomas Gründer, Mandy Keck, Dietrich Kammer, and Rainer Groh. DepthTouch: an elastic surface for tangible computing. In Proceedings of the International Working Conference on Advanced Visual Interfaces
 - AVI '12, page 770, New York, New York, USA, 2012. ACM Press.
 - [PH20] Lucie Pantera and Charles Hudin. Multitouch Vibrotactile Feedback on a Tactile Screen by the Inverse Filter Technique: Vibration Amplitude and Spatial Resolution. *IEEE Transactions on Haptics*, 2020.
 - [PKB06] Pekka Parhi, Amy K. Karlson, and Benjamin B. Bederson. Target size study for one-handed thumb use on small touchscreen devices. In Proceedings of the 8th conference on Human-computer interaction with mobile devices and services -MobileHCI '06, page 203, New York, New York, USA, 2006. ACM Press.
 - [PM03] Ivan Poupyrev and Shigeaki Maruyama. Tactile interfaces for small touch screens. In Proceedings of the 16th annual ACM symposium on User interface software and technology - UIST '03, pages 217–220, New York, New York, USA, 2003. ACM Press.
 - [POM04] Ivan Poupyrev, Makoto Okabe, and Shigeaki Maruyama. Haptic Feedback for Pen Computing: Directions and Strategies. In Extended abstracts of the 2004 conference on Human factors and computing systems - CHI '04, page 1309, New York, New York, USA, 2004. ACM Press.
- [PST⁺11] Andreas Paepcke, Bianca Soto, Leila Takayama, Frank Koenig, and Blaise Gassend. Yelling in the hall: using sidetone to address a problem with mobile remote presence systems. In Proceedings of the 24th annual ACM symposium on User interface software and technology - UIST '11, page 107, New York, New York, USA, 2011. ACM Press.
 - [PT19] Gozde Gunduz Poyraz and Ozgur Tamer. Different Haptic Senses with Multiple Vibration Motors. In ELECO 2019 - 11th International Conference on Electrical and Electronics Engineering, 2019.

- [RBM⁺10] Bernhard E. Riecke, Bobby Bodenheimer, Timothy P. McNamara, Betsy Williams, Peng Peng, and Daniel Feuereissen. Do We Need to Walk for Effective Virtual Reality Navigation? Physical Rotations Alone May Suffice. In Christoph Hölscher, Thomas F. Shipley, Marta Olivetti Belardinelli, John A. Bateman, and Nora S. Newcombe, editors, *Spatial Cognition*, pages 234–247. Springer, Berlin, Heidelberg, 7th edition, 2010.
 - [RF12] Bernhard E. Riecke and Daniel Feuereissen. To move or not to move: Can active control and user-driven motion cueing enhance self-motion perception ("vection") in virtual reality? In *Proceedings of the ACM Symposium on Applied Perception*, Los Angeles, California, 2012. ACM Press.
 - [RFR09] Bernhard E. Riecke, Daniel Feuereissen, and John J. Rieser. Auditory self-motion simulation is facilitated by haptic and vibrational cues suggesting the possibility of actual motion. ACM Transactions on Applied Perception, 6(3):1–22, aug 2009.
- [RGOB15] Alexander Russomanno, R. Brent Gillespie, Sile O'Modhrain, and Mark Burns. The design of pressure-controlled valves for a refreshable tactile display. In 2015 IEEE World Haptics Conference (WHC), pages 177–182. IEEE, jun 2015.
- [RMT⁺14] Irene Rae, Bilge Mutlu, Leila Takayama, Irene Rae, Bilge Mutlu, and Leila Takayama. Bodies in motion. In Proceedings of the 32nd annual ACM conference on Human factors in computing systems - CHI '14, pages 2153–2162, New York, New York, USA, 2014. ACM Press.
 - [RN17] Irene Rae and Carman Neustaedter. Robotic Telepresence at Scale. In Proceedings of the 2017 CHI Conference on Human Factors in Computing Systems - CHI '17, pages 313–324, New York, New York, USA, 2017. ACM Press.
- [RPPH12] Majken K. Rasmussen, Esben W. Pedersen, Marianne G. Petersen, and Kasper Hornbæk. Shape-Changing Interfaces: A Review of the Design Space and Open Research Questions. In Proceedings of the 2012 ACM annual conference on Human Factors in Computing Systems - CHI '12, page 735, New York, New York, USA, 2012. ACM Press.
 - [RSP11] Yvonne. Rogers, Helen. Sharp, and Jenny Preece. Interaction design : beyond human-computer interaction. Wiley, 2011.
 - [RSP13] Bernhard E. Riecke and Jörg Schulte-Pelkum. Perceptual and Cognitive Factors for Self-Motion Simulation in Virtual Environments: How Can Self-Motion Illusions ("Vection") Be Utilized? In Human Walking in Virtual Environments, pages 27–54. Springer New York, New York, NY, 2013.
- [RTM12] Irene Rae, Leila Takayama, and Bilge Mutlu. One of the gang: supporting in-group behavior for embodied mediated communication. In *Proceedings of the 2012 ACM* annual conference on Human Factors in Computing Systems - CHI '12, page 3091, New York, New York, USA, 2012. ACM Press.
- [RVGG17] Yosra Rekik, Eric Vezzoli, Laurent Grisoni, and Frédéric Giraud. Localized Haptic Texture: A Rendering Technique Based on Taxels for High Density Tactile Feedback Yosra. In Proceedings of the 2017 CHI Conference on Human Factors in Computing Systems - CHI '17, pages 5006–5015, New York, New York, USA, 2017. ACM Press.

- [RVSP09] Bernhard E. Riecke, Aleksander Väljamäe, and Jörg Schulte-Pelkum. Moving sounds enhance the visually-induced self-motion illusion (circular vection) in virtual reality. ACM Transactions on Applied Perception, 6(2):1–27, feb 2009.
 - [SAL06] Sriram Subramanian, Dzimitry Aliakseyeu, and Andrés Lucero. Multi-layer interaction for digital tables. In Proceedings of the 19th annual ACM symposium on User interface software and technology - UIST '06, page 269, New York, New York, USA, 2006. ACM Press.
- [SBC⁺16] Paul Strohmeier, Jesse Burstyn, Juan Pablo Carrascal, Vincent Levesque, and Roel Vertegaal. ReFlex: A Flexible Smartphone with Active Haptic Feedback for Bend Input. In Proceedings of the TEI '16: Tenth International Conference on Tangible, Embedded, and Embodied Interaction - TEI '16, pages 185–192, New York, New York, USA, 2016. ACM Press.
 - [SC18] Anurag Sarkar and Seth Cooper. Comparing paid and volunteer recruitment in human computation games. In ACM International Conference Proceeding Series, pages 1–9, New York, New York, USA, aug 2018. Association for Computing Machinery.
 - [SF11] Rebecca J. St George and Richard C. Fitzpatrick. The sense of self-motion, orientation and balance explored by vestibular stimulation. *The Journal of Physiology*, 589(4):807–813, feb 2011.
- [SFJ⁺17] Sidharth Sahdev, Clifton Forlines, Ricardo Jota, Bruno De Araujo, Braon Moseley, Jonathan Deber, Steven Sanders, Darren Leigh, and Daniel Wigdor. GhostID: Enabling Non-Persistent User Differentiation in Frequency-Division Capacitive Multi-Touch Sensors. In Proceedings of the 2017 CHI Conference on Human Factors in Computing Systems - CHI '17, pages 15–27, New York, New York, USA, 2017. ACM Press.
- [SGB⁺18] Hyungki Son, Hyunjae Gil, Sangkyu Byeon, Sang Youn Kim, and Jin Ryong Kim. Realwalk: Feeling ground surfaces while walking in virtual reality. In Conference on Human Factors in Computing Systems - Proceedings, 2018.
 - [SH06] Masanori Sugimoto and Keiichi Hiroki. HybridTouch: An Intuitive Manipulation Technique for PDAs Using Their Front and Rear Surfaces. In Proceedings of the 8th conference on Human-computer interaction with mobile devices and services -MobileHCI '06, page 137, New York, New York, USA, 2006. ACM Press.
 - [SH09] Orit Shaer and Eva Hornecker. Tangible User Interfaces: Past, present, and future directions. Foundations and Trends in Human-Computer Interaction, 3(1-2):1–137, 2009.
 - [SHS16] Deepak Ranjan Sahoo, Kasper Hornbæk, and Sriram Subramanian. TableHop: An Actuated Fabric Display Using Transparent Electrodes. In Proceedings of the 2016 CHI Conference on Human Factors in Computing Systems - CHI '16, pages 3767–3780, New York, New York, USA, 2016. ACM Press.
 - [SJM13] Jürgen Steimle, Andreas Jordt, and Pattie Maes. Flexpad: Highly Flexible Bending Interactions for Projected Handheld Displays. In Proceedings of the SIGCHI Conference on Human Factors in Computing Systems - CHI '13, page 237, New York, New York, USA, 2013. ACM Press.

- [SKBH18] Paul Strohmeier, Jarrod Knibbe, Sebastian Boring, and Kasper Hornbæk. zPatch: Hybrid Resistive/Capacitive eTextile Input. In Proceedings of the Twelfth International Conference on Tangible, Embedded, and Embodied Interaction - TEI '18, pages 188–198, New York, New York, USA, 2018. ACM Press.
- [SMSJ⁺15] Shaikh Shawon Arefin Shimon, Sarah Morrison-Smith, Noah John, Ghazal Fahimi, and Jaime Ruiz. Exploring User-Defined Back-Of-Device Gestures for Mobile Devices. In Proceedings of the 17th International Conference on Human-Computer Interaction with Mobile Devices and Services - MobileHCI '15, pages 227–232, New York, New York, USA, 2015. ACM Press.
 - [SO13] Oliver Schoenleben and Antti Oulasvirta. Sandwich keyboard. In Proceedings of the 15th international conference on Human-computer interaction with mobile devices and services - MobileHCI '13, page 175, New York, New York, USA, 2013. ACM Press.
 - [SPB13] Mike Sinclair, Michel Pahud, and Hrvoje Benko. TouchMover. In Proceedings of the 2013 ACM international conference on Interactive tabletops and surfaces - ITS '13, pages 287–296, New York, New York, USA, 2013. ACM Press.
 - [SPB14] Mike Sinclair, Michel Pahud, and Hrvoje Benko. TouchMover 2.0 3D touchscreen with force feedback and haptic texture. In 2014 IEEE Haptics Symposium (HAPTICS), pages 1–6. IEEE, feb 2014.
 - [SPC⁺16] Ben Shneiderman, Catherine Plaisant, Maxine (Maxine S.) Cohen, Steven M. Jacobs, and Niklas Elmqvist. *Designing the user interface : strategies for effective human-computer interaction*. Addison-Wesley Longman Publishing Co., Inc., USA, 3rd edition, 2016.
 - [Sut03] Alistair Sutcliffe. Multimedia and virtual reality: Designing multisensory user interfaces, volume 9781410607. Psychology Press, 2003.
- [SYT⁺17] Evan Strasnick, Jackie Yang, Kesler Tanner, Alex Olwal, and Sean Follmer. shiftIO: Reconfigurable Tactile Elements for Dynamic Affordances and Mobile Interaction. In CHI 2017, 2017.
 - [TBS13] Luca Turchet, Paolo Burelli, and Stefania Serafin. Haptic Feedback for Enhancing Realism of Walking Simulations. *IEEE Transactions on Haptics*, 6(1):35–45, 2013.
 - [TDS99] James N. Templeman, Patricia S. Denbrook, and Linda E. Sibert. Virtual Locomotion: Walking in Place through Virtual Environments. *Presence: Teleoperators* and Virtual Environments, 8(6):598–617, dec 1999.
 - [TH13] Leila Takayama and Helen Harris. Presentation of (Telepresent) Self: On the Double-edged Effects of Mirrors. In Proceedings of the 8th ACM/IEEE International Conference on Human-robot Interaction, HRI '13, pages 381–388, Piscataway, NJ, USA, 2013. IEEE Press.
- [TMM⁺12] Leo Terziman, Maud Marchal, Franck Multon, Bruno Arnaldi, and Anatole Lecuyer. The King-Kong Effects: Improving sensation of walking in VR with visual and tactile vibrations at each step. In 2012 IEEE Symposium on 3D User Interfaces (3DUI), pages 19–26. IEEE, mar 2012.

- [TMM⁺13] Leo Terziman, Maud Marchal, Franck Multon, Bruno Arnaldi, and Anatole Lecuyer. Personified and Multistate Camera Motions for First-Person Navigation in Desktop Virtual Reality. *IEEE Transactions on Visualization and Computer Graphics*, 19(4):652–661, apr 2013.
 - [TR18] Hsin-Ruey Tsai and Jun Rekimoto. ElasticVR. In Extended Abstracts of the 2018 CHI Conference on Human Factors in Computing Systems - CHI '18, pages 1–4, New York, New York, USA, 2018. ACM Press.
 - [Tsa00] Lung-Wen Tsai. Mechanism design: enumeration of kinematic structures according to function. Journal of Mechanical Design, 122(4), 2000.
 - [UCP08] Hajime Uchiyama, Michael A. Covington, and Walter D. Potter. Vibrotactile Glove guidance for semi-autonomous wheelchair operations. In Proceedings of the 46th Annual Southeast Regional Conference on XX - ACM-SE 46, page 336, New York, New York, USA, 2008. ACM Press.
- [VAB⁺20] Preeti Vyas, Feras Al Taha, Jeffrey R. Blum, Antoine Weill-Duflos, and Jeremy R. Cooperstock. Ten Little Fingers, Ten Little Toes: Can Toes Match Fingers for Haptic Discrimination? *IEEE Transactions on Haptics*, 2020.
- [VBV⁺12] Ramiro Velázquez, Omar Bazán, Jorge Varona, Carlos Delgado-Mata, and Carlos A. Gutiérrez. Insights into the Capabilities of Tactile-Foot Perception. International Journal of Advanced Robotic Systems, 9(5):179, nov 2012.
 - [VDH14] Yon Visell, Keerthi Adithya Duraikkannan, and Vincent Hayward. A device and method for multimodal haptic rendering of volumetric stiffness. In International Conference on Human Haptic Sensing and Touch Enabled Computer Applications, volume 8618, pages 478–486. Springer, Berlin, Heidelberg, 2014.
 - [VJ17] Jered Vroon and Jered. Responsive Social Positioning Behaviors for Semi-Autonomous Telepresence Robots. In Proceedings of the Companion of the 2017 ACM/IEEE International Conference on Human-Robot Interaction - HRI '17, pages 383–384, New York, New York, USA, 2017. ACM Press.
- [VSA⁺15] Eduardo Velloso, Dominik Schmidt, Jason Alexander, Hans Gellersen, and Andreas Bulling. The Feet in Human–Computer Interaction. ACM Computing Surveys, 48(2):1–35, sep 2015.
- [VSBH10] Dimitar Valkov, Frank Steinicke, Gerd Bruder, and Klaus H. Hinrichs. Traveling in 3D Virtual Environments with Foot Gestures and a Multi-Touch enabled WIM. In Proceedings of the Virtual Reality International Conference (VRIC), pages 171–180. ACM Press, 2010.
- [VTBC20] Preeti Vyas, Feras Al Taha, Jeffrey R. Blum, and Jeremy R. Cooperstock. HapToes: Vibrotactile Numeric Information Delivery via Tactile Toe Display. In *IEEE Haptics Symposium, HAPTICS*, 2020.
 - [Wal11] Nicholas Walliman. Research Methods: The Basics. Routledge, 2nd edition, 2011.
 - [Wei68] Sidney Weinstein. Intensive and extensive aspects of tactile sensitivity as a function of body part, sex and laterality. In *The Skin Senses*, 1968.

- [WF95] Toshio Watanabe and Shigehisa Fukui. Method for controlling tactile sensation of surface roughness using ultrasonic vibration. In Proceedings - IEEE International Conference on Robotics and Automation, 1995.
- [WFB⁺07] Daniel Wigdor, Clifton Forlines, Patrick Baudisch, John Barnwell, and Chia Shen. Lucid touch. In Proceedings of the 20th annual ACM symposium on User interface software and technology - UIST '07, page 269, New York, New York, USA, 2007. ACM Press.
- [WFK⁺19] Andrew M. Webb, Hannah Fowler, Andruid Kerne, Galen Newman, Jun-Hyun Kim, and Wendy E. Mackay. Interstices: Sustained Spatial Relationships between Hands and Surfaces Reveal Anticipated Action. In Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems - CHI '19, pages 1–12, New York, New York, USA, 2019. ACM Press.
 - [WI16] JO Wobbrock, JA Kientz Interactions, and Undefined 2016. Research contributions in human-computer interaction. *dl.acm.org*, 2016.
- [WLF⁺06] Daniel Wigdor, Darren Leigh, Clifton Forlines, Samuel Shipman, John Barnwell, Ravin Balakrishnan, and Chia Shen. Under the table interaction. In Proceedings of the 19th annual ACM symposium on User interface software and technology -UIST '06, page 259, New York, New York, USA, 2006. ACM Press.
- [WLT⁺20] Alexander Wilberz, Dominik Leschtschow, Christina Trepkowski, Jens Maiero, Ernst Kruijff, and Bernhard Riecke. FaceHaptics: Robot Arm based Versatile Facial Haptics for Immersive Environments. In Conference on Human Factors in Computing Systems - Proceedings, 2020.
- [WMA08] Jacob O. Wobbrock, Brad A. Myers, and Htet Htet Aung. The performance of hand postures in front- and back-of-device interaction for mobile computing. *International Journal of Human Computer Studies*, 66(12):857–875, dec 2008.
- [WRS17] Qinglong Wang, Xiangshi Ren, and Xiaoying Sun. Enhancing Pen-based Interaction using Electrovibration and Vibration Haptic Feedback. In Proceedings of the 2017 CHI Conference on Human Factors in Computing Systems - CHI '17, pages 3746– 3750, New York, New York, USA, 2017. ACM Press.
- [WRSS16] Qinglong Wang, Xiangshi Ren, Sayan Sarcar, and Xiaoying Sun. EV-Pen: Leveraging Electrovibration Haptic Feedback in Pen Interaction. In Proceedings of the 2016 ACM on Interactive Surfaces and Spaces - ISS '16, pages 57–66, New York, New York, USA, 2016. ACM Press.
 - [WS98] Bob G. Witmer and Michael J. Singer. Measuring Presence in Virtual Environments: A Presence Questionnaire. Presence: Teleoperators and Virtual Environments, 7(3):225–240, jun 1998.
- [WSR14] Katrin Wolf, Robert Schleicher, and Michael Rohs. Ergonomic characteristics of gestures for front- and back-of-tablets interaction with grasping hands. In Proceedings of the 16th international conference on Human-computer interaction with mobile devices & services - MobileHCI '14, pages 453–458, New York, New York, USA, 2014. ACM Press.

- [WTC⁺20] Yue Wang, Takara E. Truong, Samuel W. Chesebrough, Pete Willemsen, K. Bo Foreman, Andrew S. Merryweather, John M. Hollerbach, and Mark A. Minor. Augmenting Virtual Reality Terrain Display with Smart Shoe Physical Rendering: A Pilot Study. *IEEE Transactions on Haptics*, 2020.
- [WYLS13] Yuntao Wang, Chun Yu, Jie Liu, and Yuanchun Shi. Understanding performance of eyes-free, absolute position control on touchable mobile phones. In Proceedings of the 15th international conference on Human-computer interaction with mobile devices and services - MobileHCI '13, page 79, New York, New York, USA, 2013. ACM Press.
- [XHW13] Xiang Xiao, Teng Han, and Jingtao Wang. LensGesture. In Proceedings of the 15th ACM on International conference on multimodal interaction - ICMI '13, pages 287–294, New York, New York, USA, 2013. ACM Press.
- [YILK15] Nesra Yannier, Ali Israr, Jill Fain Lehman, and Roberta L. Klatzky. FeelSleeve: Haptic Feedback to Enhance Early Reading. In *Proceedings of the 33rd Annual ACM Conference on Human Factors in Computing Systems*, page 10, Seoul, Republic of Korea, 2015. Association for Computing Machinery (ACM).
- [YJNS18] Lillian Yang, Brennan Jones, Carman Neustaedter, and Samarth Singhal. Shopping Over Distance through a Telepresence Robot. Proceedings of the ACM on Human-Computer Interaction, 2(CSCW):1–18, nov 2018.
 - [YK08] Gi-Hun Yang and Dong-Soo Kwon. KAT II: Tactile Display Mouse for Providing Tactile and Thermal Feedback. *Advanced Robotics*, 22(8):851–865, jan 2008.
- [ZDE16] Qi Zhang, Haiwei Dong, and Abdulmotaleb El Saddik. Magnetic Field Control for Haptic Display: System Design and Simulation. *IEEE Access*, 4:299–311, 2016.

Appendix A

Questionnaires

Questionnaire 1 (actuated screen Section 4.2)	-3	-2	-1	0	1	2	3
I could perform the experiment well using the separation							
between display and tablet	0	0	0	0	0	0	0
The pose to control the tablet was comfortable	0	0	0	0	0	0	0
I could fellow the speed markers easily	0	0	0	0	0	0	0
I was fully satisfied with the feedback of the system	0	0	0	0	0	0	0
per condition							
The interaction metaphor was good for this application	0	0	0	0	0	0	0
The tablet was easy to adjust to interact with the data	0	0	0	0	0	0	0
The position and inclination helped understanding 'physics'	0	0	0	0	0	0	0
I was fully satisfied with the app	0	0	0	ο	0	ο	0

Questionnaire 2 (flexible material Section 4.3)					
per condition	1	2	3	4	5
I did not have to concentrate much while detecting a difference	0	0	0	0	0
Perceiving tension cues was easy	0	0	0	0	0
I found the system easy to use	0	0	0	0	0
I found the system easy to learn	0	0	0	0	0
I felt confident using the system	0	0	0	0	0
I did not feel frustrated when performing the tasks	0	0	0	0	0
I found the responsiveness of tension feedback adequate	0	0	0	0	0
I could well receive cues with the pen	0	0	0	0	0
The noise produced by the system did not disturb me while performing the task	0	0	0	0	0
I would like to use a similar system in the future	0	0	0	0	0

Questionnaire 3 (actuated pins Section 4.4)							
per condition	1	2	3	4	5	6	7
I could perform the task precisely	0	0	0	0	0	0	0
I could complete the task quickly	0	0	0	0	0	0	0
The feedback quality was excellent	0	0	0	0	0	0	0
matrix selection							
Different feedback intensities enabled a better orientation in the Matrix	0	0	0	0	0	0	0

Questionnaire 4 (robot control Section 5.2)	1	2	3	4	5	6	7
How much were you able to control events?	0	о	о	о	о	0	0
How responsive was the environment to actions that you initiated							
(or performed)?	0	0	0	0	0	0	0
How natural did your interactions with the environment seem?	0	0	0	0	0	0	0
How much did the visual aspects of the environment involve you?	0	0	0	0	0	0	0
How natural was the mechanism which controlled movement							
through the environment?	0	0	0	0	0	0	0
How compelling was your sense of objects moving through space?	0	0	0	0	0	0	0
How much did your experiences in the virtual environment							
seem consistent with your real world experiences?	0	0	0	0	0	0	0
Were you able to anticipate what would happen next							
in response to the actions that you performed?	0	0	0	0	0	0	0
How completely were you able to actively survey or search							
the environment using vision?	0	0	0	0	0	0	0
How compelling was your sense of moving around							
inside the virtual environment?	0	0	0	0	0	0	0
How closely were you able to examine objects?	0	0	0	0	0	0	0
How well could you examine objects from multiple viewpoints?	0	0	0	0	0	0	0
How involved were you in the virtual environment experience?	0	0	0	0	0	0	о
How much delay did you experience between your							
actions and expected outcomes?	0	0	0	0	0	0	0
How quickly did you adjust to the virtual environment							
experience?	0	0	0	0	0	0	0
How proficient in moving and interacting with the virtual environment							
did you feel at the end of the experience?	0	0	0	0	0	0	0
How much did the visual display quality interfere or distract you							
from performing assigned tasks or required activities?	0	0	0	0	0	0	0
How much did the control devices interfere with the performance							
of assigned tasks or with other activities?	0	0	0	0	0	0	0
How well could you concentrate on the assigned tasks or required							
activities rather than on the mechanisms used to perform those							
tasks or activities?	0	0	0	0	0	0	0
How much did the auditory aspects of the environment involve you?	0	0	0	0	0	0	0
How well could you identify sounds?	0	0	0	0	0	0	0
How well could you localize sounds?	0	0	0	0	0	0	0
How well could you actively survey or search the virtual environment							
using touch?	0	0	0	0	0	0	0
How well could you move or manipulate objects in the virtual							
environment?	0	0	0	0	0	0	0

Questionnaire 5 (walking-like cues Section 5.3)	0	1	2	3	4	5	6	7	8	9	10
I had a strong sensation of self-motion	0	0	0	0	0	0	0	0	0	0	0
I could judge velocity travelled well	0	0	0	0	0	0	0	0	0	0	0
I could judge distance travelled well	0	0	0	0	0	0	0	0	0	0	0
I was not aware of my real environment	0	0	0	0	0	0	0	0	0	0	0

Questionnaire 6 (walking-like cues Section 5.3)	0	1	2	3	4	5	6	7	8	9	10
per condition											
Walking on wood was convincing	0	0	0	0	0	0	0	0	0	0	ο
Walking on gravel was convincing	0	0	0	0	0	0	0	0	0	0	ο
I felt comfortable	0	0	0	0	0	0	0	0	0	0	ο
I was able to concentrate on the task	0	0	0	0	0	0	0	0	0	0	ο
I was able to navigate and follow the object	0	0	0	0	0	0	0	0	0	0	ο
The interface was easy to learn	0	0	0	0	0	0	0	0	0	0	0
It was fun using the interface	0	0	0	0	0	0	0	0	0	0	ο
I could imagine using the interface for											
a longer period of time	0	0	0	0	0	0	0	0	0	0	ο
I had a strong sense of self-motion	0	0	0	0	0	0	0	0	0	0	ο
I was fully satisfied with the usability of the interface	0	0	0	0	0	0	0	0	0	0	ο
the following stimuli (cues) have had											
a positive influence on vection:											
audio	0	0	0	0	0	0	0	0	0	0	ο
light vibrations	0	0	0	0	0	0	0	0	0	0	ο
strong vibrations (heel)	0	0	0	0	0	0	0	0	0	0	ο
head bobbing	0	ο	0	0	ο	0	0	0	0	0	0

Appendix B

Further Results



Figure B.1: Results of psychophysical study performed to analyse our third prototype in the first focus area. It shows the psychometric curves obtained in experiment (see Section 4.4)



Figure B.2: Results of the in-study questionnaire of session 4. The graphs showed the worse scores of the walking in place metaphor (see Section 5.3.2).



Figure B.3: Results of the post-study questionnaire of session 4. Again, the graphs showed the worse scores of the walking in place metaphor (see Section 5.3.2).